

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Grau, Aaron

Research Report Market power in the German dairy supply chain

Studies on the Agricultural and Food Sector in Transition Economies, No. 90

Provided in Cooperation with: Leibniz Institute of Agricultural Development in Transition Economies (IAMO), Halle (Saale)

Suggested Citation: Grau, Aaron (2018) : Market power in the German dairy supply chain, Studies on the Agricultural and Food Sector in Transition Economies, No. 90, ISBN 978-3-95992-062-9, Leibniz Institute of Agricultural Development in Transition Economies (IAMO), Halle (Saale), https://nbn-resolving.de/urn:nbn:de:gbv:3:2-94091

This Version is available at: https://hdl.handle.net/10419/180396

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU

Studies on the Agricultural and Food Sector in Transition Economies

Aaron Grau

Market power in the German dairy supply chain

Leibniz Institute of Agricultural Development in Transition Economies

Market power in the German dairy supply chain

Studies on the Agricultural and Food Sector in Transition Economies

Edited by Leibniz Institute of Agricultural Development in Transition Economies IAMO

Volume 90

Market power in the German dairy supply chain

by Aaron Grau

IAMO

2018

Bibliografische Information Der Deutschen Bibliothek

Die Deutsche Bibliothek verzeichnet diese Publikation in der Deutschen Nationalbibliografie; detaillierte bibliografische Daten sind im Internet über http://dnb.ddb.de abrufbar.

Bibliographic information published by Die Deutsche Bibliothek

Die Deutsche Bibliothek lists the publication in the Deutsche Nationalbibliografie; detailed bibliographic data are available in the internet at: http://dnb. ddb.de.

This thesis was accepted as a doctoral dissertation in fulfillment of the requirements for the degree "doctor agriculturarum" by the Faculty of Natural Sciences III at Martin Luther University Halle-Wittenberg on 05.04.2017.

Date of oral examination:	17.07.2017
Supervisor and Reviewer:	Prof. Dr. Heinrich Hockmann
Co-Reviewer:	Prof. Dr. Martin Odening

Diese Veröffentlichung kann kostenfrei im Internet unter <www.iamo.de/dok/sr_vol90pdf> heruntergeladen werden.

This publication can be downloaded free from the website <www.iamo.de/dok/sr_vol90pdf>.

© 2018 Leibniz-Institut für Agrarentwicklung in Transformationsökonomien (IAMO) Theodor-Lieser-Straße 2 06120 Halle (Saale) Tel.: 49 (345) 2928-0 e-mail: iamo@iamo.de http://www.iamo.de ISSN 1436-221X ISBN 978-3-95992-062-9

ACKNOWLEDGEMENTS

Firstly, I would like to express my sincere gratitude towards my advisor Prof. Dr. Heinrich Hockmann for the continuous support of my PhD study and related research, for his encouragement, and immense knowledge. He presented me with the original idea, which would later become this PhD thesis. I am very grateful that his office door was always open to me and that he took all the time in the world to discuss patiently my ideas.

Besides my advisor, I would like to thank the second member on my committee Prof Dr. Martin Odening for his insightful comments and encouragement.

I am very grateful to Prof. Dr. Thomas Glauben for giving me the opportunity and the financial support to conduct and finish my PhD thesis as well as to attend many courses of the doctoral Certificate Program in Agricultural Economics as well as conferences around the globe. I also would like to thank the entire staff at IAMO for their support in many different aspects of my work and personal life as a PhD student.

My sincere thanks also go to Dr. Inna Levkovych for her motivating support, encouraging discussions, and successful collaboration on many diverse research topics. Her persistence and constructive advice helped me to improve my skills as a researcher and the overall quality of my work.

Further, I have to thank my colleagues Florian, Klodjan, Lena, Ivan, Maria, Miranda, Giorgi, Brett, Tim, and many others here at IAMO, who generated this pleasant working environment and whom I proudly call my friends.

Last but not least, I would like to especially thank my wife, parents, and family for believing in me, their love, and limitless spiritual and financial support throughout my PhD studies as well as my entire educational career.

ABSTRACT

The German dairy supply chain is one of the most important agrifood supply chains in Germany. In 2015, raw milk production accounted for 19.2% of the total domestic agricultural production value and dairy products summed up 8.0% of total German agrifood exports. Nevertheless, the economic success of the supply chain is overshadowed by reports on market power abused by dairies and retailers. Fueled by complaints of dairy farmers and dairies on non-competitive behavior of downstream supply chain agents, the German anti-trust agency conducted a sector analysis between 2008 and 2012. In its final report the anti-trust agency acknowledged the threat of oligopsony power at both markets, raw milk and dairy output market, but could not find any evidence. Thus, the main premise of the presented work is to empirically investigate the German dairy supply chain for the existence of oligopsony power at the raw milk and dairy product market.

Three approaches, structure-conduct-performance paradigm (SCPP), new empirical industrial organization (NEIO), and asymmetric price transmission (APT), are revised and their suitability for the empirical application discussed. The theory review shows that none of the discussed approaches provide a model that fits the structure of the German dairy supply chain of oligopsonistic threat on two vertically integrated markets. Additional characteristics of the theoretical frameworks limit their suitability for the study further. Either the data requirements are high and the models highly complex, like in the case of NEIO, but an index of market power can be measured, or the data requirements are rather low and the theoretic models rather simple, like in the case of APT, but market power or its extent are not clearly identified. Consequently, a modification of one of the approaches is not sufficient and a new theoretical model merging aspects of NEIO and APT had to be developed.

In a first step the two markets for raw milk as the material input and dairy product as the material output to the transformation process are modeled structurally in the spirit of the NEIO framework. However, the system of derived supply and demand equations is then used to obtain explicit solutions of the endogenous variables, the material input and output prices as well as their quantities. After rearrangement and substitution, two pricing equations, one depicting the farmprocessor relation, and the other depicting the processor-retailer relation, are found which only depend on material input and output prices as well as the non-material input prices, technological change, and the measures for oligopsony power. Consequently, quantity data is obsolete. Unfortunately, the market power indices cannot be uniquely identify from this two pricing equations with the use of standard estimation techniques.

In order to be able to evaluate the extent of market power a new estimation strategy was developed that links the methods vector error correction model (VECM), Kalman-Filter, and dynamic factor analysis (DFA) together. The VECM accounts for stationarity in combinations of the variables, commonly found between price series of integrated markets, and allows estimating the two pricing equations simultaneously as cointegration vectors. The error correction representation is transformed via the simple two step estimation technique of Lütkepohl and Krätzig (2004) in order to be able to apply the Kalman-Filter and to obtain time-varying parameters. The time-variation is assumed to be caused solely by changes in oligopsony conduct and is extracted in form of dynamic factors through the application of the DFA. The results of the DFA do not only permit to uniquely identify the time-varying market power indices, but also the price elasticities of supply. With knowledge of these elasticities as well as the market power indices the so-called buyer power indices can be calculated which measure the relative effects of the oligopsony conduct on the prices.

The derived theoretical model and the estimation strategy were applied to the German dairy supply chain using monthly data over the time horizon from January 2003 to December 2015. The data consisted of price time series comprising the price for raw milk, a dairy output price derived by share-weighting the prices of important dairy products, as well as prices of other cost variables involved in the product transformation process along the chain. The initial statistical analysis revealed non-stationary of the variables and two long-rung cointegration relationships among them, justifying the theoretical model and the use of a VECM. The estimation strategy was applied to a total of nine different VECM specifications differing in lag length and assumptions on the other cost variables. The specification with a lag length of two and weakly exogenous variables was the best fit for the data.

The empirical application was successful and revealed theoretically consistent and statistically significant market power indices on the German raw milk market and dairy output market for the entire time period. While the raw milk market is close to perfect competition with a market power index value range of 0.038 to 0.048, the dairy output market is subject to larger market imperfections with a value of 0.14 to 0.22 and reaches Cournot level at times. The rather inelastic supply of raw milk and dairy products due to asset specificity and high investment costs lead in collaboration with the derived market power indices to large mark-downs. While the German raw milk price was marked-down by 21% to 48%, the mark-down of the share-weighted dairy output price ranges between 49% and 82% over the analyzed time period. Overall, large rents were shifted in disadvantage of dairies and in particular of dairy farmers downstream the supply chain and overall welfare was reduced.

To overcome the market imperfections on each market of the German dairy supply chain different steps can be taken. While low levels of oligopsony conduct are observed on the raw milk market, the inelastic response of raw milk supply to price changes causes the rather large mark-downs. Consequently, policies that enhance the responsiveness of supply, e.g. credit availability and technology transfer to dairy farmers, should be implemented. For the dairy output market, the extent of market power and, because of a similar inelastic supply, the markdowns are larger. Consequently, measures to decrease market power need to be promoted. One possibility is to increase the number of buyers of dairy products and thus the marketing options for dairies. This could be done in the case of the German dairy supply chain by promoting exports or dissolving procurement alliances between larger and smaller retailers.

Several innovations make the presented approach a novelty in the field of market power studies. To our knowledge no attempts to model oligopsony power on two vertically integrated markets exist so far in the economic literature. While the markets are model in the fashion of NEIO, the derivation of partial equilibrium equations and the final pricing equations significantly lower the data requirement and the necessary number of equations for estimation. Moreover, by assuming time-variation in the oligopsony conduct and applying the developed estimation strategy, the market power indices are adaptive rather than 'static'.

TABLE OF CONTENT

A	cknowled	gements	. i
A	ostract		ii
Та	ble of co	ntent	vi
Li	st of table	2S	ix
Li	st of figur	'es	xi
Li	st of abbr	eviations	xiii
1	Introdu	ction	1
2	German	a dairy supply chain	5
	2.1 Daii	ry supply chain	5
	2.1.1	Raw milk production	7
	2.1.2	Milk processing	11
	2.1.3	Marketing	16
	2.1	.3.1 Retailers	16
	2.1	.3.2 Consumers	17
	2.2 Eco	nomic relationships	19
	2.2.1	Vertical relationships	19
	2.2.2	Horizontal relationships	22
	2.2.3	Anti-trust investigations	22
	2.3 Sum	nmary	23
3	Theory	Review	25
	3.1 Stru	cture conduct performance paradigm (SCPP)	25
	3.1.1	Methodology	25
	3.1.2	Criticism	26
	3.2 Nev	v empirical industrial organization (NEIO)	27
	3.2.1	Methodology	28
	3.2.2	Oligopsony model	30

	3.2.3	Criticism	35
	3.2.4	Dynamic models	36
	3.2	4.1 Strategic dynamic models	37
	3.2	4.2 Fundamental dynamic models	40
	3.2	4.3 Criticism	42
	3.3 Pric	transmission analysis	43
	3.3.1	Asymmetric price transmission (APT)	43
	3.3	1.1 APT methodology	47
	3.3	1.2 Cointegration and threshold APT	48
	3.3.2	Equilibrium displacement model (EDM)	51
	3.3.3	Criticism	54
	3.4 Sum	mary	55
4	Empiric	l applications in the dairy sector	57
	4.1 NE)	57
	4.1.1	Oligopoly	57
	4.1.2	Oligopoly/Oligopsony	58
	4.1.3	Oligopsony	59
	4.2 APT		59
	4.3 'Firs	t-pass' test	60
	4.4 Sum	mary	61
5	Theoret	cal model	65
	5.1 Olig	ppsony model of two markets	67
	5.2 Pric	ng equations and determining the level of market power	69
	5.3 Sum	mary	72
6	Method	ology	73
	6.1 VEC	Μ	73
	6.2 Stat	e-space representation of the transformed VECM	75
	6.3 Dyn	amic factor analysis	75

	6.4	Sum	mary	76
7	Em	pirica	al application	79
	7.1	Dese	criptive statistics	79
	7.2	Initi	al cointegration analysis	82
	7.3	Resu	ılts	87
	7	.3.1	Model selection	87
	7	.3.2	State-space representation	89
	7	.3.3	Dynamic factor analysis	92
	7	.3.4	Market power indices	95
	7	.3.5	Elasticities of supply	97
	7	.3.6	Buyer power indices	98
	7.4	Sum	mary	101
8	Dis	cussi	on	103
	8.1	Mar	ket power indices	103
	8.2	Elas	ticities of supply	105
	8.3	Buy	er power indices	106
	8.4	Acti	ons against price mark-downs	109
	8.5	Sho	rtcomings of the approach	110
	8.6	Sum	mary	111
9	Cor	nclusi	ions	115
Re	fere	nces		119
Aŗ	pen	dix .		135

LIST OF TABLES

Total stock of dairy cows, average milk production per cow, total milk production, milk delivered to dairies, average fat and protein content of delivered milk, and average milk price at farm gate in selected years.	. 7
Structure of German dairy cow farming in 2003 and 2013	. 8
Total German production of dairy products, import and export, consumption and self-sufficiency rate in selected years.	12
Number of German dairies, total dairy employees, (domestic and external) revenue of German dairies, and average revenue per dairy and employee in selected years.	13
Structure of German milk processing in 2003 and 2012.	14
Revenue of the TOP 10 German dairy companies in 2015	15
Structure of German food retailing in 2003 and 2014	16
Share of total sales for selected dairy products for discounters, larger retailers, supermarkets and other establishments in 2014.	18
Per capita consumption of selected dairy products in Germany in 2003 and 2014.	18
NEIO studies in the dairy sector.	62
APT studies in the dairy sector.	63
Descriptive statistics of used dataset.	80
Augmented Dickey Fuller unit root test results	83
Saikkonen & Lütkepohl cointegration test results for system with seven dimensions including a deterministic constant and trend term.	84
	Total stock of dairy cows, average milk production per cow, total milk production, milk delivered to dairies, average fat and protein content of delivered milk, and average milk price at farm gate in selected years

Table 15:	List of model specifications differing in lag length and	
	the nature of the non-material input variables with their	
	corresponding FPE results for AICc and the weighted	
	AICc values	88
Table 16:	DFA results for the constant parameters ωs of the FPE with corresponding standard errors and significance level.	94
Table 17:	DFA results for the constant parameters τs of the PRE with	
	corresponding standard errors and significance level	95

LIST OF FIGURES

Figure 1:	Structure of the German dairy supply chain (2014/15)	6
Figure 2:	German milk quota and raw milk delivered to dairies from 2004 to 2015.	9
Figure 3:	German annual averaged raw milk price from 2003 to 2015.	10
Figure 4:	German consumer prices (in €) for fresh milk, non-blended butter and Gouda cheese in selected years.	19
Figure 5:	Illustrations of cases of symmetry (upper left), asymmetry in speed (upper right), asymmetry in magnitude (lower left), and asymmetry in speed and magnitude (lower right) in the price transmission of P^d (downstream price) and P^u (upstream price), assuming that a shock in P^d is transmitted to P^u .	44
Figure 6:	Illustration of positive APT (left plot) and negative APT (right plot).	45
Figure 7:	Monopsony <i>m</i> and competitive <i>c</i> outcome on the material input (left) and output market (right).	69
Figure 8:	Estimation strategy.	77
Figure 9:	Price for German brand butter (formed), SMP, and Emmentaler in €/100g, and the calculated dairy output price in €/l.	81
Figure 10:	Deflated prices for raw milk and dairy output as well as their margin in €/l.	82
Figure 11:	p-values of the Chow forecast test (100 bootstrap replications) for the VECM specification with 1, 2, and 12 lags.	85

Figure 12:	Time-varying parameter (TVP) estimates for the farm-processor cointegration vector including dotted lines for the corresponding 95%-confidence intervals
Figure 13:	Time-varying parameter (TVP) estimates for the processor-retailer cointegration vector including dotted lines for the corresponding 95%-confidence intervals
Figure 14:	Common factor estimates for the farm-retailer cointegration parameters including dotted lines for the corresponding 95%-confidence interval. 92
Figure 15:	Common factor estimates for the processor-retailer cointegration parameters including dotted lines for the corresponding 95%-confidence interval
Figure 16:	Calculated market power index Ξ^{P}_{t} of dairies
Figure 17:	Calculated market power index Ξ^{R}_{t} of retailers
Figure 18:	Calculated price elasticity of raw milk supply η^{F}_{t}
Figure 19:	Calculated price elasticity of dairy output supply η_{t}^{P}
Figure 20:	Calculated buyer power index of dairies 100
Figure 21:	Calculated buyer power index of retailers 100
Figure 22:	Observed raw milk price (continuous line), observed raw milk price plus absolute mark-down (taunted line), and their difference (dotted line) for the analyzed time period
Figure 23:	Observed dairy output price (continuous line), observed dairy output price plus absolute mark-down (taunted line), and their difference (dotted line) for the analyzed time period

LIST OF ABBREVIATIONS

AC	Average cost
ADF	Augmented Dickey Fuller Test
AIC	Akaike Info Criterion
AICc	Akaike Info Criterion with correction
APT	Asymmetric price transmission
BPI	Buyer power index
CE	Conjectural elasticity
CPI	Consumer price index
CPM	Conduct parameter method
CRS	Constant returns-to-scale
CV	Conjectural variation
DFA	Dynamic factor analysis
ECM	Error correction model
ECT	Error correction term
EDM	Equilibrium displacement model
EM	Expectation-maximization algorithm
EU	European Union
FP	Final Prediction Error
FPE	Farm-processor equation
GIM	General identification methodology
HQC	Hanna-Quinn Criterion
MARSS	Multivariate autoregressive state-space
MFC	Marginal factor cost
MC	Marginal cost
ME	Milk equivalent
ML	Maximum likelihood
MR	Marginal revenue
MRP	Marginal revenue of product
MS	Markov-switching
NEIO	New empirical industrial organization
OLS	Ordinary least squares
PCM	Price-cost margin
PRE	Processor-retailer equation
PTA	Production-theoretic approach
S2S	Simple two step estimation procedure
SC	Schwarz Criterion
SCPP	Structure conduct performance paradigm
SMP	Skim milk powder
UK	United Kingdom

USA	United States of America
VC	Variable cost
VECM	Vector error correction mode
VMP	Value of marginal product

1 INTRODUCTION

The study of market power in agri-food supply chains possesses a long history in economic literature. However, it has been primarily dominated by oligopoly studies. For a long time agricultural markets were even regarded as prime example of perfect competition (Sexton, 2013). The reason that the threat of seller power by the food processing industries was early identified as plausible and a rich field for empirical application is that heavy concentration and consolidation processes shaped most agri-food supply chains in developed countries into a form of an hour glass until the 1950s and 1960s. In the U.S. food processing sector, for example, 50% of the value added in this sector was achieved by only 20 manufacturers in 1995, while in 1954 they already accounted for half of the share (Sexton, 2000).

On the head and tail of agri-food supply chains concentration remained low. Agricultural producers persisted to be atomistic and retailers heterogeneous. This structure allowed the processing sector to gain a dominant position and to drive distribution with the development of strong brands and the use of whole-salers and retailers as a link to consumers (Dobson et al., 2003). Consequently, research until the 1980s focused more and more on the food processing sector as a possible source of oligopoly power with welfare loss implications, in particular, for retailers and final consumers. The notion of competitive farm commodity markets persisted (Swinnen & Vanderplas, 2010).

In the late 1980s and 1990s research focus shifted slowly towards the idea of oligopsony power on the raw agricultural commodity markets. Finally, the possibility of buyer power of a concentrated processing sector towards the atomic farm sector was acknowledged (e.g. Muth & Wohlgenant, 1999; Schroeter, 1988). Apart from different levels of concentration, further sector specific characteristics of the agricultural sector were quickly found that promote the threat of buyer power. Agricultural products are often less 'mobile' due their perishableness, processors are highly depended on non-substitutable, specialized agricultural inputs, and farmers are highly specialized in the production of specific goods through extensive investment, which represent exit barriers for farmers and cause raw supply to be inelastic (Rogers & Sexton, 1994).

Starting in the 1990s, mergers and acquisition activities have heavily promoted concentration at the retail level in the EU and the USA. By mid-1990 retail concentration in the EU surpassed food sector concentration, and today is much higher than concentration in the food processing industry has ever been (Swinnen & Vandeplas, 2010). As a result, the TOP 20 retailers in the EU accounted for 40%

of the total revenue at the beginning of the 2000s. In comparison, the TOP 20 food processor only accounted for 15% (Clarke et al., 2003). By the mid-2000s, in some EU countries, such as Germany or the UK, the five largest enterprises encompass more than 70% of retailers' total turnover (Consultative Commission on Industrial Change, 2008).

These rather new developments at the retail level seemed to have reshaped the nature of agri-food supply chains. Apart from the concentration processes at this stage, retailers have frequently integrated wholesaling into their business, thus extinguishing this stage, and further taken control over the upstream stages by demanding specific products and creating own brands (Dobson et al., 2003). Consequently, the manufacturers were replaced by retailers as driver of the agri-food supply chain (Bell et al., 1997).

The new structure of agri-food supply chains has caused a debate in economic literature whether the growing power of retailers might be welfare enhancing or diminishing. A series of authors (e.g. Chen, 2003; Dobson & Waterson, 1997) argue that the growing buyer power in form of bargaining power allows retailers to countervail the original oligopoly of the food sector and extract beneficial conditions for retailers, which ultimately are passed on to consumers. However, for countervailing power to be applicable to the food sector must actually earn rents above the competitive level (Chen, 2008).

Consequently, other authors argue that the retail sector might really exert oligopsony power and push the procurement price by buying less. This does not only harm the food sector, but also consumers, since it has quantity distortion effects (OECD, 2008). However, even if higher concentration in the retail sector might increase oligopsony power it is argued by some authors that gains in efficiency (Demsetz, 1973; Guy et al., 2004; Swinnen & Vanderplas, 2010), reduction in market failures through vertical integration (Slade, 1998), or increasing innovation activities, since firms in an highly concentrated sector benefit from R&D and thus are more inclined to invest (Pray et al., 2005), could offset the negative welfare effect of oligopsony.

The debate of buyer power of retailers was ultimately shifted into the public domain in the year 2009. While consumer food prices and agricultural producer prices experienced a similar trend of rapid growth over the years 2007 and 2008, producer prices quickly dropped below the pre-2007 level in 2009 but consumer prices remained high. Apart from farmers protesting in front of retailers' branches in several EU countries (Swinnen & Vanderplas, 2010), also the European Commission (EC) warned of negative long-term effects of the oligopsony

power of retailers for the entire agri-food sector. The caused reduced profitability and quantity distortions limit food processors' incentives to invest in improved product quality and innovation of production process and consequently lower the future efficiency and competitiveness of the entire chain (EC, 2009).

Nevertheless, research on oligopsonistic behavior of retailers remains scarce. Only a few studies have factored in this development and the possible threat of retailers' buyer power on the food industry output market in their empirical applications (i.e. Fałkowski, 2010; Gohin & Guyomard 2000; Lloyd et al., 2009; Madau et al., 2016). The primary cause for this is the availability of data. Even though, data at the producer and consumer level are frequently publicly available, the terms between retailers and food processors remain mostly in the dark (Lloyd et al., 2009; Sexton, 2013).

The aim of the following work is to overcome this deficit and add to the literature. However, in contrast to the before mentioned studies on oligopsony power by retailers, also oligopsony behavior of food processors towards farmers is considered to approximate the structure of modern agri-food supply chains in developed countries as accurate as possible. In addition, the data prerequisite is significantly lowered by deriving a model that does not require any kind of quantity data, but only requires data on prices, for outputs and inputs along an agri-food supply chain. To our knowledge we are not aware of any similar approach in the economic literature so far.

For empirical application the German dairy supply chain was selected due to characteristics, which make it a primary target for the study of subsequent oligopsony along an agri-food supply chain. As mentioned before, the retail sector is dominated by a handful of firms which generate the major share of food retailing revenue. These retailers face around a hundred dairy companies on the markets for dairy products. Nevertheless, also the dairy stage can be considered as highly concentrated with up to 50% of German raw milk processed by only five companies (Loy et al., 2015). German dairies, on the other hand, source raw milk almost entirely from a domestic and atomistic primary production consisting of ten thousands of dairy farmers. Not surprisingly, the German anti-trust agency has received a growing number of complaints by dairy farmers and dairies on the procurement behavior of downstream buyers, respectively the dairies and retailers, since the start of the 2000s. While dairy farmers have been criticizing that dairies abuse their buyer power by asymmetrically transmitting prices, price decreases on the dairy output market are faster transmitted to the upstream stage than prices increases, dairies have been accusing retailers to abuse their market position to lower the prices for dairy products by colluding. As a result

an official investigation was started by the anti-trust agency in 2008 with its final report published in 2012. The German anti-trust agency stated that even though procurement prices are low and the structure along the chain seems to favor the position of buyers, no evidence on the abuse of market power was found (Bundeskartellamt, 2009).

With the model developed we are able to identify market power and its level along the German dairy supply chain. For this purpose, the time horizon January 2003 to December 2015 was chosen, since it does not only cover the period under investigation by the German anti-trust agency, but also phases of heavy structural change, in particular at the farm and processing level, the EU policy reforms attempting to liberate the milk market, and the before mentioned growing asymmetry in producer and consumer price between 2007 and 2009. The ending date was chosen to also uncover the effects of the recent final abolishment of EU milk quota system in April 2015 on the entire dairy supply chain and possible market power distribution along the chain.

The work is structured in the following way. Chapter 2 provides a description of the German dairy supply chain and its recent developments, also covering vertical and horizontal relationships between chain agents and latest policy changes. From its characteristics and the comment of the German anti-trust agency it is assumed that the supply chain is affected by subsequent oligopsony conduct on the raw milk and dairy product market. The adjacent Chapter 3 delivers an overview of relevant theoretic models in economic literature that have been developed to investigate market power so far. It will become apparent that no theoretic model of subsequent oligopsony has been developed so far. Chapter 4 presents an overview of all known empirical applications of the theoretic models of Chapter 3 to the dairy sector. Chapter 5 derives a theoretic model of oligopsony on two vertically integrated markets and the partial equilibrium equations that only consist of price data. The methodology applied to estimate an index of oligopsony power of the dairies and retailers respectively is introduced in Chapter 6. Chapter 7 presents the empirical application to the German dairy supply chain with its various steps and the obtained results for the level of oligopsony and the actual price mark-down for both markets due to the oligopsony conduct. Chapter 8 gives explanations for the estimated level of market power and lists shortcomings of the approach. Chapter 9 provides final conclusions.

2 GERMAN DAIRY SUPPLY CHAIN

The German dairy supply chain is one of the most important branches of the German food industry. German dairy farmers generated around 10.1 billion € of production value in 2015, which accounted for about 19.2% of the total domestic agricultural production value (52.6 billion €). The dairy industry itself created revenue of 23.7 billion € in the same year. 28% of its revenue was generated outside the domestic market and the total share of milk and dairy products in overall German exports made up 0.7% and of total agri-food exports 8.0%. The German dairy supply chain is also of great significance to global dairy production. In 2015 around 20.0% of all EU's raw milk (163.2 million t) was produced in Germany (32.8 million t). This production volume makes Germany the 5th largest producer after the USA and three of the BRIC countries (Brazil, China, and India). Furthermore, German dairies ranked 6th in fresh milk, 5th in butter, and 2nd in cheese production worldwide (AMI, 2016).

The following sections will give an overview over the German dairy supply chain. First, the structure of the chain and each level directly involved in the transformation of the product from production over processing to marketing will be discussed. In this context also relevant policy changes affecting the structure of each level and the chain will be depicted. Second, the economic interactions between the stages will be briefly portrayed. And last but not least, a summary will merge all subsection to deduce a coherent picture of the entire chain and possible market power distribution along the chain.

2.1 DAIRY SUPPLY CHAIN

The German dairy supply chain with data from the years 2014 and 2015 is schematically illustrated in Figure 1. German dairy farmers almost exclusively deliver their output product raw milk to German dairies. In 2015, 96.3% of the raw milk produced was delivered to dairies operating in Germany. The rest of raw milk is either directly marketed, used for the production of around 0.1 million t of butter or used for own consumption by farmers (AMI, 2016). This is a consequence of delivery obligations, defined in contracts towards their dairy cooperatives or investor-owned dairies (Bundeskartellamt, 2012). Most raw milk is contracted over an average period of 3 years, if delivered to cooperative dairies, and 3 years and 9 months, if delivered to investor-owned dairies (Bundeskartellamt, 2009).

The raw milk is picked up by dairies at the farm gate on a frequency of 1 to 3 days (Bundeskartellamt, 2009). The amount of raw milked delivered to German dairies increased by 18.3% compared to 2003. Furthermore, around 2.1 million t

were imported. Overall, German dairies processed approximately 33.8 million t in 2015. The costs of purchased raw milk accounts for about 55% of the dairies' total costs (Bundeskartellamt, 2012).



Figure 1: Structure of the German dairy supply chain (2014/15).

Source: own illustration based on data from BMLEV, 2015; Bundeskartellamt 2009/2012; MIV, 2016.

Only small amounts of raw milk, 1% to 6% of total production, and other fast perishable, unprocessed dairy products, are traded on spot markets by milk traders and dairies, in most cases to balance short-time oversupply and undersupply of dairies. Milk traders mostly act as an important export agent for dairies (Bundeskartellamt, 2009).

The raw milk is then processed into a diverse number of products, most importantly milk for consumption, fresh milk products (e.g. yoghurt), cheese, long-life milk products (such as milk powder), and butter. The produced products are mainly sold to a small number of retailers (around 37% of ME in 2015), especially discounters, or exported (around 49% of ME in 2015). The remaining 14% of ME are delivered to wholesalers, large consumers, or the food industry.

2.1.1 Raw milk production

In Germany the stock of dairy cows declined by around 2.0% from 2003 to 2015 (see Table 1). However, since the milk output per cow increased by approximately 21.4%, the overall raw milk production grew by around 4.8 million t (+17.4%) (see Table 1). Thus, Germany was the largest producer of raw milk in the European Union in 2015 with approximately 32.7 million t, of which only around 2.3% was produced organically in 2015 (BMLEV, 2006-2015).

	2003	2006	2015	change 2003-2015	in %
total stock of dairy cows (in 1,000)	4,373	4,054	4,285	-182	-2.0%
avg. milk production per cow (in kg)	6,272	6,849	7,616	1,344	21.4%
total milk production (in 1,000 t)	27,874	27,995	32,721	4,847	17.4%
milk delivered to dairies (in 1,000 t)	26,621	26,876	31,495	4,874	18.3%
share of total milk production	95.5%	96,0%	96.3%		0.8%
avg. milk fat content	4.20%	4.16%	4.09%		-0.1%
avg. milk protein content	3.42%	3,40%	3.42%		0.0%
avg. milk price (€/100 kg) at farm gate	31.4	27.4	29.1	-2.3	-7.3%

Table 1: Total stock of dairy cows, average milk production per cow, total milk production, milk delivered to dairies, average fat and protein content of delivered milk, and average milk price at farm gate in selected years.

Source: own illustration based on data from BMLEV, 2006-2015.

A fierce structural change affected German dairy farmers over the analyzed period. The number of dairy farmers dropped by around 38.4% between the years 2003 and 2015. Only 74,800 farms were still producing in 2015, which implies that around 46,700 farms stopped producing since 2003. Furthermore, more and more cows are held on average by a single farm (on average 54.0 in 2015 compared to 36.0 in 2003). The number of dairy farms holding 100 to 199 cows grew approximately by 121.5% between 2003 and 2015. The stock of dairy cows living on farms with less than 50 animals decreased, while the stock held on farms with more than 50 animals increased (see Table 2).

Over the period 2003 to 2015 policy changes occurred that heavily affect raw milk production. Since its introduction in 1984 the EU milk quota system regulated the maximum amount of raw milk a member state was allowed to produce. Each farm was assigned a quota of milk production. However, in 2015 the national raw milk quota system was abolished to liberalize the market. The available quota had been annually increased, from 2006 to 2008 by 1.5% and from 2008 on by 1.0%, to help the dairy sector to adjust slowly to a completely liberalized market (Gerlach et al., 2006; Fahlbusch et al., 2009).

To allow individual firms to expand their production, a milk quota exchange was introduced in the fall of 2000. Prior to this, EU dairy farmers were only permitted to increase their production capabilities on agricultural land that was originally covered by the quota restricting the trade to be rather local if dairy farmers

	dairy farms			share of total			
distribution of stocks	2003	2013	Change	2003	2013	Change	
1 to 9	20,433	9,800	-52.0%	16.8%	12.4%	-4.4%	
10 to 19	27,694	12,700	-54.1%	22.8%	16.1%	-6.7%	
20 to 49	49,666	28,400	-42.8%	40.9%	36.0%	-4.8%	
50 to 99	19,069	19,900	4.4%	15.7%	25.3%	9.6%	
100 to 199	3,160	7,000	121.5%	2.6%	8.9%	6.3%	
more than 200	1,502	2,200	46.5%	1.2%	2.8%	1.6%	
total dairy farms	121,524	78,800	-35.2%	100.0%	100.0%	0.0%	
	dairy cows (in 1,000)			share of total			
distribution of stocks	2003	2013	Change	2003	2013	Change	
1 to 9	109	50.7	-53.5%	2.5%	1.2%	-1.3%	
10 to 19	400.6	186.5	-53.4%	9.2%	4.4%	-4.8%	
20 to 49	1,556.6	921.4	-40.8%	35.6%	21.7%	-13.9%	
50 to 99	1,257.4	1,314.7	4.6%	28.8%	30.9%	2.2%	
100 to 199	406.7	909.7	123.7%	9.3%	21.4%	12.1%	
200 and more	641.7	868.3	35.3%	14.7%	20.4%	5.7%	
total dairy cows	4,372.0	4,251.4	-2.8%	100.0%	100.0%		
avg. dairy cows per farm	36.0	54.0	50.0%				

 Table 2:
 Structure of German dairy cow farming in 2003 and 2013.

Source: own illustration based on data from BMLEV, 2006 and 2015.

wanted to avoid geographic dispersion. Even though the quota exchange lifted this requirement, firms were still restricted to trade quota within their own region. 21 of these smaller dairy regions existed until 2007 in Germany and limited expansion regionally. In 2007, the number of dairy regions was decreased to two, and thus quota could be traded across larger distances. As a consequence, quota was mainly transferred from the states of Baden-Württemberg and Lower Saxony to North Rhine-Westphalia, Mecklenburg-West Pomerania, and Bavaria (Bundeskartellamt, 2009). Bavaria was the largest raw milk producer of the German federal states in 2015 with around 8.1 million t (AMI, 2016).

Figure 2 displays the quota assigned to the German dairy farmers by the EU and the milk, corrected for the actual fat content, delivered to dairies since 2004. While between 2004 and 2008 more milk was delivered to dairies than allowed, the annually increasing quota allowed German farmers to expand production within the growing quota regime until 2013. In the last two years of the quota policy, German farmers were increasing their production vastly (see Figure 2), and large fees had to be paid to the EU, alone 0.3 billion € in 2015 (AMI, 2006-2016).



Figure 2: German milk quota and raw milk delivered to dairies from 2004 to 2015.

Source: own illustration based on data from AMI, 2010-2016.

Many mid-sized dairy farms tried to prepare for the abolishment of the quota by increasing their economies of scale and thus their stock of dairy cows (see Table 2). The number of farms with 50 or more dairy cows grew by more than 20%. Consequently, smaller-sized dairy farms (< 20 dairy cows), which lack capital

or even space to grow further, were forced to shut down their production. More than 50% of these farms quit the dairy business.

Apart from the abolishment of the quota, dairy farmers saw other forms of government intervention in their favor vanish. Gradual reductions in the intervention levels for dairy products, which are now quantity bound, and the expiration of dairy export subsidies were implemented (Bouamra-Mechemache et al., 2008). The intervention prices for butter and skim milk powder (SMP) can be seen as a minimum price for producers at which a government agency clears the market for these products to reduce supply. Since 2003, the raw milk price has never fallen below the intervention level and thus lost its meaning as a raw milk target price. However, in 2009 the intervention regime stabilized prices at the intervention level. For butter the invention price is currently at 2.22 € per kg and for SMP at 1.75 € per kg with a maximum intervention quota of 30.000 t and 109.000 t respectively (Fahlbusch et al., 2011).

Overall, the economic situation for dairy farmers, especially smaller ones, was rather insecure over the period from 2003 to 2015. Figure 3 illustrates the annual average German raw milk price. Drastic drops in the price after peaks, e.g. 2007/2008 and 2013/14, hit small German dairy farmers particularly hard and many had to give up producing. During those brief periods of high prices many small farmers loaned money to increase their production capacities, since they saw only one possibility to survive in a liberalized market setting by growing.



Figure 3: German annual averaged raw milk price from 2003 to 2015.

Source: own illustration based on data from AMI, 2009-2016.

When prices dropped, these farmers lacked financial liquidity and had to declare bankruptcy, further promoting structural change (see Table 2) (Lfl & LEL, 2015; 2016).

Even though prices for organically produced raw milk continued to increase after 2012, while the price for conventionally produced milk declined, switching the production to organic was not seen as a possibility as a result of strict regulations, i.e. in form of a two year transition period, lack of dairies with organic production or the necessary capital (Bundeskartellamt, 2009). In 2015 only 2.3% of all raw milk delivered to German dairies was produced organically (AMI, 2016), and demand for organic dairy products remains marginal (Bundeskartellamt, 2009).

2.1.2 Milk processing

Around 33.8 million t of raw mild were processed by 143 German dairies in 2015. This is an increase of 18.6% compared to 28.5 million t in 2003. As mentioned before, almost all the raw milk was sourced from German dairy farmers (around 93.8%). Nevertheless, raw milk imports from neighboring countries have become more important, since domestic production accounted for 96.5% of dairies' total procurement in 2003.

In 2014, the processing of raw milk into dairy products split as follows (for further details see Table 3):

- 5.0 million t of milk for consumption (-13.6% compared to 2003)
- 2.9 million t of fresh milk products (e.g. yoghurt) (+11.0% compared to 2003)
- 1.4 million t of long-life milk products (e.g. milk powder) (+12.0% compared to 2003)
- 2.5 million t of cheese (+24.7% compared to 2003)
- 0.5 million t of butter and milk fat products (+8.2% compared to 2003)

In four of these five dairy product groups Germany was self-sufficient in 2014 and exported large amounts of these products. Only for butter and other milk fat products Germany was not 100% self-sufficient. However, its self-sufficiency rate was raised to 99.0% in 2014 and grew by 15.4%-points compared to 2003 (see Table 3).

		milk for consumption			fresh milk products (e.g. yoghurt)			
in 1,000 t	2003	2014	change	2003	2014	change		
production farmers	174	125	-49	-28.2%	0	0		
production dairies	5,821	5,030	-791	-13.6%	2,688	2,984	296	11.0%
total production	5,996	5,155	-841	-14.0%	2,688	2,984	296	11.0%
import	27	87	60	222.2%	176	169	-7	-4.0%
import/total produc- tion	0.5%	1.7%		1.2%	6.5%	5.7%		-0.8%
export	840	759	-81	-9.6%	513	746	233	45.4%
export/total produc- tion	14.0%	14.7%		0.7%	19.1%	25.0%		5.9%
consumption	5,183	4,493	-690	-13.3%	2,351	2,407	56	2.4%
self-sufficiency rate	115.7%	111.9%		-3.8%	114.3%	124.0%		9.7%
	long-life r	nilk produc	ts (e.g. mill	(powder)	cheese			
in 1,000 t	2003	2014	change	2003	2014	change		
total production	1,245	1,395	150	12.0%	1,984	2,475	491	24.7%
import	236	390	154	65.3%	576	727	151	26.2%
import/total produc- tion	19.0%	28.0%		9.0%	29.0%	29.4%		0.4%
export	761	1,216	455	59.8%	701	1,188	487	69.5%
export/total produc- tion	61.1%	87.2%		26.1%	35.3%	48.0%		12.7%
stock change	-45	39			69	22		
consumption	765	530	-235	-30.7%	1,790	1,992	202	11.3%
self-sufficiency rate	162.7%	263.2%		101.5%	110.8%	124,2%		13.4%
butter and milk fat products								
in 1,000 t	2003	2014	change					
production farmers	1	0.1	-0.9	-90.0%				
production dairies	452	490	38	8.4%				
total production	453	490	37	8.2%				
import	168	148	-20	-11.9%				
import/total produc- tion	37.1%	30.2%		-6.9%				
export	89	149	60	67.4%				
export/total produc- tion	19.6%	30.4%		11.8%				
changes in stock	-10	-5						
consumption	542	495	-47	-8.7%				
self-sufficiency rate	83.6%	99.0%		15.4%				

Table 3: Total German production of dairy products, import and export, consumption and self-sufficiency rate in selected years.

Source: own illustration based on data from BMLEV, 2005 and 2015.

While the output in general increased, the domestic consumption decreased. Except for cheese products (+11.3%) and fresh milk products (+2.4%), which experienced small level of growth in consumption, smaller amounts of milk (-13.3%), long-life milk products (-30.7%), and butter and milk fat products (-8.7%) were consumed (see Table 3).

Due to the growth in production more dairy products had to be exported. Overall, exports grew drastically compared to 2003: exports of fresh milk products by 45.4%, exports of long-life milk products by 59.8%, exports of cheese by 69.5%, and exports of butter and milk fat products by 67.4%. Only exports of milk for consumption declined by 9.6%, however, relatively less than the overall production. Consequently a larger share of domestic fresh milk production was exported in 2014 than 2003. Imports of some dairy product groups also grew tremendously percentagewise, but in all Germany evolved as a net exporter of dairy products (see Table 3).

149 dairies, with more than 50 employees, operated in Germany in 2014 (see Table 4). This is a decline of around 39.7% compared to 2003. The dairies earned approximately 26.1 billion \notin of revenue, a little bit more than ¼ of it outside of Germany. The total revenue increased by around 27.0%, while the number of employees increased by 12.5%. Thus the average revenue per firm (+110.6%) and per employee (+36.8%) increased considerably from 2003 to 2014.

•				
	2003	2014	Change	Change in %
dairies ¹	247	149	-98	39.7%
total number of employees	34,238	38,532	4,294	12.5%
total revenue (in million €)	20,555.8	26,104.0	5,548.2	27.0%
domestic revenue (in million €)	16,879.5	19,067.9	2,188.4	13.0%
revenue on external markets (in million ${\mathfrak {C}}$)	3,677.3	7,036.1	3,359.8	91.3%
avg. revenue per dairy (in 1,000 €)	83,170	175,195	92,025	110.6%
avg. revenue per employee (in 1,000 €)	600	821	221	36,8%

 Table 4:
 Number of German dairies, total dairy employees, (domestic and external) revenue of German dairies, and average revenue per dairy and employee in selected years.

Source: own illustration based on data from BMLEV, 2005 and 2016.

Notes: ¹ With more than 50 employees.

The structure of German milk processing is dominated by large dairies. In 2012, 25 dairies processed annually more than 300,000 t of raw milk (see Table 5). In total, these 25 firms processed more than 70% of the entire milk processed in Germany. Their share of total milk processed increased by around 8.2%-points between 2003 and 2012. In the meantime, around 17.7% of German dairies processed less than 5,000 t per year, but accounted only for 0.2% of German overall milk production. 96.1% of all German raw milk was processed in 2014 by firms with more than 50,000 t annual processing capacities, in total 78 dairies.

		dairies		:	share of tota	I
processed milk in 1,000 t	2003	2012	Change	2003	2012	Change
less than 5	47	26	-44.7%	24.1%	17.7%	-6.4%
5 to <50	35	43	22.9%	17.9%	29.3%	11.3%
50 to <300	84	53	-36.9%	43.1%	36.1%	-7.0%
300 and more	29	25	-13.8%	14.9%	17.0%	2.1%
total	195	147	-24.6%	100.0%	100.0%	
	amou	nt of milk (in 1,0)00 t)	!	share of tota	I
processed milk in 1,000 t	2003	2012	Change	2003	2012	Change
less than 5	66	53	-19.4%	0.2%	0.2%	0.0%
5 to <50	1,594	1,095	-31.3%	4.3%	3.7%	-0.6%
50 to <300	12,399	7,585	-38.8%	33.1%	25.5%	-7.6%
300 and more	23,400	21,021	-10.2%	62.5%	70.6%	8.2%
total	37,459	29,754	-20.6%	100.0%	100.0%	

Table 5: Structure of German milk processing in 2003 and 2012.

Source: own illustration based on data from BMLEV, 2003 and 2015.

The concentration of the German dairy industry can further be observed in the list of the TOP 10 German dairy companies (regarding revenue) in 2015 (see Table 6). Four of them are cooperatives, one being a Danish-Swedish cooperative (Arla Foods). Overall, these 10 companies accounted for approximately 11.0 billion € of domestic revenue, around 41%. Table 6 also illustrates that the TOP 10 dairies earn less than 50% of their revenues domestically and their main income

source are foreign markets. Thus the export business on external markets was becoming more essential to large dairy companies' performance.

	domestic revenue	total revenue
DMK Deutsches Milchkontor	3.18	5.30
Unternehmensgruppe Theo Müller	1.60	3.70
Arla Foods	1.55	10.61
FrieslandCampina	1.37	11.35
Hochwald Foods	0.68	1.59
Bayernland	0.63	0.90
Hochland	0.58	1.28
Savencia	0.50	4.60
Meggle	0.47	1.00
Zott	0.43	0.96

Table 6: Revenue of the TOP 10 German dairy companies in 2015.

Source: own illustration based on data from Lfl & LEL, 2015 and 2016.

Overall, concentration has increased in the German dairy industry through mergers and acquisitions tremendously from 2003 to 2015. However, in comparison to some other food processing industries the share of the ten largest dairy enterprises in the sector's total revenue still remains rather low. While in poultry meat processing the ten largest firms account for 70.6% of the total revenue, the ten largest dairies accumulated 43.3% of total revenues in 2012. Nevertheless, in meat processing in general, pastry production, and the overall food industry the TOP 10's share is significantly smaller than in the dairy industry with 25.8%, 22.3%, and 12.3% respectively. Thus the dairy industry's concentration can be seen as a concern (BMLE; 2016).

Even though the domestic market for dairy products grew only slightly or in some product groups even decreased, dairy companies increased their processing capacities and thus their demand for raw milk. The demand for raw milk actually outpaced the growth in production, which was enabled by the annually increasing quota, and was covered by increasing raw milk imports. The growing possibility to market the extra production through exports gave the German dairy industry a window of growth.
2.1.3 Marketing

In 2015, the output of the dairies was primarily exported (49% milk equivalent) or sold to retailers (37%), and other larger consumers (14%) including wholesalers, large costumers, and the food industry for further processing (MIV, 2016). Even though a large share of the processed milk was exported, exports only accounted for approximately 28.0% of dairies' revenue. More than half of the revenue from exports was generated with cheese products, followed by long-life dairy products with around 31.3%. While cheese is almost exclusively sold to EU member countries, more than 90%, around 40% of long-life dairy products were marketed to a diverse number of non-EU countries, primarily located in the Middle and Far East (AMI, 2016). Nevertheless, only 17.6% of export revenue from dairy products was earned outside the Common Market (BVE, 2016).

2.1.3.1 Retailers

The main income for the German dairy industry is earned domestically. The primary outlets on the domestic markets are retailers with approximately 76.3% of all milk equivalent marketed in Germany. The structure of the German food distribution sector is characterized by a dominance of small-to-medium sized stores. Table 7 substantiates a dominance of small stores measured by square meter per shop. In 2014 small shops with less than 200 m², in particular discounters, were the dominant form of food distribution in Germany. As a counterpoint, 874 large retailers with more than 5,000 m² were counted in 2014. In German

	number of stores		revenue in € billion		revenue per store in € million	
	2003	2014	2003	2014	2003	2014
retailers >5000m ²	2,688	874	32.9	18.9	12.24	21.62
discounters	14,214	16,195	50.1	66.9	3.52	4.13
supermarkets	8,620	11,855	29.7	55.5	3.45	46.86
other food stores	37,350	9,600	13.8	4.9	0.37	0.51
overall	62,872	38,525	126.5	146.2	2.01	29.75

Table 7: Structure of German food retailing in 2003 and 2014.

Source: own illustration based on data from BMLEV, 2006-2015.

national statistics discounters accounted for 16,195 stores (no breakdown by size is available) (BMLEV, 2015).

The highest revenue per store (21.6 million €) in food retailing was generated by large retailers in 2014. However, their overall revenue declined by 14 billion € between 2003 and 2014. Discounters generated 66.9 billion €, around 45.7% of the total food retailing revenue, but in comparison to large retailers and supermarkets with a fairly low revenue per shop of 4.1 million €. Table 7 displays that other and in most cases smaller shops accounted for a large number of stores, but their overall revenue was marginal (3.3% of total revenue) and they earned only 0.5 million € per shop in 2014 (BMLEV, 2006-2015).

Additionally, discounters in Germany are expanding. In 2004 discounters opened 401 new stores which were 2.8% more than in 2003. Between the years 2003 and 2014 the number of discounter stores increased by about 13.9%. While large retailers are becoming fewer but earn more on average per unit, smaller retailers who do not co-operate to achieve scale effects have competition disadvantages and are less economically viable. Alone between 2003 and 2004 7.8% of the small retailers (<100m²) exited the food retailing market (BVE, 2016; BVE, 2007).Despite the dominance of small units, food retailing in Germany is heavily concentrated. The five largest food retailers achieved a market share of 72.3% of 191 billion € total revenue in grocery sales in 2015 (BVE, 2016). A series of mergers and acquisitions, e.g. the Edeka Group bought the discounter Plus in 2007, resulted in this fast growing and high concentration ratio. The five largest food retailers in Germany were the Edeka Group (25.3% of food revenue in 2014), Rewe Group (15.0%), Schwarz Group (14.7%), Aldi Group (11.9%), and Metro Group (5.4%) in 2015 (BVE, 2016; BVE, 2007).

2.1.3.2 Consumers

Most consumers buy dairy products in discounters (see Table 8). Around half of the amount of dairy products was sold to consumers via discounters in 2014. While larger retailers account for a bit less than 1/5 of total dairy sales in the individual product categories, supermarkets market more than ¼. Less than 5% of dairy products were sold in other establishments. The shopping habits of German consumers didn't change much between 2003 and 2014. Discounters and supermarkets increased their shares of total sales slightly, while larger retailers lost some.

The overall consumption of dairy products in Germany decreased between 2003 and 2014 (see Table 9). Germans consumed on average around 127.6 kg of dairy

	•			
	discounters	retailers >5000m ²	supermarkets	other establishments
cheese	53.1%	17.2%	25.2%	4.5%
butter	49.0%	20.3%	28.1%	2.5%
milk	53.1%	18.0%	26.0%	2.9%
yoghurt	51.3%	18.4%	27.9%	2.4%

Table 8: Share of total sales for selected dairy products for discounters, larger retailers, supermarkets and other establishments in 2014.

Source: own illustration based on data from BMLEV, 2015.

products in 2014 (a decrease of 17.4 kg compared 2003). Some dairy products are more and more popular among consumers, e.g. cheese (see Table 9). Other products, such as buttermilk (-31.3%), cream (-21.6%), and butter (-13.6%) were consumed less than in 2003.

kg per capita	2003	2014	Change 2003-2014	
milk	65.1	53.2	-11.9	-18.3%
butter milk	1.6	1.1	-0.5	-31.3%
mixed milk products	12.0	12.9	0.9	7.5%
yoghurt	17.6	17.0	-0.6	-3.4%
cream	7.4	5.8	-1.6	-21.6%
other fresh milk products	4.8	1.6	-3.2	-66.7%
cheese	21.3	24.2	2.9	13.6%
butter	6.6	5.7	-0.9	-13.6%
long-life milk products	8.6	6.1	-2.5	-29.1%

Table 9: Per capita consumption of selected dairy products in Germany in 2003 and 2014.

Source: own illustration based on data from BMLEV, 2015.

Figure 4 presents the consumer prices for fresh milk, non-blended butter and Gouda from 2003 to 2015. While the price for Gouda almost steadily increased over the period, the price of butter fluctuated heavily. In 2006, consumers paid less than $80 \in \text{cents}$; in 2013 it was more than $1.10 \in \text{.}$ The price for fresh milk did



Figure 4: German consumer prices (in €) for fresh milk, non-blended butter and Gouda cheese in selected years.

Source: own illustration based on data from BMLEV, 2005-2015.

not fluctuate heavily, but increased by about 15 € cents between 2003 and 2014. However, in 2012 consumers only paid 6.9% more for fresh milk than in 2003.

2.2 ECONOMIC RELATIONSHIPS

After describing the structure of dairy farmers, milk processors, and food retailers in the previous section, the following section depicts existing forms of vertical and horizontal economic relationships and issues arising from these.

2.2.1 Vertical relationships

The bargaining power of dairy farmers towards dairies is seen as relatively low. Because of recent concentration processes at the processing level, often only a small number of large-scale dairies or even only one dairy exist in the region - a radius of around 200 km due to transport costs and the perishableness of raw milk - and are the only buyers for raw milk, while there exist hundreds of small to medium-sized dairy farmers (Bundeskartellamt, 2009). Furthermore, long-lasting contracts with long cancelation periods of up to 12 months and the low number of dairies in the region limit the ability to change processing partners and thus strengthen the position of dairies. After ending contractual relationships with one dairy in the region, dairy farmers reported to the German anti-trust agency that others would refuse to accept them as suppliers (Bundeskartellamt, 2009). To determine the price of raw milk most commonly investor-owned, but also more and more cooperatives, German dairies observe the prices of surrounding dairies and pay an average of this to dairy farmers, who due to their negligible size have to act as price takers (Bundeskartellamt, 2012). However, a series of extra payment can be added to the standard price, e.g. depending on the fat and protein content of the delivered milk or in form of a loyalty bonus. Payoffs are carried out on a monthly basis (Bundeskartellamt, 2009).

The payment system of cooperatives is slightly different. Cooperatives determine the raw milk price after a purchaser for the final dairy product is found and the dairy product price determined. Not having the price of raw milk fixed, gives cooperatives less incentives to include the raw milk price as a cost in their negotiations with possible buyers. Thus, it can be argued that cooperatives are less eager to achieve a high price, but rather try to sell all products. In addition to the later determined raw milk price, cooperatives pay their members a bonus at the end of the year, which consist of the entire profits earned (Bundeskartellamt, 2009).

Dairy cooperatives, who are owned by their members, often act as vertical integrated enterprises when they negotiate with the retailing sector. However, most of the cooperatives consist of up to 15% of inactive members. Furthermore, the still large numbers of active members, who have diverse production cost structures, limit the influence that a single small-scaled farmer might have on the board of the dairy and thus on the milk prices paid out. Consequently, many dairy cooperatives have distanced themselves from their active members and optimize the dairies economic performance, thus acting more and more as private companies with the raw milk price being solely a cost variable rather than the income of its members (Bundeskartellamt, 2009).

Due to EU policy changes that promote market liberalization, intrinsic organizational problems to adapt to these, e.g. lack of capital to boost innovations, and driving cost-minimizing strategies, dairy cooperatives are often not able to promote their members' interest in receiving a high raw milk price, but instead fail to pay higher prices than their private competition (Bundeskartellamt, 2009; Fahlbusch, 2013; Grau et al., 2015).

Even though, from a national perspective the dairy stage is heavily concentrated with the six largest dairies accounting for 50% of market share, four of which are cooperatives (Loy et al., 2015), and we could expect an oligopoly, pricing in the German dairy supply chain is mostly characterized by 'top down' negotiations. The dairies negotiate with a very small number of retailers, most often

discounters, which purchase the largest share of dairy products, namely the Aldi Group, Schwarz Group, Rewe Group, and Edeka Group. Regarding the product group milk for consumption the Aldi Group, Rewe Group, and Edeka Group cover around 50% to 65% of all domestic demand. Through establishment of procurement alliances between larger retailers and smaller retailers with market shares of less than 5% as well as the fact that retailers manage dairies' milk exports, of which 50% is marketed internationally by only two retailers, at most six retailers market more than 90% of the German dairy industry's milk for consumption output (Bundeskartellamt 2014).

In general, Fahlbusch et al. (2009) and others (e.g. Hellberg-Bahr et al., 2010) stated that large shares of German dairy product exports are actually marketed by German retailers. In their EU campaign, they offer so-called 'Euro contracts' that allow them to procure vast quantities of dairy products, in particular less perishable ones like cheese, through central buying and distribute them to their European subsidiaries.

Regarding contractual arrangements the negotiations with the Aldi Group are often seen as signal negotiations for the rest of the sector, since the Aldi Group in most cases offers the lowest prices to consumers for a variety of basic dairy products (Bundeskartellamt, 2009). Negotiations, in particular have become more frequent. While standard products are negotiated two or more times, e.g. the prices for butter are determined on a monthly basis, brand product contracts are arranged annually (Fahlbusch et al., 2009). Competitors are able to respond quickly to price changes of the discounters and their everyday low pricing strategies due to high information transparency provided by market research institutions, which makes collusion unnecessary (Bundeskartellamt, 2009).

Negotiated contracts are often so large that several dairies have to form production alliances to fulfil the quantity requirement. Through public calls for bids and dairies undercutting each other's offers, retailers are well informed on their suppliers' production costs and consequently marginal costs, and can force dairies more easily to provide the product at these (Fahlbusch et al., 2011). In recent years, private labels of retailers emerged, which allow retailers to have full control over the cost pass-through process and bind dairy companies more tightly to them (Loy et al., 2015). The Aldi Group and Schwarz Group sum up more than 60% in private label cheese and 50% in milk products. All this, high transparency of information and the dominant position of retailers, might lead to fierce procurement strategies and oligopsonistic conduct (Loy et al., 2015).

2.2.2 Horizontal relationships

To increase their bargaining power in negotiations with dairies, German dairy farmers are allowed to form producer collectives and even collaborations of producer collectives. In 2014, around 143 dairy producer collectives existed, but only 2 collaborations (BMLEV, 2000-2012). The largest producer collective was in 2009 Bayern-MeG with around 9,000 members and a milk production of 1.4 million t (Bundeskartellamt, 2009).

More and more dairies form co-operations to use their facilities more efficiently. Examples of cooperation among dairies in Germany are the following: they share the investment risk for new facilities and technologies, they exchange already delivered raw milk and semi-processed goods to optimize the usage of their processing facilities and costs, and they exchange milk delivery contracts to optimize the supply and transport costs. However, co-operations of dairies that constrain the economic competition for raw milk supply or actual divide regional markets among them are prohibited (Bundeskartellamt, 2012). The merger of Nordmilch AG and Humana Milchunion eG in 2011 preceded a joint-venture of the firms in 2005, the wheyco GmbH, a firm that produced specialized dairy products (Grau et al., 2015).

2.2.3 Anti-trust investigations

With the concentration and consolidation processes at the processing and retail level as well as the vertical relationships in mind, it is not surprising to hear that since the start of the 2000s the German anti-trust agency has received a growing number of complaints by dairy farmers and dairies on the procurement behavior of downstream buyers, respectively the dairy industry and retail sector (Bundeskartellamt, 2009).

Apart from the before mentioned issues regarding changing the buyer, dairy farmers have been criticizing that dairies asymmetrically transmit prices, price decreases on the dairy output market are faster transmitted to them than price increases. However, dairies themselves have been accusing retailers to abuse their bargaining position to lower the prices for dairy products by colluding. As a result an official investigation was started by the anti-trust agency in 2008 with its final report published in 2012. The German anti-trust agency stated that even though procurement prices are low and the structure along the chain seems to favor the position of buyers, no evidence on the abuse of market power was found (Bundeskartellamt, 2009; 2012).

2.3 SUMMARY

The German dairy supply chain is one of the, if not the most important agri-food supply chain in Germany. Even though, restricted by the milk quota and a saturated domestic market for dairy products, the chain has boosted production at every level since the early 2000s. Primary destination for the surplus is the Common Market and in particular EU neighbor states. Nevertheless, the home market is still the dominant outlet for German dairy products.

The growth in production was accompanied by heavy structural change and concentration processes at every level of the supply chain. While dairy farmers declined drastically in number, around 1/3, and grew tremendously on average in size, by 50%, they remain atomistic with around 74,800 dairy farms still being active in raw milk production. The processing level experienced a similar decline percentagewise and is showing alarming levels of concentration with six dairies dividing up 50% of the raw milk market. Nevertheless, through a series of mergers and acquisitions the number of possible buyers at the retail level has declined to about five, which control almost the entire dairy product supply via procurement alliances and central buying for their European campaign. High levels of information transparency allow retailers to harmonize the procurement and prices of dairy products without the necessity of collusion.

Even though, the dairy sector is characterized by cooperative action, four of the six largest dairies being cooperatives, farmers hardly benefit in form of higher raw milk prices. Due to cost-minimizing strategies and production of standard dairy products, these cooperatives are not able to pay higher prices for the milk delivered by their members than the private competition, which earn higher margins with their brand products. Facing a highly concentrated retail sector with good knowledge of the dairies' production cost, it is hard to imagine that dairy cooperatives act as something else than price takers on the dairy output market.

The structure and vertical relationship give suspicion to the subsequent abuse of buyers' market power. Fueled by complaints of dairy farmers and dairies on non-competitive behavior of downstream supply chain agents, the German antitrust agency conducted a sector analysis between 2008 and 2012. In its final report the anti-trust agency acknowledged the threat of oligopsony power at both markets, raw milk and dairy output market, but could not find any evidence. Thus, the main premise of the presented work is to empirically investigate the German dairy supply chain for the existence of oligopsony power at the raw milk and dairy product market.

3 THEORY REVIEW

Throughout the past 100 years several approaches with the aim to analyze market power emerged and partially submerged again. In the following chapter we will present the most notable approaches: the structure conduct performance paradigm (SCPP), the new empirical industrial organization (NEIO) approach, and the price transmission framework. Furthermore, the strengths and weaknesses of any method will be separately discussed. In a concluding summary, the approaches are contrasted to each other and the appropriateness of methods for the field of study is discussed.

3.1 STRUCTURE CONDUCT PERFORMANCE PARADIGM (SCPP)

3.1.1 Methodology

The structure conduct performance paradigm (SCPP) was first introduced by Mason (1939; 1949) more as a case study approach and later augmented by Bain (1951; 1968) through the use of statistical methods (Stiegert et al., 2009). The primary idea of the SCPP is to explain the performance of an industry on the basis of its market structure. It is assumed that market structure causes firms' conduct, which influences the performance of the industry. Consequently, structural variables, such as concentration ratios, are used as explanatory variables for performance, e.g. price-cost margins (PCM) or profits, derived from accounting data. With this thought in mind it appeared now easily possible to explain cross-sectional differences among industries in performance with a relatively low number of structural variables (Bresnahan, 1989).

In the first two decades after the establishment of the SCPP several different measures of market structure, in particular concentration ratios, were tested to explain intra- and inter-industry performance variety (Bresnahan, 1989). Stigler (1964) used the Herfindahl index to explain the rate of return on capital, rate of return on net worth, and ratio of market value to book value. The finding was that the more firms exist the greater the competition is. Later on, Miller (1967) and Kilpatrick (1967), using concentration ratio measures, discovered that the four firm concentration ratio tends to explain the performance of an industry more than higher numbered concentration ratios. Miller even found a negative impact of the eight firm concentration ratios on industry's profit, thus counterbalancing the effect of the four largest firms (Miller, 1967). Collins and Preston (1969) also stated that higher PCMs in industries with a higher four firm concentration ratio exist. Analyzing 94 industries and using data for the years 1958,

1963, and 1968, Cowling and Waterson (1976) emphasized that changes in the Herfindahl index are associated with changes in the PCMs.

In general, the SCPP studies found a positive relationship between firm concentration and performance in industries, mainly PCMs, what motivated anti-trust agencies around the world to take an offensive stand against mergers, predatory pricing, and resale price maintenance, throughout the 1960s and 1970s (Stiegert et al., 2009).

3.1.2 Criticism

Following the primary successes of the SCPP, criticism regarding its main assumptions emerged. While the SCPP assumes an unidirectional link between market structure and industry conduct, a series of studies argued and proved that industry conduct and market structure actually mutually influence each other (Marion et al., 1979: Martin, 1979; Clarke & Davies, 1982; Schaffer, 1982). Apart from the theoretical concerns, the endogeneity of conduct can lead to issues in the empirical application. When the exogeneity assumption does not hold, the ordinary least squares (OLS) regression estimates of the reduced-form between structural and performance variables are biased, since the structural variables are correlated with the error term and inference based on these estimates is invalid (Bresnahan, 1989; Kadiyali et al., 2001).

Confronted with the criticism, the SCPP has been altered to overcome partially its original limitations. Martin (1979) focuses on the endogeneity problem between profitability, seller concentration, and advertising demand. The methodology comprises three equations with PCM, concentration ratio, and advertising-sales ratio as endogenous variables that depend on each other and other structural, performance, and explanatory variables, including lags of the endogenous variables. For 209 industries in the year 1967, the author reports that conduct is determined by market structure and performance, higher advertising intensity in concentrated and more profitable industries, and that concentration follows dynamic adjustment processes (Martin, 1979). Another, more recent example is the study of the U.S. food manufacturing sector by Stiegert et al. (2009). By applying simultaneous estimations technique to the approach of Martin (1979), a positive relationship between industry structure and profits is identified. However, it is further elaborated that the differences between industries in performance can not only be explained by the market structure, but also other determinants, such as the regionalism of the market, and have to be accounted for (Stiegert et al., 2009).

Apart from the exogeneity issue, the SCPP was also heavily criticized for its use of accounting data and the choice of concentration ratio to represent market structure. Many authors applying SCPP used accounting data to calculate PCMs. Since economic marginal costs (MC) are not observable from accounting data, often average cost (AC) across products was used as a proxy, however with only limited explanatory power (Perloff et al., 2007). Fisher and McGowan (1983) criticized that in general accounting data poorly represent economic variables due to practices of the accounting department of firms like 'creative accounting', delayed reporting, and limited disclosure statements of firms (Blanck & Bahrs, 2011).

Concentration ratios do not only neglect the importance of imports and exports for an industry, but often overstate the level of competition when a closely related industry exists. Furthermore, most concentration ratios are calculated at the national level and disregard firm concentration on local markets and thus underestimate concentration (Perloff et al., 2007). In addition, the cross-sectional character of many SCPP studies has been criticized. Bresnahan (1989) and Kadiyali et al. (2001) doubt that comparison among industries with highly idiosyncratic characteristics are valid on the basis of a small number of structural variables and rather the distinctive characteristics cause the industries' outcome.

Due to the critic and hence dissatisfaction, the SCPP, even though still applied today, see for example Stiegert et al. (2009) and Bhuyan (2014), has lost significance, and other approaches have emerged. In particular, NEIO can be regarded as an answer to the shortcomings of the SCPP.

3.2 New EMPIRICAL INDUSTRIAL ORGANIZATION (NEIO)

NEIO emerged in the late 1970s out of dissatisfaction with the SCPP. The new approach addresses the main issues of SCPP, non-observable economic MC, endogeneity of variables, and cross-sectional comparison. Advances in game theory improved the understanding of firms' conduct and allowed to treat it as an unknown parameter. Unfortunately, frequently no unique equilibrium solution exists for game theoretic models, but rather multiples. As long as firms consider future benefits strongly for their decision process in a repeated game, a countless number of subgame perfect Nash equilibria exist. NEIO approaches circumvent the issue in most cases by focusing on a single industry and thereby restricting the possible outcome set (Kreps, 1990; Sperling, 2002). The focus on the single industry case is opposed to the cross-sectional character of SCPP studies.

The invisibility of economic MC is dealt with by NEIO approaches either by estimating firms' conduct, i.e. market power, without measuring costs at all or by deriving a measure of MC from firm behavior. In general, the market is modeled with demand and supply equations, deduced from firms' profit, production, and cost functions. Due to the endogeneity of prices and quantities in this setup, most NEIO studies rely on simultaneous estimation techniques (Bresnahan, 1989). The two most applied NEIO approaches will be discussed in detail in the following sub chapter.

3.2.1 Methodology

Bresnahan (1982) and Appelbaum (1979; 1982) presented the most common applied NEIO methodologies, the general identification methodology (GIM) and the production-theoretic approach (PTA) in their seminal works respectively. Price and quantity are determined by the intersection of supply relation and demand function in a market. In a perfectly competitive market the price is determined by the MC of the marginal producer, thus *price* (*P*) = *MC*. In the case of a non-competitive market setting, where sellers have market power, suppliers will equate their 'perceived' marginal revenue (MR) to MC to maximize their profits. A monopoly pricing behavior is present when the 'perceived' MR = MR, and competition prevails when 'perceived' MR = P. In cases of 'perceived' MR < P some form of oligopoly pricing exists (Bresnahan, 1982).

In a general NEIO oligopoly model based on Bresnahan (1982) with a homogenous product, demand and supply relations have to be formulated and derived. The demand of buyers typically takes the form,

$$Q = D(P, Y) \tag{2.1}$$

where Q is the quantity, P the price, and Y a vector of exogenous demand shifter. In case of oligopoly behavior and quantity as the strategic variable the supply relation of the profit maximizing individual firm i takes the form,

$$P(Q,Y) = \frac{\partial c_i(q_i,W)}{\partial q_i} - \varphi_i \frac{\partial P(Q,Y)}{\partial Q} q_i$$
(2.2)

where $P(\bullet)$ is the inverse industry demand, $C(\bullet)$ is the cost function, q_i the individual firm's supply, and W an exogenous cost shifter. The variable $\varphi_i = \partial Q/\partial q_i$ is an index of firm behavior. It can either measure the firm *i*'s conjectural variation (CV), the belief of a firm on how output changes affect industry output, or the wedge between price and MC (Bresnahan, 1989; Sperling, 2002). Nevertheless, φ_i quantifies firm *i*'s ability to mark-up the price from the competitive level and

comprises within itself different levels of competition. If φ_i takes a value of zero, price equals MC and perfect competition prevails. In case of a value of one, a monopoly exists. Values between zero and one represent different intensities of oligopoly with the special case $\varphi_i = 1/n$ being Cournot competition (Bresnahan, 1989). The methodology that estimates φ_i as a continuous variable ranging from zero to one was titled by Corts (1999) as the conduct parameter method (CPM). We therefore stick to this terminology and consequently refer to φ_i as the conduct parameter.

Given (2.2), the identification of φ_i is not straight forward possible as it appears to be. The conduct parameter and the slope of the inverse industry demand, $\partial P(\bullet)/\partial Q$, form a compound parameter and the effects of the individual terms cannot be separated without additional information. The necessary information is normally obtained by estimating the demand and supply equation simultaneously, what quantifies the slope of the demand equation $\partial Q(\bullet)/\partial P$. However, the demand equation has to contain an exogenous variable that through an interaction term with the price is able to alter the demand slope in face of an exogenous change. Otherwise, in case of non-constant MC, it might not be possible to distinguish between perfect competition and monopoly (Bresnahan, 1982; Lau, 1982).

Appelbaum's (1982) seminal work is similar to the isochronal work of Bresnahan (1982) and models a homogenous product market with quantity setting firms. However, after defining the demand function that the industry faces (2.1), input demand for the i^{th} firm are derived by applying Shepard's lemma (Shepard, 1970) to its cost function. In the next step a slight alternative formulation of Bresnahan's supply relation (2.2) is deduced by taking the first-order condition with respect to the output quantity that maximizes profits,

$$P(Q,Y) = \frac{\partial c_i(q_i,W)}{\partial q_i} - \frac{\theta_i}{\varepsilon} P(Q,Y)$$
(2.3)

where $\theta_i = (\partial Q/\partial q_i)(q_i/Q)$ is the conjectural elasticity of total industry with respect to the firm *i*'s output, and $\varepsilon = (\partial Q/\partial P)(P/Q)$ is the market demand elasticity. The conjectural elasticity (CE) θ_i is a compound of the conjectural variation and the firm *i*'s output share. Thus, it nests the same market structures perfect competition to monopoly in the value range 0 to 1 (Appelbaum, 1982). To account for the lack of firm-level data, Appelbaum (1982) aggregates across all firms' cost functions by using a Gorman polar form, which allows firms to have individual, but parallel cost functions and thus marginal cost be equal across firms. Through this industry input demand functions are obtained, whose slopes enter the supply relations (2.3) via the marginal cost term and enable to restrict parameters across the system of equations. The system consists of the demand equation (2.1), the supply relations (2.3), and industry input demand functions. The advantage is that more efficient parameter estimates are obtained (Sperling, 2002).

Appelbaum (1982) argued further that since MC are the same, the 'perceived' MR must be the same as well. Following this thought of line, firms equal their constant MC to the 'perceived' MR in equilibrium, consequently also the conjectural elasticity must be the same across firms in equilibrium, thus $\theta_i = \theta$ is the equilibrium value (Appelbaum, 1982).

With Θ being the equilibrium value it is now possible to compute a popular measure for monopoly power in the market, the Lerner index *L* (Lerner, 1934). By subtracting MC from both sides in (2.3) and dividing through the price, the Lerner index is obtained,

$$L = \frac{P - MC}{P} = -\frac{\theta}{\varepsilon}$$
(2.4)

where the Lerner index, the relative difference between industry price and MC, equals the negative elasticity adjusted conduct parameter. Consequently, it lays in the range $0 \le L \le -1/\varepsilon$, where zero stands again for perfect competition and $-1/\varepsilon$ would be the relative monopoly price mark-up. The derived formulation allows interpreting the conduct parameter as an elasticity adjusted Lerner index, but in contrast to the classical Lerner index it is able to distinguish between markets with high margins due to inelastic demand or due to uncompetitive behavior (Appelbaum, 1982; Corts, 1999).

While Bresnahan (1982) presented a purely theoretic model and left it to other researchers to apply it empirically, Appelbaum (1982) applied his approach to the U.S. textile and the U.S. tobacco industry for the period from 1947 to 1971 using annual data. While he found rather low levels of oligopolistic behavior by the textile industry (0.0368), the tobacco industry acted rather oligopsonistic (0.4019) during this period (Appelbaum, 1982).

In this section two theoretical approaches for the identification of oligopoly power were presented. The next section presents alteration of these concepts to allow for or identify also oligopsonistic market behavior.

3.2.2 Oligopsony model

The seminal works of Appelbaum (1979; 1982) and Bresnahan (1982) both model the oligopoly case. Nevertheless, due to certain characteristics described in

the introduction, agricultural supply chains rather seem to be threatened by oligopsonistic market conduct. Schroeter (1988) was the first to model oligopsony, based on the PTA method by Appelbaum (1982), and apply his model to an agricultural supply chain with input and output markets. Similar to Appelbaum (1982) the i^{th} firm's input demand for j non-material inputs, given a fixed output level q_i and assumed to be procured in competitive factor markets, are derived by applying Sheppard's lemma to the firm's non-material input cost function,

$$x_i^z = \frac{\partial c_i(q_i, W^z)}{\partial W^z}$$
(2.5)

where x_i^z is a vector of the quantities and W^z is a vector of prices for the *j* nonmaterial inputs. The function for aggregated demand of the output is according to (2.1), while the aggregated supply of the material input is given by,

$$Q = S(W^M, Z) \tag{2.6}$$

where Q is the quantity of material input supply, W^{M} price of the material input, and Z a vector of exogenous supply shifters. Since fixed proportion technology is assumed, aggregated supply of the material input and aggregated demand for the output are demarked with the same symbol Q. Given this assumption and (2.5), the profit function of the i^{th} firm can be denoted as,

$$\pi_{i} = \underbrace{P(Q, Y)q_{i}}_{Revenue} - \underbrace{\left(W^{M}(Q, Z)q_{i} + c_{i}^{Z}(q_{i}, W^{Z})\right)}_{Total \ Cost}$$
(2.7)

where the profit of i^{th} firm π_i equals the revenue minus total cost. Maximizing its profit to the chosen level of output, and through the fixed proportions assumption to the level of material input, the first-order condition takes a similar form like in Appelbaum (1982),

$$P(Q,Y)\left(1+\frac{\theta_i}{\varepsilon}\right) = W^M(Q,Z)\left(1+\frac{\theta_i}{\eta}\right) + \frac{\partial c_i(q_i,W^Z)}{\partial q_i}$$
(2.8)

where ε has the same definition as market demand elasticity as in (2.3), $\eta = (\partial S(W^M, Z) / \partial W^M)(W^M / Q)$ is the elasticity of material input supply, and θ_i is here the *i*th firm's conjectural elasticity of output but also of material input and due to the fixed proportions assumption takes the same value on the material input as well as output market. However, since the relative mark-up $(-\theta_i / \varepsilon)$ and mark-down (θ_i / η) of the *i*th firm are defined over the market demand elasticity ε and material input supply elasticity η , they differ in value. Following Appelbaum (1982), after aggregation across firms with the assumption that marginal costs are constant, an industry average value θ is obtained (Schroeter, 1988). The model for the aggregated industry is then estimated using a system of equations comprising the supply and demand relations on the output and input markets as well as the first-order condition of the industry's profit function, here (2.1), (2.5), (2.6), and (2.8). Furthermore, the conjectural elasticity is modeled to vary with market structure. Schroeter (1988) applied his approach to the U.S. beef industry for the years 1951 to 1983 using annual data. The estimated conjectural elasticity value is significant, but close to perfect competition. Moreover, its value is declining over time (1951: 0.0417, 1983: 0.0190), thus the increasing concentration in the industry has not worsened market condition. The relative mark-up of around 3% is larger than the relative mark-down of approximately 1% in the later years of the analyzed period, because of the more inelastic demand elasticity compared to the material input supply elasticity (Schroeter, 1988).

The fixed proportion assumption of Schroeter (1988) was later heavily challenged by studies (e.g. Wohlgenant, 1989; Goodwin & Brester, 1995) that found evidence that the food industry is rather characterized by input substitutability than fixed proportion technology. Azzam and Pagoulatos (1990) modify the PTA approach by using a production function for a single output instead of a cost function as well as allowing for variable input technology by taking *j* first-order conditions of the profit function for *j* inputs. Consequently, the output and input conjectural elasticity differ. However, the approach has two weaknesses. For estimation i+1structural equations are required and to identify the conjectural elasticities fixed values for the respective elasticity of output demand and input supply have to be assumed. The latter heavily relies on assumptions and is likely to overstate their significance (Wohlgenant, 2001). Nevertheless, the approach was like in Schroeter (1988) applied to the U.S. beef industry over the period from 1959 to 1982 using annual data. For the beef input market a conjectural elasticity of 0.178 and for the output market of 0.223 are obtained (Azzam & Pagoulatos, 1990). The values are drastically higher than by Schroeter (1988), what confirms Wohlgenant (2001) critic of using fixed elasticities of demand and input supply.

In contrast to Azzam and Pagoulatos (1990) the method developed by Murray (1995) does not assume values for the elasticities of output demand and input supply. The supply of *j* material input is assumed to be a function increasing in input price $(X^M = S(W^M, Z))$. Accordingly, differentiating this supply relation with respect to the material input price W^M allows obtaining the material input supply elasticity η . Assuming competitive output markets and differentiating an i^{th} firm's profit function (2.7) with respect to the level of material input choice x^{M}_{i} , gives the following profit-maximizing material input demand equation,

$$\underbrace{P(Q,Y)\frac{\partial Q(X^M,X^Z)}{\partial x_i^M}}_{i} = \underbrace{W^M(X^M,Z)\left(1+\frac{\xi_i}{\eta}\right)}_{i}$$
(2.9)

Value of Marginal Product 'perceived' Marginal Factor Cost where $Q(\bullet)$ is the total output quantity produced, $\xi_i = (\partial X^M / \partial x^M)(x^M / X^M)$ is the i^{th} firm's conjectural elasticity of material input supply or accordingly applied to an oligopsonistic market setting the gap between value of marginal product (VMP) and actual marginal factor cost (MFC) and X^M the total quantity of material input supply (Murray, 1995). Aggregating across firms again results an industry measure. The material input demand equation is then not typically estimated directly like in the approaches described previously in a system of demand and supply equations, because it still involves the marginal product term $\partial Q / \partial x^M_i$, but instead a shadow price (λ^M) equation is formulated,

$$\lambda^{M} = W^{M}(X^{M}, Z) \left(1 + \frac{z}{\eta}\right) = \frac{\partial \Pi(P, Q, X^{M}, X^{Z}, W^{M}, W^{Z})}{\partial X^{M}}$$
(2.10)

where λ^M is the shadow price and Π is industry's aggregated profit function, that equates 'perceived' MFC to VMP, i.e. $\partial \pi(\bullet) / \partial X^M$, represented by λ^M . The shadow price can then be calculated from the estimated industry's profit function. The profit function is estimated along with output supply and *j* input demand equations, obtained by applying Hotelling's lemma (Hotelling, 1932) to the industry's profit function, as a system of equations. With knowledge on λ^M and η , it is now possible to infer Ξ after solving (2.10) for it. Murray (1995) empirically investigated separately the U.S. markets for pulpwood and sawlogs, both input markets for the wood-processing industry, for the period 1958 to 1988 using annual data. For the sawlog market a mean value of 0.042 and for the pulpwood market of 0.174 for Ξ is derived (Murray, 1995).

In contrast to Murray (1995), Morrison Paul (1999; 2001) derives the industry's aggregated demand for the material input from its cost function. The cost function $C(\bullet)$ takes in Morrison Paul (2001) the form,

$$\underbrace{C(Q, W^{M}, W^{Z}, R)}_{Total Cost} = \underbrace{VC(Q, W^{M}, W^{Z}, R)}_{Variable Cost} + \underbrace{\sum_{k} W^{R} R_{k}}_{Fixed Cost}$$
(2.11)

where the total cost consists of variable cost $VC(\bullet)$ and fixed cost, R is a vector of k control variables, and W^R a vector of prices of k control variables. By applying now the 'inverse' Shepard's lemma optimization condition (Morrison Paul, 2001) the industry demand for the material input has the following formulation,

$$\frac{\partial C(\bullet)}{\partial X^M} = W^M \left(1 + \frac{z}{\eta} \right) = -\frac{\partial V C(\bullet)}{\partial X^M} = \lambda^M$$
(2.12)

Where $\partial C(\bullet)/\partial X^M$ is the MFC of the material input, and $\lambda^M = -\partial VC(\bullet)/\partial X^M$ is the shadow price. The shadow price λ^M is the "dual equivalent of the primal-based VMP" (Morrison Paul, 2001). Consequently, the expression MFC of the material input equaling the shadow price is the cost-side equivalent of the approach

of Murray (1995) profit maximization condition '*perceived*' MFC = VMP (see equation (2.10)).

Morrison Paul (2001) also applied the developed method to the U.S. beef industry in a similar fashion as Murray (1995) by estimating a system of equations comprising, the variable cost function, variable input demand equations, derived via Shepard's lemma, the optimal material input pricing equation, i.e. (2.12) solved for W^M , an output pricing equation, and a functional specification of Ξ that depends on structural variables, e.g. the number of cattle buyers. By assuming fixed proportions technology and oligopoly power on the output market, Θ also enters the output pricing equation. Consequently, the elasticities of supply and demand (ε and η) cannot be easily distinguished from Θ and Morrison Paul (2001) had to assume values for these. In contrast to the other empirical application on the U.S. beef industry, firm-level data for 43 plants in 1992-1993 were available. The results suggest very low levels of oligopsonistic behavior on the cattle input market.

For proper estimation of the method developed by Murray (1995) and Morrison Paul (1999; 2001) quantity data on non-material inputs are required, because they are part of the profit function (cost function), which in most cases are hard to obtain, and endogenous choice variables. Muth and Wohlgenant (1999) solved this issue and presented a model that allowed the estimation of oligopsony power with only two equations and without the requirement of quantity data on non-material inputs by applying the envelope theorem to the production function of a redefined profit function. Therefore, the authors use a slight different representation of the industry's profit function,

$$\Pi(P, X^{M}, Z, W^{M}, W^{Z}) = P \cdot f(X^{M}, X^{Z^{*}}) - W^{M}X^{M} - W^{Z}X^{Z^{*}}$$
(2.13)

where $f(X^M, X^{Z^*})$ is the production function and $X^{Z^*} = X^Z(X^M, W^Z, P)$ is a vector of the optimal quantities of *j* non-material inputs conditional on the level of the material input X^M . The first-order condition with respect to the chosen level of material input is,

$$\frac{\partial \Pi(\bullet)}{\partial X^{M}} = P \frac{\partial f(\bullet)}{\partial X^{M}} + P \frac{\partial f(\bullet)}{\partial X^{Z^{*}}} \frac{\partial X^{Z^{*}}}{\partial X^{M}} - W^{M}(\bullet) \left(1 + \frac{z}{\eta}\right) - W^{Z} \frac{\partial X^{Z^{*}}}{\partial X^{M}} = 0 \quad (2.14)$$

where the symbol definition is according to the previous equations. With the assumption that the non-material inputs are procured in competitive markets, rearranged (2.14) reduces to,

$$W^{M}(X^{M}, Z)\left(1 + \frac{z}{\eta}\right) = P \underbrace{\frac{\partial f\left(x^{M}, x^{Z}(x^{M}, w^{Z}, P)\right)}{\frac{\partial x^{M}}{Marginal Product}}}_{VMP}$$
(2.15)

where the marginal product term does not depend anymore on the quantities of the non-material inputs, but rather their prices, and the reduced-form VMP equals the 'perceived' MFC. Solving (2.15) for W^M and estimating this material input demand equation simultaneously with the inverse supply equation $S^{-1}(X^M,Z)$, which gives a value for η , results in an estimate of Ξ (Muth & Wohlgenant, 1999). As Schroeter (1988) and Azzam and Pagoulatos (1990) before, the elaborated framework was applied to the U.S. beef industry over the period from 1967 to 1983. Results of a constant estimate of the conjectural elasticity and a time trend adjusted one were presented. The revealed values are very close to zero and statistically insignificant (Muth and Wohlgenant, 1999).

3.2.3 Criticism

After the introduction through Bresnahan (1982) and Appelbaum (1979; 1982) NEIO quickly established itself as one of the most dominant approaches to the investigation of market power. Not even one decade after his seminal work Bresnahan (1989) reviewed 68 NEIO studies. Nevertheless, with its growing popularity criticism arose. In the following sections the limitations of NEIO framework as illustrated in the literature will be discussed.

One of the most obvious drawbacks of NEIO is that even though an estimate for the market power can be obtained, the source of the market power is never identified. Surely, as described in the previous section, some authors (e.g. Murray, 1995; Morrison Paul, 2001) tried to model the CPM as a function of exogenous variables, e.g. the number of cattle buyers, but never justified their choice of variables through economic theory. Lacking a clear idea on the market power source, NEIO empirical findings are heavily constrained in their ability to contribute to competition policy recommendations (Bresnahan, 1989).

Frequently data limitations and complexity of the model force NEIO researchers to formulate strong assumptions on technology, competition on other markets and market stages, as well as the type of competition (Wohlgenant 1989; Morrison Paul, 1999; Sexton, 2000). Even though NEIO studies mostly are restricted to the single industry case and thus can use knowledge of industry to approximate more closely the technology, still they repeatedly make questionable assumptions on the underlying technology, e.g. in the NEIO studies of the previous sub-

section. While Schroeter (1988) assumes fixed proportions technology, Morrison Paul (2001) on the other hand assumes values of the elasticity of input supply and output demand to be able to estimate the model. Furthermore, most NEIO studies heroically exclude the possibility of market imperfections on other input and/or output markets, even though market power on these markets might have a strong influence on the investigated markets (Sexton, 2000). NEIO models a priori always make a decision on the type of competition in the market, whether it is price-setting (Bertrand) or quantity-setting (Cournot), and exclude other forms, thereby strongly restricting themselves from the start and limiting their explanatory power (Sexton, 2000).

Looking past these common limiting assumptions, concerns regarding the validity of NEIO estimation results have been frequently stated. Corts (1999) demonstrated that, in a dynamic oligopoly, the conjectural elasticity is correlated with the instruments applied to identify it, unless the data is truly generated by an equilibrium outcome of firms that behave according to the imposed theoretical model of conjectural variations (Sexton, 2000). Consequently, the estimate of the mean conjectural elasticity is in most cases biased. Perloff and Shen (2012) further proved that the choice of functional form can be essential. While the choice of linear functional forms results in unreliable estimates due to multicollinearity issues, flexible functions circumvent this problem (Sexton & Lavoie, 2001). However, not only the choice of functional form, but also the choice of approach, GIM or PTA, leads to significantly different results, as shown by Perekhozhuk et al. (2016).

Apart from the importance of the choice of functional form or approach for the validity of results, the strongest critic concerning the CPM has been that even though the model assumes dynamic concepts, like conjectures, the behavior of firms is estimated as static and firms maximize one-period profits instead of a continuous discounted stream of profits over a given period (Dixit, 1986). Consequently, the obtained static estimates do not adequately represent conjectural variations or elasticities, which are in reality formed and adjusted repeatedly (Richards et al., 2001). Furthermore, the static NEIO model neglects strategic variables, such as advertising and product quality, which are tools of interfirm competition and have long-term effects, and thus possible adjustment costs (Roberts & Samuelson, 1988).

3.2.4 Dynamic models

In the static framework, as described in the previous subsection, firms maximizes profit by taking the belief on how rivals react as given, and form their decision

on price or quantity solely on observations on demand, supply, and cost shifters (Karp & Perloff, 1993a). Actions preceding this period do not affect behavior, in other words history is no concern (Perloff et al., 2007). In contrast to this so-called one-shot game, firms, which interact repeatedly, also consider the history of past actions to form their decision-rule and optimal response (Karp & Perloff, 1993a). Consequently, in a dynamic framework a firm does not just maximize its current profits but rather the discounted-profit stream,

$$\Pi_i = \sum_{t=0}^{\infty} \delta \pi_{it} \tag{2.16}$$

where Π_i is the stream of discounted profits of firm *i* over an infinite time horizon, δ a discount factor common to all firms in the range of $0 < \delta < 1$, and π_{it} the firm *i*'s profit at time *t* (Slade, 1989). The *i*th firm now uses a set of strategies formed from a function of the observable history. If all firms apply this equilibrium decision rule at all times, the arising equilibrium of the industry is the Nash equilibrium solution to the dynamic game (Perloff et al., 2007).

The cause of interfirm dynamics can be either strategic or fundamental. If rivals respond in the future to current actions by observing the history of the game, the source is strategic, while if the current action affects a stock variable that again influences future profits the source is fundamental (Perloff et al., 2007). For example, fundamentals can be capital, whose stock depreciates over time, or advertising, since the number of future customers and thus future profits depend on current decisions on the advertising intensity (Roberts & Samuelson, 1988).

3.2.4.1 Strategic dynamic models

In the strategic case, if a firm deviates from the equilibrium strategy, shaped by the history of the game, and 'cheats' on the other firms, collusive action will break down and firms enter a phase of punishment (Porter, 1983a). In the case, that small changes in the history occur and the reaction of the firms is severe, e.g. regime change from cartel to competitive pricing or 'price-wars', the firms use discontinuous and so-called trigger strategies. However, if firms react adequately to changes in the variables of the history, we consider this as continuous strategies (Perloff et al., 2007). Continuous and discontinuous strategies have been modeled and empirically applied by a series of authors, but in the following we will discuss the applications of Porter (1983b) and Slade (1989), the first being an example for the trigger strategies and the second one for continuous strategies.

In his analysis of the Joint Executive Committee, an U.S. railroad oligopolistic cartel that existed from 1880 to 1886, Porter (1983b) assumes that firms ob-

serve only their output level and the industry price and thus collusion breaks down, firms apply trigger strategies, and the industry enters a punishment phase when the actual price falls below a certain threshold. The model is based on the theoretic work of Green and Porter (1984), who also argue that 'price wars' can not only be a result of 'cheating', a firm producing more than agreed, but also of unexpected low demand due to exogenous shocks. The model starts by deriving the total quantity demanded similar to (2.1) in log linear functional form. The industry supply function is derived analogously to Appelbaum (1982). Rearranging (2.3) the industry's supply function and taking the natural logarithm gives,

$$\ln P = \beta_0 + \beta_1 \ln Q + \underbrace{\beta_2}_{-\ln\left(1 + \frac{\theta}{\varepsilon}\right)} I$$
(2.17)

where *I* is a so-called indicator variable that equals one when the industry is acting collusive with the probability ζ and 0 when it is in a revisionary period with the probability 1- ζ . Applying switching regression technique to the system of two equations (2.1) and (2.17), the probability ζ and parameter estimates are obtained. With knowledge on ζ the sample can be split in collusive and revisionary periods. With estimates for β_2 and ε , the value of θ can be approximated and compared to infer on the type of collusive arrangement, pending between, the values *H* (Herfindahl index $H = \sum_i s_i^2$ for the Cournot outcome or 1 for the monopoly outcome of an effective cartel. In Porter's (1983b) the study of the U.S. railroad cartel a value for θ of 0.336 was calculated during collusive periods, which is close to the Cournot outcome. This resulted in 66% higher price and 33% lower quantity during cooperative periods compared to competitive ones, benefiting the cartel overall with 11% higher profits (Porter, 1983b).

While, the theoretic model of Green and Porter (1984) and the empirical application (Porter, 1983b) are based on trigger strategies, discontinuous strategies that switch between periods of collusion and competition, Slade (1989) developed a model of the Vancouver gasoline market using continuous strategies in form of reaction functions. Even though, agreements between firms on the quantity produced or price charged are illegal in most countries, Slade (1989) argues that when a game, interaction between firms, is repeated frequently a tacit collusive outcome can arise. In an *N* firm differentiated product market the quantity sold at time *t* by firm *i* is a function of its own price and the prices of the competition *j*,

$$q_{it} = D(p_{it}, p_{jt}) \tag{2.18}$$

whose parameter vary with time due to occasional structural shifts in demand. Furthermore, it is assumed that marginal costs are equal across firms and since firms interact repeatedly they maximize their discounted-profit stream Π_i rather than single period profits. Given these assumptions, (2.16) can be reformulated to yield the discounted-profit stream of the i^{th} firm,

$$\Pi_i = \sum_{t=0}^{\infty} \delta[(p_{it} - MC_t)q_{it}]$$
(2.19)

for whose maximization firm i uses a strategy, depending on the history of the game. By limiting the history of the game to the prices of the preceding period and thus assuming the game to be Markovian, Slade (1989) formulates the strategy of the i^{th} firm to be,

$$p_{it} = P_t^* + \zeta_t \sum_{j \neq i} (p_{jt} - P_t^*)$$
(2.20)

where P_t^* is a stationary price that all firms charge in the tacit collusive equilibrium at time t and only depends on the structural parameters of (2.18), and the slope of the reaction function ζ_t , which vary due to strategy revision caused by changes in the exogenous parameters. When the parameters instead are constant, a stationary-equilibrium price P^* can be obtained¹. If firms do not react, $\zeta = 0$, the Bertrand-Bash price is the outcome of the one-shot game. However, if the discount factor and ζ equal one, the price is on perfectly collusive level. Consequently, ζ lies like θ in the range of zero to one, perfect competition to perfect collusion (Slade, 1989).

By applying the Kalman-Filter (Kalman, 1960), which uses measurement equations on observational data, here prices and quantities, and transition equations to approximate the unobservable, here changes in the parameters, it is possible to allow for time-variation in the estimates. In the empirical application of Slade (1989) the state-space model consist of a system of N + N measurement equations comprising N Equations of the firm *i*'s demand (2.18), and N equations of a differenced version of firm *i*'s strategy or reaction function, as well as N^2 transition equations, allowing the reaction functions to depend on lagged values, lagged prices, and a dummy variable that takes the value one, if rivaling firms increased the price and zero otherwise (for more details see Slade (1989)). The model is applied to a unique dataset from a price-war period in the Vancouver gasoline retail market consisting of weekly data of the summer 1983 on prices, variable costs, and sales volumes of individual gasoline service-stations.

¹ with $q_{it} = a + bp_{it} + d\sum_{j \neq i} p_{jt}$ and $p_{it} = p_{jt} = P^*$, $P^* = \frac{a + bMC - (n-1)\delta\zeta dMC}{2b - (n-1)d(1+\delta\zeta)}$ (Slade, 1989).

Slade (1989) reports that the service-stations were successful in tacit collusion and the equilibrium price was above the Bertrand-Nash level (Slade, 1989).

Some downsides to the strategic dynamic concept exist. For trigger strategies to be applicable, no other costs of adjustment and unlimited capacities are necessary conditions. Furthermore, in monopolistic or competitive market settings trigger strategies simply cannot exist, and if in addition no fundamental dynamics occur, these settings can be modeled as a sequence of static problems (Perloff et al., 2007).

3.2.4.2 Fundamental dynamic models

In contrast to the strategic dynamic model, fundamental dynamics always prevail in any kind of market structure as long as quasi-fixed inputs are necessary to the production process. Inputs are titled quasi-fixed if they cannot be instantaneously varied due to adjustment costs (Perloff et al., 2007).

The first ones to apply a model of oligopoly based on fundamental dynamics empirically following the theoretic works of Fershtman and Kamien (1987) and Reynolds (1987) were Roberts and Samuelson (1988) and Karp and Perloff (1989). While the model of the first focuses on the stock of advertising good will as the state variable and the approach of the latter, in their study on the rice export market, owes its dynamics due to costs when adjusting the level of output. Later on Karp and Perloff (1993b) also applied the framework to the coffee export market and generalized their model theoretically (Karp & Perloff, 1993a).

A general model of fundamental dynamics in homogenous product market incorporating adjustment costs like the one of Karp and Perloff (1993a), starts out by assuming a linear inverse demand equation (P = D(Q)) and quadratic costs of production. Furthermore, often no firm-specific differences in the marginal cost are supposed to exist ($MC = \alpha q_{it}/2$). Furthermore, firm *i* faces quadratic cost when adjusting its output level from the previous period *n* ($u_{it} = q_{it} - q_{it-n}$), which can be formulated as,

$$\left(\gamma_{0i} + \gamma \frac{u_{it}}{2}\right) u_{it} \tag{2.21}$$

where u_{it} is the rate of output level change and since again no firm-specific differences in the adjustment cost are assumed, the constant γ_{0i} is dropped from (2.21) and adjustment cost are zero if no change in output level occurs. Having made these assumptions and given a common discount factor δ the firm *i*'s aim is to maximize its discounted-profits stream Π_i ,

$$\Pi_{i} = \sum_{t=1}^{\infty} \delta^{(t-1)} \left[(P_{t} - \alpha q_{it}) q_{it} - \gamma \frac{u_{it}^{2}}{2} \right]$$
(2.22)

assuming that firms use Markov strategies, firm *i* forms its decision on changes in its output rate as a function of the state variables, its own and other *j* firms' lagged output levels, the function $J_{it}(q_t, v)$ can be defined as the value function,

$$J_{it-1}(q_{t-1},\nu) = \max(P_t - \alpha q_{it})q_{it} - \gamma \frac{u_{it}^2}{2} - \delta^{(t-1)}J_{it}(q_t,\nu)$$
 (2.23)

where $v = \partial q_{jt} / \partial q_{it}$ is defined as the market power conduct parameter, which in the static models would represent the conjectural variation. The response of rival *j* to an output change of firm *i*, is an index of market power and can range from negative one (perfect competition) over zero (Cournot) to one (monopoly) (Katchova et al., 2005). Taking the first-order condition that maximizes firms *i*'s present discounted value function with respect to the decision on its own output level and rearranging gives:

$$\underbrace{P_{t} = \alpha + (1+\nu)\frac{\partial P(\bullet)}{\partial q_{it}}q_{it}}_{Static model} + \underbrace{\gamma u_{it} - \delta^{(t-1)} \left[\frac{\partial J_{it}(q_{t},\nu)}{\partial q_{i}} + \frac{\partial J_{it}(q_{t},\nu)}{\partial q_{j}}\nu\right]}_{Dyanmics}$$
(2.24)

Katchova et al. (2005) create a link between the market power conduct parameter ν of Karp and Perloff (1993a) and the conjectural elasticity Θ of static models by deriving the equality $\theta_i = (1+\nu)q_i/Q$. Consequently, (2.24) equals the static model plus a dynamic component, which disappears when γ and $\delta(^{t-1})$ equal zero (Katchova et al., 2005). To be able to derive estimates for ν first the control function ($q_t = g(t) + \Lambda \cdot q_{t-1}$) is estimated with a matrix Λ , restricted to be symmetric across all firms, and thus implying that firms have the same conjectures, that fulfils a similar purpose to ζ in the reaction function of the strategic approach of Slade (1989).

Furthermore assumptions have to be made on whether firms adapt their strategies to unpredicted changes, use feedback strategies, or not, or use open-loop strategies. In the latter case, firms form a set of optimal response strategies from an initial state but do not revise their strategy in the incidence of unpredicted shocks. Since firms do not react to unpredicted shocks, their strategy is not subgame perfect. Consequently, the open-loop model is similar to the single period static game. However, when firms use feedback strategies they adapt their strategies to unexpected changes. Thus, a model with coherent feedback strategies is subgame perfect. Nevertheless, if the market structure is a competitive or a monopolistic one, the open-loop and feedback equilibria as well as adjustment paths coincide (Karp & Perloff, 1993b; Katchova et al., 2005). The choice of equilibrium concept defines the equations from which v and γ can be inferred (for details see Katchova et al. (2005)). In the case of the feedback model further assumptions, apart from the symmetry of Λ , implying also symmetry of v across firms, and limiting the time horizon, have to be drawn, because otherwise no unique equilibrium will result due to the existence of more unknown parameters than equations that determine the steady-state. By defining the control rule to be linear and the value function to be linear-quadratic a closed-loop equilibrium solution can be obtained (Perloff et al., 2007).

As mentioned before, Karp and Perloff (1989; 1993b) applied their model to the rice and coffee export markets, which are both characterized by oligopolies formed by a small number of large exporting countries. The results are similar for both studies. Brazil and Colombia in the coffee export market as well as China, Thailand, and Pakistan in the rice export market behaved close to the price-taking level (Karp & Perloff, 1989; 1993b).

3.2.4.3 Criticism

Dynamic models solve the issue of the static estimation of a dynamic concept. However, this accomplishment does not come without any essential drawbacks. First, dynamic models are highly complex, thus small changes in the assumptions of the model can change the outcome completely. Slade (1995) states that already slight alterations in the timing of events or information can lead the game to drastically different results. In addition, often firm-level data is required, which is frequently not available (Sheldon & Sperling, 2003). Furthermore, restrictive assumptions on quadratic-linear games have to be made to identify a unique equilibrium, thus the solution is a result of a certain family of equilibria and results in limited explanatory power of the model (Karp & Perloff, 1989; Perloff et al., 2007).

Furthermore, dynamic models are only appropriate if the source of dynamics is actually significant for the equilibrium solution. Consequently, in the case of fundamental dynamics, adjustment costs have to be substantially large to result in dynamics (Karp & Perloff, 1993a). Perloff et al. (2007) further criticize that even if current prices are observed and some sort of MC measure is obtained, a measure of market power can not necessarily be inferred, since they might depend on unobservable opportunity costs.

While it is desirable to apply a dynamic model, due to the issues arising from complexity of the dynamic models and data requirements, some authors have argued that the static conjectural variations is a good approximation of a non-modeled dynamic game in reduced form (Cabral, 1995). While Cabral (1995) restricts his model to the linear oligopoly, Dockner (1992) is able to circumvent this issue by assuming a differential game with adjustment costs. By solving this for the subgame perfect equilibrium he is able to show that the conjectural variation solution is the same as the dynamic one (Dockner, 1992).

3.3 PRICE TRANSMISSION ANALYSIS

Apart from the NEIO approach, the price transmission analysis has been a dominant framework for the investigation of market power since the 1970s. The general idea is that markets are in almost all cases at least either vertically, e.g. through the transformation process along a supply chain, or horizontally, e.g. via foreign trade, integrated. Considering vertically integrated markets, in the case where changes of prices are transmitted symmetrically in this system of markets then an price increase or decrease on one of the integrated market should be passed on instantaneously and in relative magnitude to the upstream or downstream markets (see Figure 5). Assuming that the downstream price depends on the upstream price in a system of vertically integrated markets, and a linear functional form, the relationship between the prices takes the form,

$$P_t^d = \beta_0 + \beta_1 P_t^u + \varepsilon_t \tag{2.25}$$

where P^d_t is the downstream price and P^u_t is the upstream price at time t. If prices are in natural logarithms, β_1 represents the price transmission elasticity τ , which given perfect price transmission equals one. Since its beginnings in the 1970s two quite different strands of price transmission methodology have evolved in the literature, which focus on market power in their analysis. The asymmetric price transmission framework (APT) tries to detect divergence from perfect price transmission by testing whether a dependent price series reacts symmetrically or asymmetrically to changes in an independent price. The equilibrium displacement model (EDM) instead models market power similar to conjectural variation in the NEIO methodology and then derives a measure of the price transmission elasticity including the effect of market power. In the following sections, both methodological approaches will be presented and their appropriateness for the identification of market power discussed.

3.3.1 Asymmetric price transmission (APT)

In the case of imperfect price transmission, the price transmission elasticity is unequal to one and welfare is transferred to the stage benefiting from this imperfection. The cause can be so-called asymmetries. Asymmetric price transmission

and the accompanying welfare transfer can be temporary, e.g. if a delay in the price transmission exists, but the magnitude is transmitted equally (in Figure 5 over the period t_1 to t_2). However, if the magnitude does not symmetrically adjust, the asymmetry and welfare transfer is of permanent nature (Meyer & von Cramon-Taubadel, 2004).

Figure 5: Illustrations of cases of symmetry (upper left), asymmetry in speed (upper right), asymmetry in magnitude (lower left), and asymmetry in speed and magnitude (lower right) in the price transmission of P^d (downstream price) and P^u (upstream price), assuming that a shock in P^d is transmitted to P^u.



Source: own illustration.

Peltzman (2000) furthermore distinguishes between positive and negative APT (see Figure 5). In the occurrence of positive APT, price changes that could lower the margin between upstream P^u and downstream price P^d , increases in the upstream price and decreases in the downstream price, are faster and more fully transmitted, than respectively price decreases in the upstream and increases in the downstream price changes that could increase the price margin are faster and more fully transmitted. While the downstream market agents profit from the positive asymmetry in price transmission, upstream agents benefit in the case of negative APT (Peltzman, 2000; Meyer & von Cramon-Taubadel, 2004).

The source of APT is normally not clearly identified. One strain of APT literature argues that the cause of APT lies in the abuse of market power by one stage (e.g. Ward, 1982; Kinnucan & Forker, 1987; Borenstein et al., 1997; Peltzman, 2000; Miller & Hayenga, 2001). The actors of the upstream or downstream stage use their bargaining power to delay or absolutely restrain from transmitting price changes to the other stage resulting in a transfer of welfare. Depending on which level of the price transmission system uses market power, either positive or negative APT can arise. While an oligopoly or/and oligopsony at the downstream level will likely try to act in a way which increases the price-margin, an oligopoly at the upstream market will lower the price-margin.



Figure 6: Illustration of positive APT (left plot) and negative APT (right plot).

Source: own illustration.

However, some authors have argued that the picture is not as clear as it seems at first glance. For example, Ward (1982) in his study of the farm-to-retail price transmission has argued that downstream agents acting as an oligopoly might be unwilling to increase downstream prices symmetrically to upstream price increases to avoid the risk of declining market shares or fear the spoilage of perishable products resulting in negative APT rather than positive (Ward, 1982). Borenstein et al. (1997), on the other hand, in their study of the price transmission from crude oil to gasoline prices, state that in imperfect information settings crude oil price increases are transmitted faster than decreases. The original oligopoly price of gasoline in case of a crude oil decrease appears to be a natural point of orientation in uncertainty of the price adjustment behavior of the other firms, while an increase noticeably decreases the price-margin and is adjusted more quickly.

Another often stated source of APT are adjustment costs, also called menu costs. Adjustment costs in the price transmission framework arise if the cost of adjusting the price to a price change on the other level instantaneously would cause greater economic losses than benefits. For example, Bailey and Brorsen (1989) showed that meat packers face large fixed cost of capital and lower margins due to increases in the upstream price for beef in the short-run to enable constant full utilization of the capital. Furthermore, as mentioned before, the risk of food spoilage might be a reason for retailers to be reluctant in increasing their prices (Ward, 1982). Reagan and Weitzman (1982) further argued that firms during phases of low demand would rather build up inventories then decrease their prices and in phases of high demand more quickly increase their prices. Since adjustment costs are a result of adjusting the prices to a change in another price, the welfare transfer is only temporary, while in the case of stable exercise of market power it could be permanent (Meyer & von Cramon-Taubadel, 2004).

Apart from market power and adjustment costs, a series of other causes have been mentioned less frequently in the literature for APT. In the case of food products, Hall et al. (1981) stated that one cause of asymmetries can be the lethargy of the food marketing process, which includes storage, transport, and processing of the goods. Also lethargy or differences regarding price increases and decreases in the price reporting system can cause asymmetries (Bailey & Brorsen, 1989). Moreover, government interventions can cause APT. Kinnucan and Forker (1987) state that policy interventions like intervention prices cause APT, since firms can assume that in the presence of a floor price for a certain product price decreases are only temporary, while increases can be permanent. With the beginning of the 1970s several studies were conducted to detect APT and indirectly market power as a possible source of these APT. In the following decades, theoretical considerations and methodological advancements have altered the methods applied for this purpose. In the following sections an overview over the methodological approaches that have been established to determine APT will be presented.

3.3.1.1 APT methodology

The idea of irreversibility was first discussed by Marshall (1936) and Haavelmo (1944). Farrell (1952) in his study of irreversible demand describes irreversibility using an example of a smoker that will adjust his smoking consumption differently, i.e. asymmetrically, depending on whether the price of tobacco increases or decreases. Tweeten and Quance (1969) use dummy variables to estimate an irreversible aggregate supply function, with the purpose to answer the question whether farmers adjust their output less to price decreases than increases. In their approach the total farm output is a function of the stock of productive assets, a productivity index, and a price ratio of prices received for their output and the lagged prices paid for inputs by farmers. Using a different notation to make the following approaches comparable, their linear supply function would take the form,

$$Q_t = \beta_0 + \beta_1 P_t D_t + \beta_2 P_t (1 - D_t) + \beta_3 Z_t + \varepsilon_t$$
(2.26)

where Q_t is the total output, P_t the price ratio, and Z_t a vector of the other included variables at time t. The dummy variable D_t takes a value of one if $P_t \ge P_{t-1}$ or zero otherwise. Tweeten and Quance (1969) using (2.26) argued that if $\beta_1 \ne \beta_2$ farmers' behavior would be asymmetrical.

The first to create a method to detect irreversibility of prices was Wolffram (1971). He based his approach on the framework of Tweeten and Quance (1969), but altered the price splitting methodology. The reason for this is that he criticized that the approach of Tweeten and Quance (1969) would result in biased estimates for β_1 and β_2 due to a probable non-constancy of the constant β_0 (Meyer & von Cramon-Taubadel, 2004). His model can be formulated as,

$$P_t^d = \beta_0 + \beta_1 (P_0^u + \sum_{t=1}^T D_t \Delta P_t^u) + \beta_2 (P_0^u - \sum_{t=1}^T (1 - D_t) \Delta P_t^u) + \varepsilon_t \quad (2.27)$$

where D_t has the same meaning as in (2.26), Δ is the first difference operator, and the upstream price variable is split by either adding or subtracting the first

differences of the upstream price to its original value depending on price increases or decreases compared to the previous period at time *t*. As in Tweeten and Quance (1969) symmetry of price transmission is tested by testing the hypothesis $\beta_1 = \beta_2$ (Wolfram, 1971).

By dropping the initial values of the prices from (2.27) Houck (1977) was able to give a clearer representation of the method of Wolfram (1971). Houck (1977) elaborated that when the effect of differenced prices is to be determined the initial values are obsolete for the estimation. Thus, (2.27) can be rearranged into,

$$P_t^d = \beta_0 + \beta_1 (P_0^u + \sum_{t=1}^T D_t \Delta P_t^u) + \beta_2 (P_0^u - \sum_{t=1}^T (1 - D_t) \Delta P_t^u) + \varepsilon_t \quad (2.28)$$

where the initial value of the downstream price is subtracted from its value at time t and the resulting dependent variable only depends on the cumulative price changes. He further simplifies the model by giving a representation of (2.28), which only includes the first differences without the requirement of estimating the cumulative effect,

$$\Delta P_t^d = \beta_0 + \beta_1 D_t \Delta P_t^u + \beta_2 (1 - D_t) \Delta P_t^u + \gamma_t$$
(2.29)

where, as in all specifications so far, the symmetry in price transmission is determined by testing the hypothesis that $\beta_1 = \beta_2$. Ward (1982) altered the approach of Houck (1977) by adding additional lags to the first differences of the upstream price and allowing them to differ in their number for the increasing and decreasing phases. The hypothesis of symmetry would have to be accepted if the sum of these lagged parameter polynomials $\Sigma \beta_1 = \Sigma \beta_2$ are equal (Ward, 1982).

3.3.1.2 Cointegration and threshold APT

Granger and Newbold (1974) proved that using standard regression techniques for the estimation of a relationship between non-stationary variables would create spurious results that appear to prove the relationship between variables, which truly does not exist. This is of utter importance to APT, since bundles of price time series, in particular or vertically and horizontally integrated markets, are often non-stationary, but rather have a stochastic trend in common (Hendry & Doornik, 2001; Juselius, 2006). Von Cramon-Taubadel (1998) stated that many previous applications in the field of APT suffered first-order autocorrelation, a sign of spurious regression. In addition, it was shown that if the price time-series are cointegrated the approach of Wolffram (1971) and Houck (1977) should reject asymmetry. Time series of variables are labelled as 'cointegrated' when one or more combinations of these exist that are stationary in the long-run (Juselius, 2006). Consequently, a solution to the issue of spurious regression is to test whether the variables are cointegrated and if so apply a cointegration framework (von Cramon-Taubadel, 1998).

The first to apply cointegration techniques to APT were von Cramon-Taubadel and Fahlbusch (1994). Their approach was based on the 2-step error correction model (ECM) developed by Engle and Granger (1987) and the idea of non-symmetric ECM by Granger and Lee (1989). The procedure is the following. After testing for the existences of unit roots, non-stationarity, as well as for cointegration among the time series, (2.25) represents the long-run equilibrium relationship of the downstream and upstream price. The lagged residuals ε_{t-1} are deviations from this long-run equilibrium, the so-called error correction term (ECT). Consequently, these deviations can be interpreted as adjustments to the price transmission. After splitting the ECT in positive (*ECT*⁺) and negative (*ECT*⁻) deviations, a non-symmetric error correction representation can be formulated as,

$$\Delta P_t^d = \alpha_0 + \alpha_1 E C T_{t-1}^+ + \alpha_2 E C T_{t-1}^- + \alpha_3 \Delta P_t^u + \alpha_4(L) \Delta P_{t-1}^d$$
$$+ \alpha_5(L) \Delta P_{t-1}^u + \varepsilon_t \tag{2.30}$$

where $\alpha_4(L)$ and $\alpha_5(L)$ are lag polynomials of the order L. Equation (2.30) allows to determine whether APT is present by testing if the equality of the corresponding ECTs parameters $\alpha_1 = \alpha_2$ holds (von Cramon-Taubadel & Fahlbusch, 1994; von Cramon-Taubadel, 1998). In an extension, von Cramon-Taubadel and Loy (1996) also suggest to split the first difference of the upstream price ΔP^u in positive and negative deviations to test for additional asymmetries in price transmission.

The cointegration test elaborated by Engle and Granger (1987) requires the longrun equilibrium to be constant over time. Gregory and Hansen (1996) extended the testing procedure by permitting to test the cointegration relationships of time series with the presence of structural breaks, i.e. level shifts (with and without trend) and regime switches. Furthermore Enders and Granger (1998) modified the test for cointegration to allow for asymmetric adjustment of the deviations to the long-run equilibrium in their work. As a result of the methodic advances, cointegration analysis, as mentioned above, could now be conducted with time series that failed the cointegration test in the first place due to structural breaks. Thus, it is also possible to observe asymmetric price transmission in magnitude and not just in speed as under the original framework of von Cramon-Taubadel and Fahlbusch (1994) (Abdulai, 2000; Meyer & von Cramon-Taubadel; 2004). One modification that allows for non-linear adjustment in the ECM is the threshold ECM. Instead of splitting the ECT in positive and negative deviations, it is also possible to search the data series for n thresholds that define n+1 regimes and split the ECT time series accordingly in regime respective time series. Therefore, the estimated parameters of (2.30) vary with regime (Goodwin & Holt, 1999; Goodwin & Harper, 2000; Goodwin & Piggott, 2001). The number of thresholds is not restricted. For illustration, Goodwin and Harper (2000) in their study of the farm-to-retail price transmission in the U.S. pork sector define two thresholds, a lower and upper threshold, which allows extreme deviations from the long-equilibrium to affect price changes differently than in the 'neutral' interval. The representation of the corresponding threshold cointegration model with two thresholds l_1 and l_2 is given by,

$$\Delta P_{t}^{d} = \begin{cases} ECT_{t-1} < l_{1}: \\ \alpha_{0,1} + \alpha_{1,1}ECT_{t-1} + \alpha_{2,1}\Delta P_{t}^{u} + \alpha_{5}(L)\Delta P_{t-1}^{u} + \alpha_{3,1}(L)\Delta P_{t-1}^{d} + \varepsilon_{t} \\ l_{1} \leq ECT_{t-1} \leq l_{2}: \\ \alpha_{0,2} + \alpha_{1,2}ECT_{t-1} + \alpha_{2,2}\Delta P_{t}^{u} + \alpha_{5}(L)\Delta P_{t-1}^{u} + \alpha_{3,2}(L)\Delta P_{t-1}^{d} + \varepsilon_{t} \\ ECT_{t-1} > l_{2} \end{cases}$$

$$(2.31)$$

where the estimated parameters change depending on the regime. Consequently, if thresholds are statistically determined, the speed of adjustment parameter could differ between the regimes and inequalities of these estimates indicate asymmetry in price transmission. This allows interpreting the results in a new way in the APT framework. Thresholds can be seen as a measure of adjustment costs, since the ECT has to reach a certain level before price adjustment will take place (Goodwin & Harper, 2000; Meyer & von Cramon-Taubadel, 2004). Apart from threshold cointegration, also Markov-switching (MS) techniques have been applied to the ECM to allow regime shifts. In contrast to applying grid search procedures that find a threshold that maximizes the likelihood function (Tsay, 1989) or minimizes the sum of squared errors (Balke & Fomby, 1997), it is assumed that a state variable, defining the regimes, is driven by a Markov-chain. Like in the case of applications to the dynamic CPM, the state variable depends only on the previous period and not on the whole history of the regime (Hamilton, 1989; Krolzig, 1997). Furthermore, not all parameters have to differ with the regime, but they can be selected, what allows the researcher to model the price transmission more flexible (Djuric & Götz, 2016).

3.3.2 Equilibrium displacement model (EDM)

The equilibrium displacement model (EDM) was introduced by the seminal work of Gardner (1975). It is assumed that *n* firms produce a homogenous output good using two inputs, a material one and non-material one and sell it at price *P*. Like Gardner (1975), most EDM works that followed analyzed the farm-to-retail price spread, and assumed the two inputs to be farm and marketing service input. However, to continue the denomination used in the NEIO section, the inputs will be labelled material and non-material respectively (Gardner, 1975; McCorriston et al., 1998; 2001; Weldegebriel, 2004; 2012).

The EDM model consists of supply and demand functions for the output product, production function of the i^{th} firm and an inverse of the demand function of (2.1), for the inputs, inverse supply function, depending on quantities produced and exogenous shift variables, and derived demand through applications of inverse of Shepard's lemma or the price of the inputs equaling VMP with the assumption of profit-maximizing behavior. Each supply and demand equation of this system is differentiated with respect to a change in one of the exogenous variables at either the output or each of the input levels. The new derived equations are solved for the explicit solutions of the endogenous variables, the prices and quantities of the output and inputs, which provide the equilibrium equations where the endogenous variables solely depend on the exogenous variables (Gardner, 1975; McCorriston et al., 2001).

Using the equilibrium solutions for the changes in output and the material input, it is possible to derive the price transmission elasticity τ . Assuming the shock was caused by the exogenous variable of the material supply function, the price transmission elasticity in Gardner (1975) is given by,

$$\tau = \frac{d\ln P/d\ln Z}{d\ln W^M/d\ln Z} = \frac{s^M \left(1 + \frac{\sigma}{\eta^2}\right)}{1 + s^M \frac{\sigma}{\eta^2} - s^Z \frac{\varepsilon}{\eta^Z}}$$
(2.32)

where s^M and s^Z are the cost shares of the material input and the non-material input respectively, η^Z is the price elasticity of the non-material input supply, ε is the price elasticity for the output demand, and σ is the elasticity of substitution between the material and non-material input. Gardner (1975) derived further price transmission elasticities with respect to changes due to exogenous shocks at the non-material and output level. Nevertheless, like subsequent works we will focus on the price transmission elasticity derived from a shock at the material input level. From (2.32) it can be deduced that the smaller the elasticity of substitution between the inputs, the more volatile the farm-to-retail price ratio reacts to supply shocks. Furthermore, Gardner (1975) found that the effect
on the farm-to-retail spread of retail demand shocks is larger than the effect of farm supply shocks. If the elasticity of substitution is relatively low and supply for material inputs relatively elastic, τ will approximate the value of s^{M} (Gardner, 1975; Weldegebriel et al., 2012).

Nevertheless, the EDM of Gardner (1975) has little use in the investigation of market power, since competitive markets are assumed. McCorriston et al. (1998) modeled an oligopolistic setting on the output market by deriving a supply relation similar to the static CPM model of Bresnahan (1982) (2.2). The price transmission elasticity is then derived according to the EDM of Gardner (1975) by differentiating the system of equations with respect to a material input supply shock,

$$\tau = \frac{s^{M} \left(1 + \frac{\sigma}{\eta Z}\right)}{\left(1 + s^{M} \frac{\sigma}{\eta Z}\right) (1 + \mu^{\Theta}) - s^{Z} \frac{\varepsilon}{\eta Z}}$$
(2.33)

where $\mu^{\theta} = (\partial(\theta/\varepsilon)/\partial\varepsilon)(\partial\varepsilon/\partial P)$ is the change in the industry mark-up. If θ equals zero, also μ^{θ} equals zero, (2.33) collapse to the competitive version (2.32). Comparing the price transmission elasticity τ^c in a competitive setting to the noncompetitive level τ lets McCorriston et al. (1998) conclude that in the case of linear demand and supply functions oligopoly power at the downstream market will result in growing imperfect price transmission. Furthermore, using measures for the elasticities of previous studies (Wohlgenant, 1989; Azzam & Pagoulatos, 1990; Schroeter & Azzam, 1990) on the U.S. beef and pork sector, McCorriston et al. (1998) reported that the price transmission would greatly differ depending on whether markets were assumed to be competitive and variable proportions technology or imperfect competitive markets with fixed proportion assumption (1998) (McCorriston et al., 1998).

Like Gardner (1975), the framework of McCorriston et al. (1998) assumes constant returns to scale (CRS). However, this has been questioned by empirical findings, e.g. of Morrison Paul and Siegel (1999) who found short- and longrun increasing returns-to-scale in the U.S. manufacturing sector. Consequently, McCorriston et al. (2001) extended the framework to account for non-constant returns to scale. The main result was that the scale parameter truly influences the effect of market power on the price transmission elasticity. It can be greater or less depending on the returns to scale measure. In case of increasing returns to scale, the influence of market power is 'offset' (McCorriston et al., 2001).

Weldegebriel (2004) extends the work of McCorriston et al. (2001) to account for oligopsony power in the material input market. Instead of using equation

(2.20) for the industry's material input demand, a modification of Schroeter's (1988) derived demand equation (2.8) enters the system of equations to deduce the equilibrium equations. Also accounting for non-constant returns-to-scale by incorporating the scale measure ρ the price transmission elasticity in the event of an exogenously induced supply shock takes the form,

$$\tau = \frac{\rho \psi^{M}(1+\mu^{\Xi}) \left(1+\frac{\sigma}{\eta^{Z}}\right)}{\left(\rho+\psi^{M}\frac{\sigma}{\eta^{Z}}\right) \left[(1+\mu^{\Theta})\rho+\varepsilon(\rho-1)\right] - \psi^{Z}\frac{\varepsilon}{\eta^{Z}}}$$
(2.34)

where $\psi^M = s^M (1+\Xi/\eta^M)/(1+\Theta/\varepsilon)$ and $\psi^Z = s^Z/(1+\Theta/\varepsilon)$ are the value shares of the material and non-material input respectively, η^M the supply elasticity of the material input, and $\mu^Z = (\partial (\Xi/\eta^M)/\partial \eta^M)(\partial \eta^M/\partial W^M)$ is equivalent of μ^{Θ} on the material input market, the change in the mark-down. Applying comparative static and simulations, Weldegebriel (2004) reported that no clear conclusion on the effect of oligopoly and oligopsony power can be drawn without knowledge of the functional forms. Under the assumptions of certain values for the required elasticities and the presence of market power on the upstream and downstream market the price transmission could even exceed the competitive level (Weldegebriel, 2004).

Based on the EDM approach, but not solving for the price transmission elasticity is the model of Lloyd et al. (2009). Like in the previous EDM models, n firms produce a homogenous output by using two inputs in fixed proportions and with CRS. Furthermore it is assumed that buyer exercise market power on the material input market. The first-order condition of the profit function of the i^{th} firm with respect to the material input level, aggregating across all firms, and assuming linear demand and supply functions yields,

$$P = M + W^{M} + \Xi \frac{\partial W^{M}(\bullet)}{\partial X^{M}}Q$$
(2.35)

where $M = \alpha_0 + \alpha_1 W^Z$ represents a composite variable of all other marketing cost, a linear function of the constant α_0 and the cost of the non-material input is given by $\alpha_1 W^Z$. Using (2.35) as the derived demand for material input in a system with output supply and demand as well as material input inverse supply functions, explicit solutions for the endogenous variable *P*, *Q*, and W^M can be derived. Deducing from this explicit solutions, the margin ($P - W^M$) equals in the absence of oligopsony power *M*. Consequently, a simple test for the presence of oligopsony power is to estimate an unrestricted linear representation of the margin equation, for example formulated as,

$$P = \beta_0 + \beta_1 W^M + \beta_2 M + \beta_3 Y + \beta_3 Z$$
(2.36)

and test the hypothesis of perfect competition (here: $\beta_3 = \beta_4 = 0$), since in the absence of market power exogenous supply and demand shocks should not affect the price margin, but in the presence of oligopsony power the shifts should be significant. Lloyd et al. (2009) applied their 'first-pass' test for oligopsony power to the UK food retailing sector and nine basic food product groups. The margin equation (2.36) is embedded in a vector error correction model (VECM) to account for cointegration between the variables. In seven out of nine cases, the hypothesis of perfect competition can be rejected (Lloyd et al., 2009).

3.3.3 Criticism

Apart from the estimation issues of pre-cointegration period price transmission analysis the price transmission framework has been criticized due to a series of critic points that heavily undermine its appropriateness to the analysis of market power.

One certainly big issue is the choice of time series of price transmission studies is already biased. Meyer and von Cramon-Taubadel (2004) as well as Peltzman (2000) surveyed several hundreds of APT studies, 205 and 242 respectively, and found that 48% and around 2/3 of these studies rejected the hypothesis of symmetric price transmission. Consequently, asymmetry and thus market power abuse seem to be the rule. Nevertheless, in reality anti-trust agency cases are rather the exception (Acharya et al., 2011).

Furthermore, the choice of time horizon as well as frequency plays a major role to determine the validity of the study. Meyer and von Cramon-Taubadel (2004) argued that known changes in the market power conduct must have occurred in the investigated time period, otherwise it would not be valid to compare price transmission in the competitive scenario to settings with different levels of market power due to the lack of variation in the 'treatment variable' (Meyer & von Cramon-Taubadel, 2004). Moreover, the frequency of the data has to fit the frequency of the adjustment process. If the price transmission occurs over a couple of days, monthly data will not enable the researcher to detect any asymmetries (Boyd & Brorsen, 1988).

The price transmission framework has experienced large popularity, as shown by the surveys of Peltzman (2000) and Meyer and von Cramon-Taubadel (2004), among economists, since it allows researchers to investigate market power with relatively low data requirements compared to NEIO models. Mostly only easily available data on prices are required. Nevertheless, the major drawback of these studies regarding the investigation of market power is that they hypothesize that the source of the imperfect price transmission is market power but cannot validate their hypothesis with economic theory. As shown by the EDM, less complete price transmission can result from other sources such as non-constant returns-to-scale (Bliss, 1988).

While the EDM approach derive the price transmission elasticity from a theoretic model and thus is able to determine the effect of market power on the price transmission, as shown in the previous section, inference on the effect of market power is heavily dependent on the choice of functional form and diverse technology assumptions lead to completely different outcomes. Consequently, knowledge of the true technology is required to draw valid conclusion. This requires estimation of a system of supply and demand functions similar to NEIO techniques. Apart from losing the appeal of low data requirement, also the same criticism of NEIO model applies then to the EDM framework.

3.4 SUMMARY

In the previous chapter, the three most prominent methodologies to the investigation of market power were presented, the SCPP, NEIO, and price transmission framework. The standard theoretic concepts and the advantages and disadvantages of each approach were presented and discussed.

The SCPP derives from accounting data a measure of industries' performance, e.g. PCM, and tries to explain difference between sectors by market structure measures, such as concentration ratios. Nevertheless, due to endogeneity issues between the chosen variables, accounting data not being truly representative for economic data, and a never ending discussion of which concentration ratio best depicts the market structure the SCPP has submerged into economic oblivion and is hardly ever applied in market power analysis today.

While the SCPP almost completely disappeared from the economic landscape, NEIO and price transmission analysis still enjoy great popularity in empirical applications today. The NEIO concept models firms' behavior and is able to estimate a market power conduct parameter, which nests all forms of market structure from competitive over oligopoly to monopoly, and thus the effect of market power on the market outcome. Furthermore different market settings and how they affect input and output markets, such as oligopoly, oligopsony, or oligopoly and oligopsony, have been theoretically developed and empirically applied. Nevertheless, NEIO models' data requirement are high, the choice of functional form severely affects the estimates, and if modeled dynamically strong assumptions have to be drawn to limit the possible number of equilibria. In addition, while a measure of the CPM and thus market structure is obtained, the source of the market power is never clearly identified.

In contrast, the price transmission framework only depends on price time series. The price time series are investigated for possible asymmetries and thus imperfect price transmission. One of the sources of this imperfect price transmission is hypothesized to be market power. Consequently, while the simplicity of the framework is desirable, it lacks any theoretic economic link that confirms this hypothesis. The EDM framework creates this link, however if estimated it is very similar to NEIO models, if not the same, and their demanding data requirements. The 'first-pass' test developed by Lloyd et al. (2009), which can somewhat be seen as a hybrid of EDM and NEIO, allows to test for oligopsony power with rather low data requirements. Nevertheless, the extent of market power is not measured and the authors themselves state that it can be seen as a first empirical tool to the investigation of market power (Lloyd et al., 2009).

Overall, this theory review has shown that for the investigation of market power either the data requirements are high and the models highly complex, like in the case of NEIO, but an index of market power can be measured, or the data requirements are rather low and the theoretic models rather simple, like in the case of the price transmission framework, but market power or its extent are not clearly identified. In the next chapter, we will provide an overview of all known empirical applications of the presented concepts to the milk and dairy markets.

4 EMPIRICAL APPLICATIONS IN THE DAIRY SECTOR

The following section deals with empirical applications of the previously described theoretic models of NEIO and price transmission analysis. Because the dairy agri-food supply chain is the focus of this work, we will limit the overview to applications in this field. For more comprehensive overviews the reader is referred to Bresnahan (1989) and Perekhozhuk et al. (2016) as well as Meyer and von Cramon-Taubadel (2004) and Frey and Manera (2007) for a review of empirical studies in NEIO and APT respectively.

4.1 NEIO

Perekhozhuk et al. (2016) list 38 empirical studies that estimated market power in agri-food and related industries. Eight of these studies examine milk and dairy product markets. Table 10 presents an updated version of the survey of Perekhozhuk et al. (2016) and presents all, to our knowledge, 16 empirical applications of the NEIO approach to the dairy supply chain or one of its submarkets.

Even though these studies are very diverse in their methodological approach, e.g. assuming fixed proportions technology (Gohin & Guyomard, 2000) or accounting for non-constant returns-to-scale (Bhuyan & Lopez, 1998), and scope, raw milk market (Perekhozhuk et al., 2013) to PDO cheese markets (Sckokai et al., 2013), still common grounds can be identified. Each study starts out by either following Bresnahan's (1982) GIM or Appelbaum's (1979; 1982) PTA method and in some cases modify these to either allow for product differentiation (Hovhannisyan & Bozic, 2013) and/or oligopsony market structures (Hockmann & Vőnecki, 2009). The following sections categorize the listed applications in to oligopoly, oligopoly/ oligopsony, and oligopsony models and briefly describe notable differences or modifications to the standard methods depicted in theory review chapter.

4.1.1 Oligopoly

Nine of the listed 16 studies assume an oligopolistic structure at either the processor (6) or retail (3) level. Upstream markets are assumed to be competitive. Liu et al. (1995) allow for two different regimes in their application to the U.S. dairy industry, where one of the regimes is defined as 'booming period' and the other as 'government supported'. In the first regime the market price is above the government intervention level and in the second the market price equals the intervention price. Furthermore, they distinguish between dairies that produce fluid milk and other dairy products. Both processor types act to some extent oligopolistic with the dairy products producer abusing their market power more with a CE 0.176 compared to 0.100 for the fluid milk producers' CE.

Bhuyan and Lopez (1998) modified Appelbaum's (1979; 1982) approach to allow for non-constant returns-to-scale and demand price elasticities in the food processing industry. The conjectural variation for the fluid milk sector demonstrates non-competitive market behavior by dairies with a value of 0.182 over the time period from 1972 to 1987.

Hovhannisyan and Gould (2012) as well as Hovhannisyan and Bozic (2013) alter the GIM approach to allow for differentiated products. The products, milk and yogurt respectively, are differentiated into the subgroups national brand, store brand, and other national brands. Furthermore regional markets, U.S. metropolitan city regions, are investigated separately. While Hovhannisyan and Gould (2012) do not find evidence for retailer market power for the store and national milk brands, they report low market distortions with a CV of 0.022 in the other national brand category. Hovhannisyan and Bozic (2013), on the other hand, do not explicitly report the estimates for the CPM, but state that retailers mark-up yogurt depending on the city and brand between 1.4% and up to 12.5% from the competitive price level.

4.1.2 Oligopoly/Oligopsony

Most of the applications that investigate whether one market stage abuses oligopoly power on the output market and oligopsony power on the input market are based on the approach of Schroeter (1988). The firsts to consider that retailers might exercise buyer power were Gohin and Guyomard (2000) in their analysis of the French retail sector regarding three main food product groups over the time period 1977 to 1993. While a CPM close to the competitive level is reported, the Lerner index for dairy products takes a value of 0.20 due to the inelastic demand.

Sckokai et al. (2013) were the first to apply a dynamic CPM model to the dairy sector. In their empirical study of the Italian PDO cheese market, they based their approach on Perloff et al. (2007) with fixed proportion technology, stock of cheese as the state variable, and two outputs, Parmiggiano Reggiano and Grand Padano. While significant levels of oligopoly conduct were identified, CPM of 0.25 for both products, the estimate for oligopsony market behavior was not statistically significant from zero (Sckokai, et al. 2013).

4.1.3 Oligopsony

Most empirical studies that either model an oligopoly or an oligopoly/oligopsony were conducted in industrialized countries, in particular the USA. Pure oligopsony models, always assuming a non-competitive raw milk and competitive output market, were only applied to former socialist, e.g. Hungary (Perekhozhuk et al., 2013), or emerging countries, e.g. Brazil (Scalco & Braga, 2014). Most of the oligopsony models are based on the work of Muth and Wohlgenant (1999) and use industry aggregated data. One exception is Perekhozhuk et al. (2013), who use plant level data to investigate the Hungarian raw milk market from 1993 to 2006 and reported rather high levels of oligopsony behavior with a CPM value of 0.2219. The other oligopsony studies of Hockmann and Vőnecki (2009) for Hungary, Perekhozhuk et al. (2015) for Ukraine, and Scalco and Braga (2014) for Brazil, found lower levels of market imperfections (see Table 10).

4.2 APT

In this section an overview of APT empirical applications to the dairy supply chain is presented (see Table 11). To our knowledge no empirical applications of the EDM framework to the dairy sector exist. However, most of the studies presented here derive a simplistic measure for the price transmission elasticity by assuming some relationship like in equation (2.25) or the (cumulative) sum of β_1 and β_2 equal τ (e.g. Bolotova & Novakovic, 2015; Dudová & Bečvařová, 2015). By incorporating additional cost variables to the price relationship as in the margin equation similar to Lloyd et al. (2009), where $P - W^M = f(W^M, Q, MC) - W^M$, it is also possible to test for CRS (e.g. Romain et al., 2002).

Evidently, almost all empirical studies find positive asymmetries. Even though APT, as mentioned before, is not able to detect market power directly as the source of asymmetry, most studies explained their results with an oligopoly or oligopoly/oligopsony at the downstream level, which is able to increase the spread between upstream and downstream price (e.g. Rumánková, 2012). Three exceptions exist. While Dudová and Bečvařová (2015) in their analysis of the farm-to-processor and processor-to-retail price transmission for milk and butter in the Czech Republic from 2000 to 2013 found evidence for APT in three out of these four relationships, they could not statistically prove that the processor-to-retail relationship in regards to butter is characterized by APT. Romain et al. (2002), on the other hand investigated the U.S. milk supply chain and report positive long-run price asymmetries for the period before 1991, but full transmission from 1991 to 1997. In contrast, Bakucs et al. (2012) find no evidence for asymmetry in price transmission between the Hungarian raw milk and retail prices for the entire investigated time period from 1995 to 2007.

Surprisingly, even though ECM techniques were already applied to price transmission analysis by von Cramon-Taubadel and Fahlbusch (1994), nine of the listed studies appeared after this paper and still used some form of the Houck representation. Authors either reported that the cointegration test was rejected (e.g. Lass, 2005) or did not consider it at all (e.g. Bolotova & Novakovic, 2015; Romain et al., 2002). Capps and Sherwell (2007) in their study of the milk price transmission for several large U.S. cities made the effort to compare both methods. They concluded that both methods yielded almost identical statistical results (Capps & Sherwell, 2007).

Furthermore, most of the APT studies use monthly data. As mentioned before, in the criticism of price transmission analysis, the choice of frequency of the data is of utter importance to the analysis, since data with a lower frequency than the adjustment process will not be able to reveal asymmetries in the price transmission process (Boyd & Brorsen, 1988). Serra and Goodwin (2003) in their analysis of the price transmission along the Spanish dairy supply chain examine whether the choice of weekly or monthly data will affect the result. When only data with a weekly frequency was used, asymmetries were stronger and only revealed for dairy products with long-shelf life, like cheese. However, lowering the data frequency to be monthly, also revealed APT for perishable dairy products, like fresh milk, and lowered the asymmetry. The author explained that farm prices in Spain appear to be adjusted on a monthly basis instead of a weekly one and retail and farm prices for lower value-added milk products, mostly perishable ones, are tighter connected due to scarcity of raw milk, less market power in the 'white line' due to the absence of product differentiation and cost-minimizing strategies, and thus the price transmission more symmetric (Serra & Goodwin, 2003).

4.3 'FIRST-PASS' TEST

In contrast to the NEIO and APT methodology the rather young 'first-pass' test by Lloyd et al. (2009) has to our knowledge been only applied empirically to the dairy sector three times. Lloyd et al. (2009) applied their developed test to the UK retail sector for main food products from January 1990 to October 2001. Even though oligopsony power was found for the procurement of meat products, dairy products were no subject of market power exertion according to the study. In contrast to this, Fałkowski (2010) and Madau et al. (2016) in their empirical applications of the 'first-pass' test, found oligopsony power in the Polish, from January 1996 to December 2006, and Italian, from January 2000 to December 2013, milk supply chain respectively.

4.4 SUMMARY

In general, the dairy supply chain has experienced great popularity as a sector worth studying in NEIO and price transmission analysis. Main country of focus has been the U.S. dairy sector, but recently the European dairy sector has been studied more and more frequently, in particular Eastern European countries. A reason for this might be that the dairy supply chains of these countries have undergone dramatic restructuring since the abolishment of socialism and thus are an interesting case for the empirical studies (Bakucs & Fertő, 2005).

Empirical applications of both strains of methodology in general report market power abuse in the corresponding dairy supply chain. Nevertheless, all APT studies only assume that market power could be the source of the found asymmetries in price transmission and cannot empirically prove the validity of their assumption. Furthermore, it can be criticized, in the fashion of Peltzmann (2000), since all APT studies except for one, the study on the Hungarian milk market and its farm-to-retail price transmission by Bakucs et al. (2012), find some sort of asymmetry in the respective dairy supply chain, that asymmetries are rather the rule than the exception and thus market power should be omnipresent.

NEIO studies on the dairy supply chain, on the other hand, derive via the structural modelling of markets with supply and demand functions an estimate of market power conduct. Even though, almost all of the listed studies find evidence for some level of market power abuse, the estimates for the CPM are rather low and dairy supply chain markets seem to be closer to the competitive than the monopolistic or monopsonistic market structure.

The 'first-pass' test, developed of Lloyd et al. (2009) as a synergy of EDM and NEIO, has only been applied empirically in three cases. While this test showed that the Polish and Italian retail sector in the recent past used market power in their procurement of milk, it was not able to identify oligopsony power of the UK retail sector towards the upstream dairy chain actors. Even though, the 'first-pass' test allows to test the presence of oligopsony power, no estimate of the level of market power or the mark-down is obtainable.

Furthermore, as described before, although agri-food supply chains are more and more retailer dominated and the threat of buyers power towards dairy farmers is eminent due to the perishableness of raw milk, no study or theoretic model exists that accounts for a subsequent oligopsony at the raw milk and dairy output market. The following theoretic model and empirical application to the German dairy supply chain will try to fill this gap in the literature.

year	author	country	product	time	level	aggregation	frequency	method	model	CE/CV
2010	Ahn & Lee	Republic of Korea	infant formula	1975-2009	processor	national	annual	ØIM	oligopoly	0.29
1998	Bhuyan & Lopez	USA	milk	1972-1987	processor	national	annual	PTA	oligopoly	0.182
2012	Cakir & Balagtas	USA	dairy products	2000-2007	processor/ retailer	regional	annual	PTA	oligopoly/ oligopoly	≤0.021a/≤0.0023a
2005	Chidmi et al.	USA	milk	1996-2000	retailer	regional	weekly	GIM	oligopoly	0.1663
1999	De Mello & Brandao	Portugal	milk	1962-1991	processor	national	annual	PTA	oligopoly	0.05403
2000	Gohin & Guyomard	France	dairy products	1977-1993	retailer	national	annual	GIM	oligopoly/ oligopsony	-0.0187
2009	Hockmann & Vőnecki	Hungary	milk	1998-2006	processor	national	monthly	PTA	oligopsony	0.05
2012	Hovhannisyan & Gould	USA	milk	2001-2006	retailer	regional	weekly	GIM	oligopoly	0.022
2013	Hovhannisyan & Bozic	USA	yogurt	2001-2007	retailer	regional	weekly	GIM	oligopoly	NA
1995	Liu et al.	USA	dairy products	1975-1992	processor	national	quarterly	PTA	oligopoly	0.1
			milk	1975-1993	processor					0.176
2009	Mérel	France	Comté cheese	1985-2005	processor	national	quarterly	GIM	oligopoly	0.001
2013	Perekhozhuk et al.	Hungary	milk	1993-2006	processor	firm	annal	PTA	oligopsony	0.2219b
2015	Perekhozhuk et al.	Ukraine	milk	1996-2003	processor	regional	monthly	РТА	oligopsony	0.1475b
2012	Salhofer et al.	Austria	milk	1997-2008	retailers	national	monthly	PTA	oligopoly/	0.011/NA
			cheese						oligopsony	NA/NA
			Butter/others							NA/0.102
2014	Scalco & Braga	Brazil	milk	1997-2011	processor	regional	annual	GIM	oligopsony	0.01b
2013	Sckokai et al.	Italy	PDO cheese	2002-2008	processor	national	monthly	dynamic	oligopoly/ oligopsony	0.25/none
1993	Suzuki et al.	Japan (Hokkaido)	dairy products	1979-1988	processor	regional	annual	GIM	oligopoly	0.12a
		Japan (other regions)	dairy products	1979-1989	processor	regional	annual	GIM	oligopoly	0.23a
Source: c	own illustration, base	d on Perekh	ozhuk et al. (201	6). Notes: ^a r	naximum value in	region or ove	r time report	ted; ^b aver	age value r	eported.

Table 10: NEIO studies in the dairy sector.

asymmetries	positive	positive	positive	none	positive	positive	positive	positive	positive	positive	positive	positive	positive	none	positive	positive	positive	positive	positive	positive	positive	positive	positive	positive	positive	positive	positive	positive & none	positive	positive	positive	positive	ne changes, the
model	ECM	Houck	ECM (regime)	ECM	ECM	Houck	Houck	Houck & ECM	Houck	ECM (regime)	Houck				ECM	ECM/STAR1 (regime)	ECM	Houck	ECM		Houck	Houck	ECM (regime)	ECM (regime)	Houck		ECM (regime)	Houck	ECM	threshold	ECM (regime)		alues and sudden regi
frequency	monthly	monthly	monthly	monthly	monthly	monthly	monthly	monthly	monthly	monthly	monthly				monthly	monthly	monthly	monthly	monthly		monthly	monthly	weekly	weekly	monthly		monthly	monthly	monthly	weekly & monthly	monthly		olds to be 'hard' v
relationship	farm-wholesale	farm-retail	farm-retail	farm-retail	farm-retail	farm-retail	farm-wholesale	regional	farm-retail	wholesale-retail	farm-processor	processor-retail	farm-processor	processor-retail	farm-retail	farm-retail	farm-retail	farm-retail	farm-processor	processor-retail	farm-retail	farm-retail	wholesale-retailer	wholesale-retailer	processor-retail	farm-processor	farm-retail	farm-retail	farm-wholesale	farm-retail	farm-retail		stead of the thresh
time	1991-2011	1987-1998	1987-2006	1995-2007	1995-2007	1982-2008	2000-2014	1994-2002	1999-2003	1980-2001	2000-2013				1995-2006	2000-2012	2002-2014	1971-1981	2004-2011		1982-1996	1982-2001	2005-2008	2005-2008	1995-2003		1989-2009	1980-1997	2007-2011	1994-2000	2000-2010		ion model (ins
product	milk	milk	cheese	milk	milk	milk	cheese	milk	milk	butter	milk		butter		milk	cheese	milk	dairy products	milk		milk	milk	milk & butter	milk	dairy products		milk	milk	milk	dairy products	milk	cheese	sition autoregress
country	Panama	Brazil	USA	Hungary	Poland	USA	USA	USA	USA	USA	Czech Republic				Poland	USA	Russia	USA	Slovakia		USA	USA	Germany	Germany	Hungary		Greece	USA	Czech Republic	Spain	USA		s: ¹ Smooth trans
author	Acosta & Valdés	Aguiar & Santana	Awokuse & Wang	Bakucs et al.		Bolotova & Novakovic	Bolotova & Novakovic	Capps & Sherwell	Carman & Sexton	Chavas & Mehta	Dudová & Bečvařová				Fałkowski	Hahn et al.	Kharin	Kinnucan & Forker	Lajdová & Bielik		Lass et al.	Lass	Loy et al.	Loy et al.	Popovics		Rezitis & Reziti	Romain et al.	Rumánková	Serra & Goodwin	Stewart & Blanely		n illustration. Note
year	2014	2002	2009	2012		2012	2015	2007	2005	2004	2015				2010	2016	2015	1987	2015		2001	2005	2015	2016	2015		2011	2002	2012	2003	2011		Source: ow

Table 11: APT studies in the dairy sector.

STAR allows for gradual, soft regime transition (Hahn et al., 2016)).

5 THEORETICAL MODEL

The description of the German dairy supply chain's structure as well as the report of the German anti-trust agency's sector study suggest the possibility of oligopsonistic behavior on the raw milk and dairy output market. Since the literature review has shown that no model that accounts for oligopsonistic conduct on subsequent markets, input and output, of a supply chain so far exists, such a theoretical model will be developed in the following section. The model is based on Muth and Wohlgenant's (1999) oligopsony model, described in Subsection 3.2.2, but adds the prospect of oligopsony conduct also to the output market. Furthermore pricing equations with a procedure similar to Lloyd et al. (2009) will be derived in the second part.

To limit the complexity of the model and assure empirical applicability, a series of assumptions has to be drawn first. We start by assuming that raw milk and dairy output are homogenous products. For raw milk this seems plausible, since raw milk is an unprocessed product and the importance of a possible differentiated product, such as organically produced raw milk, remains marginal.

Dairy products, on the other side, is a product category that comprises a vast number of product subgroups, from cheese to milk powder, further distinguished in brand and standard products. Nevertheless, it has been argued that brand and standard dairy products can be seen as perfect substitutes as well as products of other differentiation within one dairy product subgroup, e.g. fat-reduced milk and standard milk, since no real quality difference seems to exist (Bundeskartellamt, 2009; Davis et al., 2009). Since products can be defined as perfect substitutes, at least the product groups can be stated to be homogenous products (Stigler, 1964). Aggregation is consequently achieved by forming a weighted average dairy product price using selected individual dairy product group prices of Emmentaler cheese, butter, and skim milk powder. The procedure is further described in the data description section.

Additionally, export markets are hypothesized to be of less to no significance. Even though, this is a general quite restrictive assumption made in NEIO models to define clear markets and limit the complexity (Sckokai et al., 2013), for the German dairy supply chain this hypothesis can be proven to be a good approximation of reality. While regarding raw milk this is done at ease, since German dairies sourced around 93.8% of raw milk domestically in 2015, for the dairy product market the need for more elaboration exists. Around 49% of milk equivalent was exported in 2015, but only accounted for 28% of total revenue generated from processing raw milk. Furthermore, large shares of the German

dairies' exports are actually marketed via German retailers. Unfortunately no data is available, but it has been stated by the German anti-trust agency that regarding milk for consumption exports, two retailers market more than 50% of the quantity (Bundeskartellamt, 2014). Consequently, retailers seem to almost exclusively market dairy product exports or at least have a dominant position in this regard.

Furthermore, it is assumed that at no level of the dairy supply chain any form of stockpiling is conducted. Raw milk and dairy products, in particular the first one, are highly perishable products and have to be refrigerated to be stored over longer durations. Since this a costly form of storage, the effect of storage is assumed to be marginal and consequently not modeled (Loy et al., 2016; Sckokai et al., 2013). With the exclusion of imports and exports as well as storage, the quantity of dairy output cannot exceed the raw milk quantity produced in time period t.

Moreover, it is assumed that no agent involved in the transformation of the product from raw milk to marketed dairy products faces any form of adjustment costs. Consequently, the market power measure of the later derived model is 'static'. As mentioned in the theory review, the static NEIO model has been criticized and led to establishment of dynamic models. However, due to the complexity of dynamic models and the notion that the static outcome in some cases approximates the outcome of dynamic models or even coincides, this assumption was considered feasible, in particular, since this assumption will be further relaxed in the methodology.

In Subsection 3.2.3 it was noted that NEIO models have been heavily criticized for their choice of, mostly linear, functional form and assumptions of the applied technology in the industry such as fixed proportions technology (Schroeter, 1988). To avoid this criticism and any limiting assumptions on the technology, second order-differential quadratic forms are used for any cost, production, or revenue function stated in the following sections (Chambers, 1988). This allows approximating the true technology without a prior knowledge, and does not set any restrictions on "homotheticity, homogeneity or the elasticities of substitution between factors" (Gollop & Roberts, 1979). Furthermore, flexible functional forms also provide the necessary non-linearity to derive the CPM and avoid multicollinearity issues (Bresnahan, 1989; Perloff & Shen, 2012). Explicit forms of all functions used in the subsequent sections are presented in the appendix (see (A1) to (A5)).

5.1 OLIGOPSONY MODEL OF TWO MARKETS

In the following section the structural model of oligopsony power on two subsequent markets of a supply chain is developed. Because we defined both products, the material input and output, as homogenous the competition is in quantities as strategic variables (Sexton, 2000). With the previous stated assumptions in mind, we assume that the processing industry consist of n firms sourcing their material input exclusively on the material input market and selling their output to m retailers on the output market. Given the cost function of farmers and the assumption that farms are price takers on the material input market, due to their low bargaining power, we can derive the inverse supply function of material input, since their marginal costs equal the farm gate price for the material input W^M ,

$$W^{M} = \frac{\partial C(X^{M}, W_{h}^{F}, T^{F})}{\partial X^{M}} = S^{M^{-1}}(X^{M}, W_{h}^{F}, T^{F})$$
(5.1)

where W^{F}_{h} is a vector of prices of h non-material inputs involved in the production of the aggregated farm output X^{M} , and T^{F} is a trend variable depicting technical change at the farm level. Given this supply relation we can formulate processor i's profit function

$$\pi_{i}^{P} = r_{i}^{P}(P, x_{i}^{M}, x_{ki}^{P^{*}}, T^{P}) - W^{M} x_{i}^{M} - W_{ki}^{P} x_{ki}^{P^{*}}$$
(5.2)

where *P* demarks the output price, $r_{i}^{P}(\bullet)$ is the revenue function of the *i*th processor, $x_{ki}^{P} = x_{ki}^{P}(x_{i}^{M}, P, W_{ki}^{P})$ is a vector of *k* non-material inputs used in the production of the output at their optimal quantity conditional on processor *i*'s choice of material input x_{i}^{M} , W_{ki}^{P} a vector of the corresponding prices of the *k* non-material inputs, and T^{P} captures technological change at the processing level. Profit maximization with respect to the choice of x_{i}^{M} yields the following first-order condition:

$$\frac{\partial \pi_i^p}{\partial x_i^M} = \frac{\partial r_i^P(\bullet)}{\partial x_i^M} + \left(\frac{\partial r_i^P(\bullet)}{\partial x_{ki}^{P^*}} - W_{ki}^P\right) \frac{\partial x_{ki}^{P^*}}{\partial x_i^M} - W^M - \varkappa_i^P \frac{S^{M^{-1}}(\bullet)}{\partial x_i^M} \chi_i^M = 0$$
(5.3)

where $\varkappa^{P}_{i} = \partial X^{M} / \partial x M_{i}$ is the *i*th firm's CV. Under the assumption of procurement of the non-material inputs x^{P}_{ki} in competitive input markets, and aggregating across firms by averaging over all dairies' marginal product (5.3) can be reduced and rearranged to yield:

$$\underbrace{\frac{W^{M}\left(1+\frac{\Xi^{P}}{\eta^{F}}\right)}_{\textit{vperceived}'MFC}} = \underbrace{\frac{\partial R^{P}(P, X^{M}, X^{P}_{k}(X^{M}, P, W^{P}_{k}), T^{P})}{\partial X^{M}}}_{MRP}$$
(5.4)

where the observed material input price plus a relative mark-down Ξ^P/η^F is the 'perceived' MFC, which equals its marginal revenue of product (MRP) and with perfect competitive output market MRP = VMP, where $\Xi^P = \sum (\partial X^M / \partial x^M_i)(x^M_i / X^M)$ and $\eta^F = (\partial S^M(\bullet) / \partial W^M)(W^M / X^M)$ is the elasticity of material input supply. Ξ^P , as discussed in the theory review section, can be either interpreted as the average industry's conjectural elasticity, which measures the response in total industry input quantity to a change in the *i*th processor's input level, or as one of the elements that define the wedge between the material input price and its MRP. In general, it can be seen as an index which represents the processing industry's ability to mark-down the price of the material input given η^F . The range of the index is zero to one. While zero signals perfect competition, one stands for monopsony. Values in between the extremes represent some level of oligopsony with 1/n as the Cournot outcome (Bresnahan, 1989).

The downstream market for the output product is modelled in a similar fashion as the upstream material input market model, but using Morrison Paul's (2001) cost side approach to the oligopsony model. The supply function S^Q of the output Q is derived by differentiating the processing industry's profit function with respect to the output product price P,

$$\frac{\partial \Pi^{P}(P,W^{M},W_{k}^{P},X^{M},X_{k}^{P}(X^{M},P,W_{k}^{P}),T^{P})}{\partial P} = Q$$
(5.5)

where Π^P is the total processing industry's profit. Given the supply of the output product (5.5), the j^{th} retailer's demand can be derived. Profit maximization behavior implies that j^{th} retailer's cost function differentiated with respect to the choice of output product level as a material input as well as keeping the quantities of the I non-material inputs used in the marketing process again at their optimal quantity $q^R_{lj}^* = q^R_{lj}(q_j, P, W^R_{lj})$ conditional on the j^{th} retailer's chosen input level of the processing industry's output q_j , yields

$$\underbrace{\frac{\partial c_j^R(q_j, P, q_{lj}^R(q_j, P, W_l^R), W_l^R, T^R)}{\partial q_j}}_{\textit{perceived'MFC}} - \varkappa_i^R \frac{\partial S^{Q^{-1}}}{\partial q_j} q_j = \underbrace{P}_{VMP}$$
(5.6)

where the 'perceived' MFC of the output product equals the output price. $\kappa^{R}_{i} = \partial Q/\partial q_{j}$ represents retailer j's CV, W^{R}_{lj} is a price vector for *l* non-material inputs required for marketing, and T^{R} stands for the technological change at the retail level. Aggregating across all retailers through averaging across all retailers' marginal costs gives us

$$\frac{\partial C^{R}(Q,P,Q_{l}^{R}(Q,P,W_{l}^{R}),W_{l}^{R},T^{R})}{\partial Q} = P\left(1 + \frac{z^{R}}{\eta^{P}}\right)$$
(5.7)

where the 'perceived' MFC equals the output price plus a relative mark-down Ξ^R/η^P . Consequently, as on the material input market, $\Xi^R = \sum (\partial Q/\partial q_j)(q_j/Q)$ is the retailers' ability to mark-down the price of the processing industry's output given the supply elasticity of the output $\eta^F = (\partial S^Q(\bullet)/\partial P)(P/Q)$. Interpretation is according to Ξ^P .

The equations (5.1), (5.4), (5.5), and (5.7) are the supply and demand relations respectively for the material input and output market which incorporate the possibility of oligopsonistic behavior and provide the equilibrium outcome on both markets (see Figure 7). The system is very similar to the model of Muth and Wohlgenant (1999), but includes oligopsony power on the output market, and would still require data on material input and output quantity. However, in the following section we derive pricing equations using theoretical considerations used in the EDM framework, which do not incorporate quantity variables anymore.

Figure 7: Monopsony *m* and competitive *c* outcome on the material input (left) and output market (right).



Source: own illustration.

5.2 PRICING EQUATIONS AND DETERMINING THE LEVEL OF MARKET POWER

Due to the assumption that all non-material inputs used along the supply chain to transform the product are procured on competitive markets, the respective

prices are exogenous and only the material input and output quantities X^M and Q as well as the respective prices W^M and P remain as endogenous variables.

The equations for material input supply (5.1), processors' demand for material input (5.4), output supply (5.5), and retailers' demand for output (5.7) form a system, which allows determining the simultaneous partial market equilibria and the explicit solutions for the endogenous variables,

$$X^{M} = f_{X^{M}}(W_{h}^{F}, W_{k}^{P}, W_{l}^{R}, T^{F}, T^{P}, T^{R}, \Xi^{P}, \Xi^{R})$$
(5.8)

$$W^{M} = f_{W^{M}}(W_{h}^{F}, W_{k}^{P}, W_{l}^{R}, T^{F}, T^{P}, T^{R}, \Xi^{P}, \Xi^{R})$$
(5.9)

$$Q = f_Q(W_h^F, W_k^P, W_l^R, T^F, T^P, T^R, \Xi^P, \Xi^R)$$
(5.10)

$$P = f_P(W_h^F, W_k^P, W_l^R, T^F, T^P, T^R, \Xi^P, \Xi^R)$$
(5.11)

where the remaining endogenous variables solely depend on exogenous variables, time trends that depict the technology change at each stage, and the measures for oligopsony power on the respective markets Ξ^P and Ξ^R (for explicit forms see in the appendix (A6)).

Solving the explicit solution of P (5.11) once for one of the non-material input prices at the processing level $W^{F}{}_{h}$ and once at the retail level $W^{R}{}_{l}$ as well as substituting these solutions into the explicit solution for W^{M} permits to derive two pricing equations, denoted in the rest of the paper as farm-processor equation (FPE), from which all the retail level specific variables $W^{R}{}_{l}$ and T^{R} are excluded, and as processor-retail equation (PRE), from which the farm specific variables $W^{F}{}_{h}$ and T^{F} are omitted (see in the appendix). For reasons that will become obvious in the methodology section, the FPE is rearranged and solved for the price of non-material inputs at the farm level $W^{F}{}_{h}$ and the PRE is similarly solved for the price of non-material inputs at the retail level $W^{R}{}_{l}$. To simplify the notation, only one non-material input per stage is represented and the trends depicting technological change at each individual stage are merged to one time trend T. Thus, the FPE (5.12) and the PRE (5.13) take the form,

$$W^{F} = \underbrace{\left(\omega_{0}^{C} + \omega_{1}^{C}\Lambda^{\Xi^{P}}\right)}_{\beta_{1}^{C}} + \underbrace{\left(\omega_{0}^{W^{M}} + \Lambda^{\Xi^{P}}\right)}_{\beta_{1}^{W^{M}}} W^{M}$$
$$+ \underbrace{\left(\omega_{0}^{P} + \omega_{1}^{P}\Lambda^{\Xi^{P}}\right)}_{\beta_{1}^{P}} P + \underbrace{\left(\omega_{0}^{W^{P}} + \omega_{1}^{W^{P}}\Lambda^{\Xi^{P}}\right)}_{\beta_{1}^{W^{P}}} W^{P} + \underbrace{\left(\omega_{0}^{T} + \omega_{1}^{T}\Lambda^{\Xi^{P}}\right)}_{\beta_{1}^{T}} T$$
(5.12)

$$W^{R} = \underbrace{\left(\tau_{0}^{C} + \tau_{1}^{C}\Lambda^{\Xi^{R}}\right)}_{\beta_{2}^{C}} + \underbrace{\left(\tau_{0}^{W^{M}} + \tau_{1}^{W^{M}}\Lambda^{\Xi^{R}}\right)}_{\beta_{2}^{W^{M}}} W^{M^{*}}\Omega^{\Xi^{P}} + \underbrace{\left(\tau_{0}^{P} + \tau_{1}^{P}\Lambda^{\Xi^{R}}\right)}_{\beta_{2}^{P}}P + \underbrace{\left(\tau_{0}^{W^{P}} + \tau_{1}^{W^{P}}\Lambda^{\Xi^{R}}\right)}_{\beta_{2}^{W^{P}}} W^{P} + \underbrace{\left(\tau_{0}^{T} + \tau_{1}^{T}\Lambda^{\Xi^{R}}\right)}_{\beta_{2}^{T}}T$$

$$(5.13)$$

where the ω s and τ s are parameters consisting of the parameters of the supply and demand relations and Λ s and Ω are parameters containing the parameters of the supply and demand relations as well as the oligopsony power measures Ξ^P and Ξ^R (for the detailed forms see in the appendix (A8) and (A9)). W^{M^*} is a transformed variable obtainable after knowledge on the parameter estimates of the FPE². Interestingly, if both markets are competitive, consequently Ξ^P and Ξ^R equal zero, and also the Λ s vanish from the FPE and PRE. However, in this case, Ω would not equal zero, only the effect of Ξ^P on Ω would not be present. For this reason, Ω was not denoted with a Λ . The FPE equation has to be rewritten in the form that the dairy output P is the dependent variable, since otherwise identification of Ξ^P is not unique (for details see in the appendix (A8)).

Assuming now that all parameters of the FPE and PRE are known or could be estimated allows determining the values of Ξ^P and Ξ^R :

$$\Xi^{P} = \frac{\Lambda^{\Xi^{P}}}{\omega_{0}^{W^{M}} - \frac{\omega_{0}}{\omega_{1}^{F}}}$$
(5.14)

$$\Xi^{R} = -\Lambda^{\Xi^{R}} \frac{\tau_{1}^{P}}{\tau_{0}^{P} - \frac{\tau_{0}^{WM} \tau_{1}^{P}}{\tau_{1}^{WM}}}$$
(5.15)

where with information on η^F and η^P (for the formulas see in the appendix (A10)) the Lerner indices, also Buyer Power Index (BPI) (Blair & Harrison, 1993), called BPI^P and BPI^R could be calculated as well,

$$BPI^{P} = \frac{VMP^{M} - W^{M}}{W^{M}} = \frac{\Xi^{P}}{\eta^{F}} \text{ and } BPI^{R} = \frac{VMP^{Q} - P}{P} = \frac{\Xi^{R}}{\eta^{P}}$$
(5.16)

which provide the percentage mark-down due to oligopsony conduct and thus the effect of market power on the prices. Even though the pricing equations al-

²
$$W^{M^*} = W^M - \varphi^{PX}P - \sum_{j=1}^k \varphi_j^{XN} W_j^P - \varphi^{XT}T - \varphi^X$$

low determining the percentage mark-down without the requirement for any kind of quantity data, it is not possible to derive the welfare loss and the rents at any level without information on quantity data.

5.3 SUMMARY

In this section with the assumptions of homogenous material input and output products, exclusion of exports and storage, no adjustment costs in any variables, and flexible functional forms a theoretical model of oligopsony power on subsequent markets of a supply chain without the requirement of any kind of quantity data is developed.

First, the two markets for the material input and output are modeled as a structural model familiar from the classic static NEIO framework. The system of derived supply and demand equations is then used to obtain explicit solutions of the endogenous variables, the material input and output prices as well as quantities. After rearrangement and substitution, two pricing equations, one depicting the farm-processor relation, and the other depicting the processor-retailer relation, are found which only depend on material input and output prices as well as the non-material input prices, technological change, and the measures for oligopsony power. Consequently, quantity data is obsolete. In the next step, it is shown that if the parameters of these two pricing equations would be known the level of oligopsony power and its effect on prices, in form of Lerner indices, could be determined.

However, estimation of the parameters of the pricing equations is not possible with standard techniques usually applied in NEIO or APT studies, such as iterative least squares or VECM, since they cannot be separately identified. In the following chapter, a methodology is presented that permits to uniquely identify the parameters of the pricing equations from each other without prior knowledge.

6 METHODOLOGY

In this chapter the methodology, which allows determining the market power measures of the theoretical model developed in the previous chapter, is presented. As in Lloyd et al. (2009) we embed the derived pricing equations (5.12) and (5.13) in the long-run cointegration relationships of a VECM. Instead of estimating a standard VECM with globally or locally constant parameters, as in the case of threshold or MS-VECM, we use a state-space representation of a transformed VECM to allow for time-varying parameters. The source of the time-variation is assumed to be the changes in the market power indices. Consequently, we allow $\Xi^{P}{}_{t}$ and $\Xi^{R}{}_{t}$ to vary over time, which is denoted in the rest of the paper with the subscript t. Since $\Xi^{P}{}_{t}$ and $\Xi^{R}{}_{t}$ are components of the compound parameters of (5.12) and (5.13) their time variation causes these parameters to be non-constant as well, and will be denoted as β_{t} in the following section.

Gollop and Roberts (1979) argued that conjectural variations, one interpretation of the derived market power indices \mathcal{Z}_{t}^{P} and \mathcal{Z}_{t}^{R} , are not fixed constants, but rather vary with time. The reason is that unlike the parameters of technology, the conjectural variations are directly formed and influenced by the firms' behavior through procedure of adaptive expectations. By allowing the market power indices, i.e. the conduct parameters, to vary over time, the critic of NEIO with its frequent static estimation of a dynamic concept is refuted.

As can be seen in (5.12) and (5.13) the time-varying β_t are further split in constant, the ω s and τ s, and non-constant components, the Λ_t s. This is achieved through the application of dynamic factor analysis (DFA), whose results permits to determine Ξ^P_t and Ξ^R_t with (5.14) and (5.15). The following sections will in detail present the above described methodology.

6.1 VECM

The non-stationarity of price time series, in particular of prices of vertically or horizontally integrated markets, is frequently present due to a common trend. Ignoring the possibility of non-stationarity might result in auto-correlation and biased estimates. As described in the APT theory review section, error correction techniques have been established to avoid spurious regression results (Hendry & Doornik, 2001; Juselius, 2006). Engle and Granger (1987) with their ECM only accounted for one long-rung equilibrium relation. Johansen (1988; 1991) developed the vector error correction model (VECM), which permits estimating a system of *n* cointegrated time series with n-1 cointegration vectors via maximum likelihood (ML). The VECM representation of Johansen (1988; 1991) is the following,

$$\Delta Y_t = \alpha \beta' Y_{t-1} + \sum_{i=1}^{p-1} \Gamma_i \Delta Y_{t-i} + u_t \tag{6.1}$$

where Y_t is the data vector of the *n* variables, Δ is the difference operator, Γ_i $(n \times n)$ is a parameter matrices quantifying the short-run response of ΔY_t to past shocks, *t* is a subscript indicating the time dimension, and *p* is the number of lags. Furthermore, β is the cointegration matrix $(r \times n)$ that quantifies the long-run equilibrium relationship with *r*, the so-called 'rank', being the number of cointegration relationships among the n time series. The α matrix $(n \times r)$ is the so-called loading matrix, which measures the speed by which the system moves back to the equilibrium after deviations, given by $\beta' Y_{t-1} = ECT_{t-1}$, with regard to *t*. The error term u_t is a $(n \times n)$ matrix of normal and identical distributed disturbances with zero mean and non-diagonal covariance matrix (Steen & Salvanes, 1999; Lloyd et al., 2009).

To simplify the estimation of a VECM Lütkepohl and Krätzig (2004) developed the simple two step (S2S) estimation procedure. By applying the Frisch-Waugh-Lovell Theorem (Lovell, 2008) in their procedure, (6.1.) can be reduced to,

$$M\Delta Y_t = \alpha \beta' M Y_{t-1} + u_t \tag{6.2}$$

where $M = I - \Delta Y_{t-1} (\Delta Y_{t-1}^T \Delta Y_{t-1})^{-1} \Delta Y_{t-1}^T$ transforms the variables Y_t and Y_{t-1} so that they incorporate the partial effect of the short-run dynamics. In the next step of the S2S, (6.2.) is estimated via simple OLS, and a parameter matrix $\gamma = \alpha \beta'$ $(n \times n)$ is obtained. By normalizing the cointegration vector β , setting the value of the corresponding β s of r variables to one and zero otherwise, permits identification of the α matrix. Furthermore, it is important to note that the choice of normalization does not affect the estimates of a cointegration relationship (Juselius, 2006). With knowledge of the loading matrix α , (6.2.) (for explicit form see in the appendix (A7)) is rearranged to yield,

$$MW_t = \beta' M Y_{t-1}^2 + v_t \tag{6.3}$$

where $W_t = (\alpha^T \sum_u a^{-1} \alpha)^{-1} \alpha^T \sum_u a^{-1} (\Delta Y_t - \alpha Y_{t-1}^1)$ is a r scaled vector, \sum_u being the residual covariance matrix of the error term u_t and $Y_{t-1} = (Y_{t-1}^1, Y_{t-1}^2)$ is split into Y_{t-1}^1 , the r time series whose cointegration parameters were normalized, and Y_{t-1}^2 , the remaining data vector (Lütkepohl & Krätzig, 2004). Even though, the S2S approach uses OLS for estimation, its estimator has the asymptotic distribution of a maximum likelihood estimator (Ahn & Reinsel 1990; Reinsel, 1993).

6.2 STATE-SPACE REPRESENTATION OF THE TRANSFORMED VECM

To permit the cointegration parameters β_t to vary with time, instead of using standard OLS techniques, as suggested by Lütkepohl and Krätzig (2004), the Kalman-Filter (Kalman, 1960) is applied to the transformed VECM (6.3). This allows accounting for the firms' behavior of adaptive expectations and updating Ξ^P_t and Ξ^R_t , in each single time period. The state-space representation of (6.3) takes the form,

$$MW_t = \beta'_t M Y_{t-1}^2 + v_t \tag{6.4}$$

$$\beta_{t+1}' = \beta_t' + \varsigma_t' \tag{6.5}$$

where (6.4) represents the observation equation and (6.5) the state equation. The error term ς_t ($r \times 1$) is a vector of normally and identically distributed disturbances assumed to be Gaussian. The Kalman-Filter is a recursive procedure in which the estimates of the unknown state, here the parameters β_t as well as the corresponding covariance matrix Σ_t , are estimated using last observed values. In each time period, with new observations on the observable data, here W_t , floating in, these estimates are updated (Kalman, 1960). Even if the assumption of Gaussian error terms fails to be true, the ML estimates of β_t are still the best linear unbiased estimates (Wildi, 2013).

The Kalman-Filter requires to be initialized by a set of chosen starting values at time t_0 . In this case, we use the OLS estimates, as initially suggested by Lütkepohl and Krätzig (2004) for their S2S method, for β and Σ to initiate the Kalman-Filter. After obtaining the optimized ML values for the initial parameters βt_0 and Σt_0 , the procedure is repeated (Wildi, 2013).

6.3 DYNAMIC FACTOR ANALYSIS

The estimates of the time-varying cointegration parameters β_t can be further separated into a constant and a non-constant component, a so-called common factor. To accomplish this, the time series techniques dynamic factor analysis (DFA) is applied (Stock & Watson, 2005). The general idea is that n univariate time series form a multivariate system of variables, here after referred to as 'response variables'. The system responds to changes in m common factors, where $1 \le m < n$, instead of trends unique to each individual univariate time series (Zuur et al., 2003). The DFA with one common dynamic factor, here Λ^r_t , based on either Ξ^P_t and Ξ^R_t , for the response variables, here β_t s, can be formulated by,

$$\beta_t^{r'} = \mu_0^{r'} + \mu_1^{r'} \Lambda_t^r + \varepsilon_t$$
 (6.6)

$$\Lambda_t^r = \kappa^r \Lambda_{t-1}^r + \eta_t^r \tag{6.7}$$

where the μ s represent the constant parameters ω s and τ s from the pricing equations (5.12) and (5.13), and the κ s are weighing the effect of past values of the common trend Λ^r_{t-1} on current values Λ^r_t . The error terms, here ε_t and η_t , are assumed to be Gaussian normally distributed with zero mean (Zuur et al., 2003).

The state-space representation consisting of (6.6) and (6.7) is estimated with ML. Nevertheless, since the dynamic factors are unknown the log likelihood function cannot be optimized directly. Instead the expectation-maximization (EM) algorithm is applied, an iterative procedure that "successively maximizes the conditional expectation of the complete data likelihood function" (Zuur et al., 2003). Hereby the ML estimates of so-called hyperparameters are obtained. These hyperparameters comprise the variances of the error terms, the parameter μ s and κ s, and the initial values of the dynamic factors and its variances at t_0 . After determining the ML values of the hyperparameters the dynamic factors and their variance is obtained with help of the Kalman-Filter (Shumway & Stoffer, 2000).

Furthermore, some of the parameters have to be restricted to find a unique solution (Harvey, 1989). In this approach the initial variance of each dynamic factor is chosen to be zero at t_0 . However, similar to the estimation of the time-varying parameters in (6.2), an initial ML estimation provides the optimized starting values for the final estimation. The dynamic factor analysis is carried out by using the MARSS package in the software R (Holmes et al., 2012).

As already inclined by incorporating the superscript r into the notation of statespace representation of the dynamic factor model, the time-varying cointegration parameters of the two long-run equilibrium relationships representing the pricing equations (5.12) and (5.13) are pooled (see in the appendix A(11)) and a dynamic factor model for each of the two parameter vectors is estimated separately. The reason is that \mathcal{Z}^{P}_{t} enters the PRE through Ω_{t} and thus must be determined prior to estimating the dynamic factor that extracts the common trend depending solely on \mathcal{Z}^{R}_{t} .

6.4 SUMMARY

In this chapter the methodology that allows determining the market power measures at the processor and retail level was presented. Figure 8 displays the estimation strategy and all estimation techniques applied to derive the market

power indices. Due to the common cointegration of vertically integrated price time series, the pricing equations of the structural model are embedded as cointegration vectors in a VECM. Nevertheless, the standard VECM would not allow identifying the market power indices. Instead, the original VECM is transformed via the simple two step estimation procedure (S2S) developed by Lütkepohl and Krätzig (2004). Assuming two cointegration vectors, this technique permits to estimate the farm-processor equation (FPE) and the processor-retailer equation (PRE) separately. Since the β_t parameters of the PRE rely on both market power indices, Ξ^P_{t} and Ξ^R_{t} , it is necessary to first derive a measure for Ξ^P_{t} in order to simplify the equation to incorporate only one unknown market power measure.





Source: own illustration.

Due to the assumption of time-variation in the market power indices, the cointegration parameters, the compound parameters of the derived pricing equations, have to change with time if the level of market power changes. Time-varying parameters are derived with the help of the Kalman-Filter, which is applied in step (1) to the FPE.

The obtained time-varying cointegration parameters can now be split in step (2) into constant parameters and a non-constant element via dynamic factor analysis (DFA). This non-constant element is a factor common to all parameters and consists of the corresponding market power index and constant parameters of the derived two market structural model. With the estimates of the DFA for the FPE it is now possible to calculate the market power index of the processing sector, the price elasticity of farm supply, the processors' buyer power index (BPI), and a parameter Ω . The parameter Ω incorporates the effect of the processors' market power on the PRE and thus simplifies the latter to only one unknown market power index instead of two.

In step (3) and (4) the Kalman-Filter and the DFA are applied to the 'simplified' PRE to derive in the same manner the market power index of retailers, the price elasticity of processor supply, and the corresponding retailers' BPI. The following chapter will present descriptive statistics on the data and results of an application of the outlined estimations strategy to the German dairy supply chain.

7 EMPIRICAL APPLICATION

The presented theoretical model and the methodology that allows estimating the market power indices is applied empirically to the German dairy supply chain for the years from 2003 to 2015. At first, the used database is described and the choice of included variables justified. After initial cointegration analysis, the statistically favored VECM specification regarding cointegration, number of differenced lags, and restrictions on the parameter matrices is chosen. The in the methodology chapter explained estimation procedures of the Kalman-Filter and the DFA are then applied to the transformed state-space representation of the selected VECM specification. The results of the estimations of the state-space model and the DFA as well as the calculation of important parameters derived from the theoretical model such as the market power indices, Ξ^P_t and Ξ^R_t , the elasticities of supply, η^F_t and η^P_t , and the corresponding BPI indices are presented thereafter. Finally, the procedure and findings are summarized.

7.1 DESCRIPTIVE STATISTICS

In this section the database used in the empirical application is presented and described, variable selection explained, as well as data sources revealed. In addition, necessary data manipulation procedure is elaborated. The time period for the study spans from January 2003 to December 2015 thus covering major policy changes and events including the gradual and actual abolishment of the milk quota, EU enlargements, and intensive structural change at all levels of the supply chain, that have altered irrevocably the business environment and the structure of the German dairy supply chain.

The database includes all publicly available relevant material and non-material input costs for the production and marketing process along the dairy supply chain (see Table 12). The costs of capital, here in form of the money market rate, and transport, here in form of the diesel price, are assumed to be a significant cost variable at all levels. Unfortunately, no data on actual capital costs were available, but it was approximated with the money market rates of the German federal bank. Raw milk is sourced on average by German dairies in a radius of 170 km. Consequently, transport costs play a major role in the procurement of raw milk. Furthermore dairy products are distributed throughout Germany, which again makes the inclusion for transport costs a necessary requirement for the analysis (Tribl & Salhofer, 2013). Apart from capital and transportation costs, it is assumed that the relevant cost in the production of raw milk is dairy cow feed. The procurement of feed summed up to around 41.1% of the intermediate consumption in German agricultural production in 2014 (BMLE, 2016). The

dairies main matter of expense is raw milk with a cost share of approximately more than 55% (Bundeskartellamt, 2012). In addition, labor and energy costs are treated as significant at the processing stage with cost shares of 6.8% and 2.1% respectively in 2015 (BMLE, 2016). Aside from the procurement price for dairy products, the main matter of expense for the retail level are wages, e.g. accounting for up to 30% of the retail price in the case of U.S. retailers (Hovhannisyan & Gould, 2012).

item	unit	frequency	symbol	min	max	mean	source
raw milk price	€/I	monthly	W^M	0.22	0.41	0.31	BMVEL/BMELV/BMEL
implied dairies' output price ^a	€/I	monthly	Р	0.44	0.61	0.48	BMVEL/BMELV/BMEL
skim milk powder ^c	€/kg	monthly		1.39	3.63	2.13	BMVEL/BMELV/BMEL
German brand butter (formed) ^c	€/kg	monthly		2.17	4.44	3.24	BMVEL/BMELV/BMEL
Emmentaler ^c	€/kg	monthly		3.93	5.50	4.32	BMVEL/BMELV/BMEL
milk performance feed	€/kg	monthly	W^F	0.14	0.28	0.19	AMI/ZMP
avg. wage dairy industry ^b	€/h	monthly	W^{P}_{1}	15.25	28.11	21.77	BMVEL/BMELV/BMEL
energy price ^d	€/kWh	monthly	W^{p}_{2}	0.07	0.12	0.09	BMWE/Statistisches Bundesamt
avg. wage retail em- ployee ^e	€/h	quarterly	W^R	9.87	13.13	11.53	Statistisches Bundes- amt
money market rate ^f	%	monthly	W^{F+P+R} 1	-0.20	4.30	1.49	Deutsche Bundesbank
diesel price ^d	€/I	monthly	W^{F+P+R}_{2}	0.95	1.52	1.22	BMWE/Statistisches Bundesamt

Table 12:	Descrip	otive	statistics	of	used	dataset.

Source: own elaboration.

Notes: ^a calculated using technical conversion factors and shares on processing; ^b seasonally adjusted using x12arima; ^cafter March 2012 continued with index data, since price data has not been published; ^d calculated using index series from Statistisches Bundesamt and avg. price of 2010 for energy procured by industry (0.0971 €/kWh) and diesel (1.23 €/l) provided by BMWE; ^e interpolated to monthly frequency using Eviews Software; ^fEONIA.

Except for one data series, the frequency of the data was monthly. For the average wage of retail employees only data with quarterly frequency was available. Therefore, this time series was interpolated to a monthly frequency (see Table 12). All price time series were deflated via the Consumer Price Index (CPI) of the German federal statistic service. While data on the raw milk price is publicly available, the assumption of a homogenous dairy output product implies the construction of a corresponding price time series. As mentioned before this dairy output price P is a share weighted price of major dairy products. It consists of the wholesale prices for Emmentaler, SMP, and German brand butter (formed). The exact formula for calculating the implied dairies' output price is given by,

$$P = \left(\frac{butter \ price}{ME \ butter} + \frac{SMP \ price}{ME \ SMP}\right) \frac{total \ butter \ production * ME \ butter}{total \ milk \ processed} + \frac{Emmentaler \ price}{ME \ Emmentaler} \left(1 - \frac{total \ butter \ production * ME \ butter}{total \ milk \ processed}\right)$$
(7.1)

where ME stands for milk equivalent and thus for the conversion rate of raw milk amount required (measured in liter) to produce one kg of the dairy product. The conversion rates for butter, SMP, and Emmentaler are 25 liter/kg, 10 liter/kg, and 7.5 liter/kg respectively. Figure 9 illustrates the price time series for the selected dairy products as well as the derived share weighted dairy output price P.

Figure 9: Price for German brand butter (formed), SMP, and Emmentaler in €/100g, and the calculated dairy output price in €/I.



Source: own illustration, based on data from BMVEL/BMELV/BMEL, 2003-2016.

The development of the deflated share weighted dairy output price and the raw milk price are depicted in Figure 10. The raw milk price demonstrates a seasonal

pattern up to the year 2007, which is not shared by the share weighted dairy output price. While both price time series evolved slightly downward trending but stable from 2003 to mid-2007, both series drastically rose afterwards. After peaking in the beginning of 2008, raw milk price at 0.42 € and dairy output price at 0.65 € in December and November of 2007 respectively, both prices lost dramatically in value. Interestingly, the raw milk price continued its downward slope, reaching its minimum value over the analyzed period in September 2009 with slightly less than 0.22 €, while the dairy output price stabilized several months earlier. Consequently, the price margin increased during this period to a maximum of almost 0.28 € in October 2009. After 2010 the raw milk price recovered guickly and even reduced the price margin to less than it was before the price peaks, roughly between 0.15 € and 0.20 €, about 0.10 € to 0.11 € until the end of 2010. During almost the entire rest of the analyzed periods both time series evolved parallel, keeping the difference stable. However, since the end of 2014 both prices entered a phase of decline with the raw milk price losing more value, thus increasing the price margin slightly again to around 0.16 €.



Figure 10: Deflated prices for raw milk and dairy output as well as their margin in €/l.

Source: own illustration, based on data from BMVEL/BMEL/BMEL, 2003-2016.

7.2 INITIAL COINTEGRATION ANALYSIS

At first we test whether the selected time series are non-stationary, i.e. have a unit root, by applying the Augmented Dickey Fuller Test (ADF) to each data series individually (Dickey & Fuller, 1981). The results of the ADF test for each

symbol	selection criterion	lags	t-statistic
Wм	HQC/SC	2	-0.5893
	AIC/FP	12	-0.5758
Р	HQC/SC	1	-0.7668
	AIC/FP	2	-0.6090
WF	AIC/FP/HQC/SC	1	-0.0758
$W^{P}{}_{1}$	SC	2	0.8474
	HQC	5	1.2663
	AIC/FP	8	1.7479
W^{p}_{2}	HQC/SC	0	2.5321
	AIC/FP	12	1.5801
W ^R	SC	3	-0.0889
	HQC	4	-0.1083
	AIC/FP	5	-0.0933
WF+P+R ₁	SC	1	-1.3897
	HQC	2	-1.5041
	AIC/FP	6	-1.2216
W^{F+P+R} 2	HQC/SC	0	-0.2529
	AIC/FP	2	-0.4312

Table 13: Augmented Dickey Fuller unit root test results.

Source: own elaboration, calculated using the JMulti software.

Notes: Akaike Info Criterion (AIC), Final Prediction Error (FP), Hanna-Quinn Criterion (HQC), and Schwarz Criterion (SC). Critical values at the 1%/5%/10%-level: -2.56/-1.94/-1.62.

data series and the choice of lag length suggested by the four selection criterions, Akaike Info Criterion (AIC), Final Prediction Error (FP), Hanna-Quinn Criterion (HQC), and Schwarz Criterion (SC), is provided in Table 13. The presence of a unit root is not rejected in any series or any of their specifications. Consequently, all data series are non-stationary.

Because all series are non-stationary, the next step is to test whether a combination of these exists that is stationary in the long-run and the time series accordingly cointegrated. The Saikkonen & Lütkepohl Test (Lütkepohl et al., 2004) test for cointegration is applied to the system of eight data time series including a time trend and a constant in the possible long-run relationships. The maximum rank, number of cointegration vectors, in this system is seven. Table 14 presents

selection criterion	no. of	$H_0: r = r_0$	LR	с	ritical value	S
	lagged differences			1%	5%	10%
Schwarz Criterion	1	r ₀ = 0	215.69	163.38	152.59	147.04
		r ₀ = 1	127.63	129.39	119.77	114.84
		r ₀ = 2	80.12	99.40	90.95	86.64
		r ₀ = 3	53.15	73.42	66.13	62.45
		r ₀ = 4	19.51	51.45	45.32	42.25
		r ₀ = 5	5.66	33.50	28.52	26.07
		r ₀ = 6	2.70	19.71	15.76	13.88
		r ₀ = 7	1.31	9.73	6.79	5.47
Final Prediction Error/Hanna-	2	r ₀ = 0	192.25	163.38	152.59	147.04
Quinn Criterion						
		r ₀ = 1	137.98	129.39	119.77	114.84
		r ₀ = 2	76.20	99.40	90.95	86.64
		r ₀ = 3	48.15	73/42	66.13	62.45
		r ₀ = 4	22.34	51.45	45.32	42.25
		r ₀ = 5	11.47	33.50	28.52	26.07
		r ₀ = 6	4.11	19.71	15.76	13.88
		r ₀ = 7	2.08	9.73	6.79	5.47
Akaike Info Criterion	12	r ₀ = 0	390.19	163.38	152.59	147.04
		r ₀ = 1	281.33	129.39	119.77	114.84
		r ₀ = 2	187.66	99.40	90.95	86.64
		r ₀ = 3	118.68	73/42	66.13	62.45
		r ₀ = 4	74.64	51.45	45.32	42.25
		r ₀ = 5	40.66	33.50	28.52	26.07
		r ₀ = 6	10.23	19.71	15.76	13.88
		r ₀ = 7	2.23	9.73	6.79	5.47

Table 14: Saikkonen & Lütkepohl cointegration test results for system with seven dimensions including a deterministic constant and trend term.

Source: own elaboration, calculated using the JMulti software.

the results of the cointegration test for all three specifications regarding the included number of lagged differences suggested by AIC, FPE, HQC, and SC. For all three specifications, at least two cointegration vectors can be diagnosed at the 10% significance level. With a lag length of 12, suggested by the AIC, even more than five long-run relationships can be identified. Consequently, the previously described methodology, based on two cointegration vectors, is confirmed by the cointegration test results.

However, since we allow the parameters of the cointegration variables to vary with time, we indirectly model 'structural breaks' in the cointegration relationship. Figure 11 displays the p-values of the Chow forecast test results for the VECM with two cointegration vectors and a differing number of included lags. The Chow forecast compares the residual variance of the full sample with those of the first subsample. If these differ, the null hypothesis of constant residual covariance matrix and thus constant parameters has to be rejected (Lütkepohl & Krätzig, 2004). The test confirms our assumption that the parameters are not constant. Until around mid-2009 the hypothesis of constancy is rejected at the 5% level. However, the specification including 12 lags never rejects, except for one month at the end of the analyzed time period, the null hypothesis, which might be caused by the relative small sub period, which only ranges from November 2010 to November 2015, during which all other specifications confirm the stability of the system.





Source: own elaboration, calculated using the JMulti software.

The VECM, in general, regards all variables of the data vector as endogenous. However, as assumed in the model outlaid in the theory section, only the prices and quantities of the material input and the output are assumed to be endogenous. All other prices are treated as being exogenous. Setting linear restrictions on the loading matrix α as well as the parameter matrices that quantifies the short-run response of ΔY_t to shocks in the past, Γ_i , allows to treat these variables as exogenous to the system. Restrictions are set by setting the corresponding parameters of these matrices to zero. When solely α is restricted, the variables, which are affected by the restriction, are treated as "weakly exogenous for the cointegrating parameters if none of the cointegration relations enter the equation for that variable" (Lütkepohl & Krätzig, 2004). Further restricting the Γ_i as well limits the affected variables to be truly exogenous, not reacting to past developments of the endogenous variables or shocks to the system, and their presence in the VECM to the cointegration relationship. Consequently, these exogenous variables enter the VECM in a similar way as the deterministic components, the trend and the constant term.

Applying these restrictions to the model developed in the theory section, the α matrix will collapse to a (2×2) matrix in the case of weak and complete exogeneity assumption for all variables except for the raw milk and dairy output price. In case of weakly exogenous variables the $p \Gamma$ matrices remains unrestricted, but with full exogeneity take the same dimension as the restricted α matrix. All three model specifications with either endogenous, weakly exogenous and fully exogenous variables, will be estimated and tested.

The statistical analysis conducted and described in these previous paragraphs leads to a variety of model specifications. The four criteria suggest three different lag lengths for the cointegration analysis. Even though, the explanatory power of any of the criteria could be challenged, e.g. the AIC in many cases overestimates the lag order (Lütkepohl & Krätzig, 2004), all three different cases are treated equally. Consequently, one, two, and 12 differenced lags of the endogenous variables are incorporated in different model specifications. The Saikkonen & Lütkepohl Test for cointegration revealed that in all cases, up to two cointegration vectors are accepted. Even though one specification with 12 lags would also accept up to five cointegration relationships, the forthcoming empirical application is limited to two long-run equilibrium relationships as it is coherent with theory. The two cointegration vectors are normalized respectively on one of the non-material input prices at the farmers and retailers level. With the three different specifications regarding non-material input price variables to be either endogenous, weakly exogenous, or exogenous, in total nine different model specifications were estimated.

7.3 RESULTS

7.3.1 Model selection

The previous section elaborated that nine different models due to different choices of lag length and three different assumptions on the nature of nonmaterial variables were specified. All nine models were transformed as in (6.3.) and the state-space representations of the FPE were estimated. After obtaining the Kalman-Filter estimates and the corresponding standard errors, the p-values of the time-varying parameter were calculated for each *t*. Graphical depiction of the p-values of the time-variant parameters of all model specification can be found in the appendix (for detail see Figure (A1)-(A9)). To be able to derive $\Xi^{P}{}_{t}$ the β_{t} parameter of W^{M} and W^{F} need to be statistically significant. Table 15 lists all model specification and whether the necessary parameters were statistically significant at the 10% significance level and thus a $\Xi^{P}{}_{t}$ could be deduced. If the required parameters did not demonstrate any statistical significance and no market power measure could be derived, then these model specifications were dropped from further analysis and consideration.

If the estimates of a specified model yielded theoretically consistent values in the range from zero to one for \mathcal{Z}^{P}_{t} , also the state-space representation of the PRE equation could be estimated with the obtained knowledge on certain parameter values, in particular Ω , derived from the estimates of the FPE. The time-varying parameters of P and either of the transformed variable W^{M^*} or one of the non-material input variables at the processing level are required to be statistically significant at some point in time t to proceed with the DFA and derive a statistically significant measure for \mathcal{Z}^{R}_{t} . Was this not the case, then the model specification was not estimated via the DFA. An overview of which model specification allowed to derive a theoretically consistent market power index at the retail level is presented again in Table 15.

From these nine specifications the model that best describes the data was chosen for the further analysis. Apart from the pre-requirements of certain parameter significance and the theoretically consistent values for Ξ^P_t and Ξ^R_t , the Δ AlCc and, based on the Δ AlCc, the weighted AlCc were used to identify the best fitting specification. The Δ AlCc is the difference between the AlCc (Akaike Information Criterion with correction) value of the model and the model with the lowest AlCc score. The AlCc is calculated in the following way (Hruvich & Tasi, 1989; 1995),

$$AICc = \underbrace{-2\log(\mathcal{L}) + 2K}_{AIC} + \underbrace{\frac{2K(K+1)}{n-K-1}}_{correction \ term}$$
(7.2)
where L is the maximum likelihood value, n is the sample size, and K is the number of estimated parameters. The correction term sanctions overfitting models and the tendency of AIC to favor these (Claeskens & Hjort, 2008). The AICc values convey no direct interpretation of the suitability of a model but have to be compared to other specifications. The model with the lowest AICc value is the best fit. Consequently, the difference between the lowest AICc value and the AICc of a model specification allows ordering models according to their suitability. Furthermore the weighted AICc permits to evaluate the probability that one model specification is the best fit for the data in comparison to the other model specifications (Burnham & Anderson, 2004).

Table 15:	List of model specifications differing in lag length and the nature of the
	non-material input variables with their corresponding FPE results for AICc
	and the weighted AICc values.

no. of differenced lags	non-material input variable	$\begin{array}{c} {\rm required} \\ \beta_{{\rm 1}t}{\rm S} \\ {\rm significant} \end{array}$	$0 \leq {\Xi^p}_t \leq 1$	$\begin{array}{c} {\rm required} \\ \beta_{2t}s \\ {\rm significant} \end{array}$	$0 \leq \Xi^{R}_{t} \leq 1$	ΔAICc	weighted AICc
1	endogenous	yes	no	-	-	-	-
2	endogenous	yes	no	-	-	-	-
12	endogenous	yes	yes	yes	yes	2347.3	0%
1	weakly exog- enous	no	-	-	-	-	-
2	weakly exog- enous	yes	yes	yes	yes	0	100%
12	weakly exog- enous	yes	no	-	-	-	-
1	exogenous	no	-	-	-	-	-
2	exogenous	yes	no	-	-	-	-
12	exogenous	no	-	-	-	-	-

Source: own elaboration, AICc values were obtained from the estimations within the MARSS package in R.

The results of the Δ AICc and the weighted AICc for the FPE are listed in Table 15 as well. The DFA results of the PRE are not listed, since, as can be seen in the estimation strategy (see Figure 8), the estimates of the FPE affect the PRE estimation and consequently the models do not share the same data input and are not comparable anymore. Six out of the original nine model specifications revealed the necessary time-varying parameters to be statistically significant at

one point in time t at the 10%-significance level. Only two of these specifications allowed deriving theoretically consistent values for $\Xi^P{}_t$. Furthermore, the further estimation of the PRE for these two specifications revealed that $\Xi^R{}_t$ lied in the range of zero to one for both model specifications. Based on these results and the AICc values, the best fit for the data is a model specification with two lags of the differenced endogenous variables and weakly exogenous non-material input variables. Furthermore, this specification, according to the weighted AICc measure, compared to the other specifications explains the data with virtual certainty. Due to this high probability and being one of the two specifications that yield theoretically consistent values for $\Xi^P{}_t$ and $\Xi^R{}_t$, solely the results of this model specification will be presented and discussed in the following subsections, since it explains the data best.

7.3.2 State-space representation

After S2S-VECM transformation the two long-run equilibrium vectors can be separately estimated. By fitting each one individually into a state-space representation and applying the Kalman-Filter, the corresponding β parameters vary over time. The results of the application to the rearranged FPE (see in the appendix (A9)) are depicted in Figure 12 and to the PRE in Figure 13. The corresponding calculated p-values for each point in time of the estimated parameters can be found in the appendix (Figure (A5) and Figure (A11) respectively).

The estimates of the FPE demonstrate that the cointegration parameters indeed change over time (see Figure 12). In general, most parameters evolve stable from the start until the end of the year 2007. Then either a sudden decline or increase in value depending on the parameter sets in. The exceptions are the time-varying parameter for the diesel and feed price. Around the end of 2009 another sudden upwards or downwards bump leads to greater changes in the parameter values, except for the money market rate and trend, but is accompanied by a quick recovery in the opposite direction. The time period afterwards can be generally characterized as period of rather stable parameter evolution.

According to the p-values all but one parameter are statistically significant at least at the 5%-level over the entire period (see in the appendix Figure (A5)). Only the time-varying parameter of the feed price drops for brief periods out of the 10%-significance range. However, since the parameter is statistically significant at the 10%-significance level for most of the time, it is included in the further analysis in form of the DFA.





Source: own elaboration.





Source: own elaboration.

In contrast to the estimates of the state-space representation of the FPE, the results of the PRE show greater absolute movement in the parameter values (see Figure 13), in particular at the start of the period. While until around the beginning of the year 2010 no stable period for any parameter estimate can be observed, some of the parameters, e.g. the time-varying parameter for the constant or the average wage in the dairy industry, start to

stabilize and evolve rather stable until the end of the investigated time horizon. After the start of the year 2011 all parameters evolve rather stable, but most of them on a different level as at the start, e.g. the parameter of the dairy output price.

The p-value graphs for the time-varying parameters of the PRE state-space representation (see Figure (A11)) reveal that all but one parameter are statistically significant at the 10%-significance level over the entire time series. The exception is the parameter of the trend variable, which drops out of this level of statistical significance around the start of the year 2010. Nevertheless, since all estimated time-varying parameters are statistically significant at the 10%-level most of the time, they are treated as dependent variables in the DFA of the PRE.

7.3.3 Dynamic factor analysis





Source: own elaboration.

The results of the Kalman-Filter enter the DFA as dependent variables, whose purpose it is to extract a common factor that explains their variation over time from these. The common factor is a function of the market power indices and parameters of the derived supply and demand functions, and thus allows determining the level of market power abuse on each of the two analyzed markets as described in the theory section.

Figure 15: Common factor estimates for the processor-retailer cointegration parameters including dotted lines for the corresponding 95%-confidence interval.



Source: own elaboration.

Figure 14 and 15 depict the estimated common factor for the FPE and PRE respectively. Both factors are statistically significant at the 1%-significance level (see in the appendix (A12) and (A13)) and show drastic relative values changes over the entire analyzed time period from 2003 to 2015. The dynamic factor depending on the market power of the processors ranges from about -0.06 to -0.05. Around the end of the years 2003, 2006, 2010, and 2012 sudden drops in value can be observed. The sudden decline in value at the end of the year of 2006 is reversed by an even stronger increase in value at the beginning of 2007. Afterwards a continuous decline until around the aforementioned drop at the end of the year 2010 sets in. From 2012 a general positive trend, fragmented by brief periods of quick decline and recovery, evolves and the dynamic factor ends on a level comparable to the pre-2010 drop.

The common factor subject to the retailers' market power varies between a minimum of -40.50 and maximum of -26.67 drastically. The first six years of the

analyzed time horizon Λ^{PRE} ranges on a level between approximately -33 to -40. At the end of the year 2009 the dynamic factor raises drastically in value in a few months to reach a local maximum of around -29. Following a slowed growth the absolute maximum is reached in end-2013. The maximum is trailed by a steady decrease in the factor's value. Thus the dynamic factor of the PRE finishes at -33.29 at the end of the analyzed time period. Even though, the values of Λ s do not convey any meaning, their evolution represents changes in the level of market power for processors and retailers respectively.

Alongside the dynamic factor also the constant components of which the timevarying parameters consist of are estimated (see Table 16 and 17). Because some of them are necessary for calculating Ξ^{P}_{t} and Ξ^{R}_{t} and almost all are required to obtain the price elasticities of supply, their statistical significance plays a crucial role for the reliability of the desired measure. If any parameter estimate did not bear a statistical significance at least at the 10%-significance level it did not enter the calculation of any aforementioned measure. This procedure combined with

variable	symbol	corresponding	value	standard
		parameter		error
raw milk price	W^M	$\omega_0^{W^M}$	-1.1694***	0.0230
milk performance feed	W^F	$\omega_0^{W^F}$	0.9326***	0.1779
		$\omega_1^{W^F}$	8.2354***	1.2335
money market rate	W_1^{F+P+R}	$\omega_0^{W_1^{F+P+R}}$	-0.7486	0.5045
		$\omega_1^{W_1^{F+P+R}}$	23.5763***	3.1322
diesel price	W_2^{F+P+R}	$\omega_0^{W_2^{F+P+R}}$	-0.0271	0.0600
		$\omega_1^{W_2^{F+P+R}}$	-2.8855***	0.6857
trend	Т	ω_0^T	-0.0014	0.0014
		ω_1^T	0.0050	0.0297
constant	С	ω_0^C	-0.5403***	0.0811
		ω_1^c	3.4974***	0.7199
state factor loading parameter	κ	κ^{FPE}	1.0000***	0.0014

Table 16: DFA results for the constant parameters ωs of the FPE with corresponding standard errors and significance level.

Source: own elaboration. Notes: ***Significance at the 1 per cent level **Significance at the 5 per cent level *Significance at the 10 per cent level.

the high statistical significance of the two dynamic factors permits to conclude that any measure calculated on basis of these parameters and the common factors will be statistically significant at least at the 10%-significance level as well.

variable	symbol	corresponding	value	standard
		parameter		error
(adjusted) raw milk price	W^{M^*}	${ au}_0^{W^{M^*}}$	-0.1961***	0.0069
		$ au_1^{W_1^P}$	0.0041***	0.0006
dairies' output price	Р	$ au_0^P$	40.1471***	1.4883
		$ au_1^P$	0.2693***	0.0405
avg. wage dairy industry	W_1^P	$ au_0^{W_1^P}$	1.3810***	0.0219
		$ au_1^{W_1^P}$	0.0041***	0.0006
energy price	W_2^P	$ au_0^{W_2^P}$	-118.0769***	7.5162
		$ au_1^{W_2^P}$	2.6570***	0.0456
money market rate	W_1^{F+P+k}	$ au_0^{W_1^{F+P+R}}$	-31.1944***	3.7575
		$ au_1^{W_1^{F+P+R}}$	0.6187***	0.1169
diesel price	W_2^{F+P+k}	$ au_0^{W_2^{F+P+R}}$	5.6696***	0.3144
		$ au_1^{W_2^{F+P+R}}$	0.0355***	0.0075
trend	Т	$ au_0^T$	-0.0069**	0.0028
		$ au_1^T$	-0.0007***	0.0001
constant	С	$ au_0^C$	-79.7669***	2.3920
		$ au_1^C$	-0.8094***	0.0521
state factor loading parameter	κ	κ^{PRE}	1.0000***	0.0015

Table 17: DFA results for the constant parameters **ts** of the PRE with corresponding standard errors and significance level.

Source: own elaboration. Notes: ***Significance at the 1 per cent level **Significance at the 5 per cent level *Significance at the 10 per cent level.

7.3.4 Market power indices

The results of the DFA allow determining the market power indices Ξ_t^P and Ξ_t^R with (5.14) and (5.15) respectively. As mentioned in the section before, the cal-

culated indices are statistically significant at least at the 10%-level due to the statistical significance of all parameters involved in their calculation.

Figure 16 and 17 display the calculated market power level at the processing and at the retail stage and their evolution over the analyzed time period. Table 15 already revealed that both market power indices lie in the theoretically consistent range from zero, perfect competition, to one, monopsony. Both market power indices show a mirrored development to the dynamic factors which lay at the foundation to their derivation.

The dairies' market power index Ξ_t^P fluctuates from 0.037 to 0.048 between the years 2003 and 2015. Starting with a value of around 0.042, the market power level reaches local maxima of around 0.044 at the end of 2003 and mid-2007. The last local maximum is quickly followed by a sudden drop to the absolute minimum of around 0.037 in the fall of 2007. Until the beginning of 2012 the time series is characterized by a general increasing trend with the maximum value of about 0.048 at the end of this period. The last 36 months are coined with an overall decline in value.

The retailers' market power ranges from 0.14 to 0.22 (see Figure 17). While increasing steadily at the beginning of the analyzed period from around 0.18 to





Source: own elaboration.



Figure 17: Calculated market power index Ξ^{R}_{t} of retailers.

Source: own elaboration.

its absolute maximum of 0.22 in the fall of the year 2009, \mathcal{Z}^{R}_{t} drops drastically in value after the maximum to around 0.15 only a few months later in mid-2010. This drastic change splits the time sample into two distinct time periods. From this date to around mid-2014 the market power index preserves around the value range 0.14 to 0.16. The last one and half years of the analyzed timeframe the index starts to grow in value again and reaches its starting value of 0.18 again.

7.3.5 Elasticities of supply

Even though the indices \mathcal{Z}_{t}^{P} and \mathcal{Z}_{t}^{R} allow to determine the extent of market power, the effect of this market power on the respective price in form of a mark-down can only be evaluated in combination with the corresponding price elasticity of supply. As well as the market power indices, the derived price elasticities of supply η_{t}^{F} and η_{t}^{P} (for details on the calculation see in the appendix (A10)) are statistically significant at least at the 10%-level due to the statistical significance of all parameters involved in their calculation. If a parameter was not significant at least at the 10%-mark it was dropped from the corresponding equation (see Table 16 and 17).

The development of the price elasticity of raw milk supply η^F_t over the analyzed time period can be split into three dissimilar sub-periods (see Figure 18). From the start of the analysis in 2003 to around end-2009 the elasticity fluctuates in

the value range of around 0.09 to 0.16. The second sub-period begins with a sudden shift in the elasticity's value range. The price elasticity of raw milk supply continues to vary, but on a higher level of approximately 0.15 to 0.22 from 2010 to 2014. For the last couple of months of the investigated time horizon $\eta^{F}{}_{t}$ drops again abruptly from about 0.18 to 0.11.

The derived price elasticity of dairy output supply is shown in Figure 19. In contrast to η^{F}_{t} , η^{P}_{t} oscillates around a value of 0.26 for the first four years. At the start of 2007 the elasticity erupts in a brief period to reach its maximum value of 0.40 at the end of the same year. After reaching its maximum η^{P}_{t} quickly declines to 0.25 in mid-2009, only to stabilize itself around the 0.30-mark for the following two years. From 2011 to the end of the time horizon the elasticity of dairy output supply ranges from its minimum value 0.24 to about 0.30 at the turn of the year 2013/2014. As with the price elasticity of farm supply only parameters entered the calculation of η^{P}_{t} that are statistically significant at the 10% level giving the derived elasticities at least the same statistical significance.





Source: own elaboration.

7.3.6 Buyer power indices

After deriving the market power indices \mathcal{Z}_t^P and \mathcal{Z}_t^R and the price elasticities for raw milk and dairy output supply, η_t^F and η_t^P , it is now possible to calculate the



Figure 19: Calculated price elasticity of dairy output supply η^{P}_{t} .

Source: own elaboration.

corresponding buyer power indices (BPI) as in (5.16). These measure the effect of market power on the actual prices. Figure 20 and 21 display the results for the German raw milk and dairy output market respectively.

Even though the market power of German dairies lies close to the perfect competition case with a value between 0.037 and 0.048, the BPI of dairies reaches values up to 0.48 during the analyzed period, which is due to the relatively inelastic supply of raw milk. The German raw milk price was marked-down by around 47.6% in mid-2009. The mark-down ranges from around 0.21 to 0.48 between 2003 and 2015. The evolution is rather erratic with large peaks at the beginning of the analyzed period around summer 2004, mid-2007, mid-2009, and after the fall of 2013. While every peak is followed by a sudden drop in the BPI's value, the period from the beginning of 2010 to the fall of 2013 is characterized by rather low values in the range from 0.21 to 0.30.

Similar to the dairies' BPI the retailers' BPI's value lies far above the value of the corresponding market power index \mathcal{Z}_{t}^{R} . However, since the retailers' market power index evolves rather stable in the aforementioned two distinct time periods, and the price elasticity of dairy output supply also evolves rather stable throughout the period, apart from the explosive period between the years 2007 and 2009, the movement of the retailers' BPI is a combination of both patterns. While the two distinct periods of the market power index can be observed with

high values around 0.67 to almost 0.78 between the start of the analyzed period and mid-2007, and rather lower values of 0.49 to approximately 0.6 between mid-2010 and mid-2014, also the sudden rise and drop of the elasticity of supply influences the BPI's value strongly between mid-2007 and mid-2009, which can be seen in a mirrored movement in the BPI and its maximum value of 0.82 in the year 2009. As with the market power index, the BPI of retailers starts to grow at the end of the timeframe again to finish with a value of about 0.72.



Figure 20: Calculated buyer power index of dairies.

Source: own elaboration.

Figure 21: Calculated buyer power index of retailers.



Source: own elaboration.

7.4 SUMMARY

The previously derived structural model and the methodology with its estimation strategy were applied to the German dairy supply chain over the time horizon from January 2003 to December 2015 using monthly data. The data consisted of price time series comprising the price for raw milk, a dairy output price derived by share-weighting the prices for skim milk powder, Emmentaler and butter, as well as of other cost variables involved in the product transformation process along the chain.

The initial test for stationarity revealed that all data series possess a unit root. Based on this result, a cointegration test was conducted to test for one or more relationships among the variables that are stationary in the long-run. Depending on the suggested length of lagged differences included in the analysis the number of cointegrated vectors differed. However, all specifications suggested up to two cointegration vectors, which are required to perform the described methodology. Furthermore, the Chow forecast test results support the assumption of time-variant parameters.

With the confirmation of two cointegration vectors and time-varying parameters the previously described estimation strategy was applied to the data using various different model specifications concerning lag length and assumption of the nature of non-material input variables to the system. In the following one model specification was selected for further analysis. From the model specifications that resulted in theoretically consistent values for the derived market power indices, the one best describing the data in accordance to the Δ AICc criterion was selected for further analysis. The model specification with a differenced lag length of two and weakly exogenous non-material input variables revealed to be the best fit.

The first step of the estimation strategy, the application of the Kalman-Filter to the transformed VECM, produced the time-varying parameter estimates. For both long-run equilibrium equations, FPE and PRE, all parameters were statistically significant at the 10%-significance level at least at some point in time over the analyzed period. Furthermore, all parameter estimates vary over time. The time-varying cointegration parameter estimates then entered the dynamic factor analysis as dependent variables.

The DFA produced a for both equations statistically significant time-varying common factor. The other for the calculation of the market power indices required constant parameters were statistically significant as well. The estimates of the common factor cannot be interpreted independently, but their developments over time reveal changes in market power conducts. Consequently, the derived market power indices for dairies, ranging from 0.038 to 0.048, and for retailers, oscillating in the interval 0.14 to 0.22, vary over time as well.

The DFA results do not only allow calculating the market power indices of both markets, but also the price elasticities of raw milk and dairy output supply. While the first evolves rather unstable over time, and shifting to a higher level for the last years of the time horizon, the one of dairy output develops rather stable between 0.24 and 0.30, apart from a peak period with a value of 0.40 in 2009.

With knowledge on the market power indices and the price elasticities of supply it was now possible to derive the buyer power indices (BPI), a percentage measure of the actual mark-down, for both markets. Due to the instability of the raw milk supply price elasticity the development of the BPI of dairies is characterized by this development and ranges from 0.21 to 0.48. On the contrary, the BPI of retailers shows patterns of both variables involved in its calculation. While the two distinct periods of rather high and low market power index values between 2003 and 2009 and mid-2010 to the end of 2015 generate periods of rather high and low BPI values of the range 0.67 to almost 0.78 and 0.49 to about 0.60 respectively, also the sudden peak and drop of the supply elasticity of dairy output supply between mid-2007 and mid-2009 is striking and lifts the BPI to its maximum value of 0.82.

8 **DISCUSSION**

The previous section was of pure descriptive nature regarding the data, cointegration analysis and the obtained results of the empirical application. This section will provide interpretation of the estimated and calculated indices and elasticities as well as compare the obtained results with previous studies. The last section will discuss theoretical and methodological shortcomings of this approach resulting from drawn assumptions, choice of data and variable selection.

8.1 MARKET POWER INDICES

Several factors would lead to assume that the oligopsony level on the German raw milk market is high. German dairy farmers face a highly concentrated German dairy industry, in some regions more than 50% of the raw milk produced is sold to one firm (Bundeskartellamt, 2009) and almost no outside options exist. In addition, dairy farmers face the issue of a possible hold up through dairies due to the nature of raw milk production with its high asset specificity and the perishable nature of raw milk. The threat of a hold up and possible loss of output due to spoilage puts dairy farmers in a weak bargaining position with dairies (Grau et al., 2015).

Indeed, the market power index for dairies with its value range from 0.037 to 0.048 reveals market imperfections on the German raw milk market due to oligopsonistic conduct, but the level of oligopsony is rather low and close to perfect competition. Other studies on oligopsony conduct on raw milk markets report similar estimates. For example, Hockmann and Vőnecki (2009) report a market power index of 0.05 for the Hungarian raw milk market, and Scalco and Braga (2013) for the Brazilian raw milk market of 0.01. Perekhozhuk et al. (2013) and Perekhozhuk et al. (2015) find higher level of oligopsony conduct on the Hungarian (0.15) and Ukrainian raw milk market, but still far from monopsony level.

A diverse number of reasons can explain the low level of oligopsony conduct on the German raw milk market. Even though the concentration of procurement reaches levels of more than 50% on regional markets, on a national aggregated level these might be local exceptions. Due to the usage of national aggregated price data the oligopsony conduct on regional German markets cannot be evaluated and no statement given. Furthermore, while six dairies summing up a market share of approximately 50% (Loy et al., 2015), around 70% of the German raw milk is processed through cooperatives. Dairy cooperatives are obliged to process all of the raw milk delivered by their members. Consequently, quantity reduction in procurement as a result of oligopsony power is not a feasible option and not in the interest of cooperatives. With cooperatives being dominant on the German raw milk market, the possible higher levels of market power through investor-owned dairies are counterbalanced (Tribl & Salhofer, 2013)

Even without the consideration of cooperative action, an actual hold up that can ruin dairy farmers is not likely to be enforced by dairies, since the benefits of a steady flow of raw milk to fully utilize capacities and therefore achieve costminimization production are greater (Schroeter & Azzam, 1991). Apart from the goal of cost-minimization, also dairies' investments in highly specific assets lower the incentive to use market power. The gain from higher profits in the shortrun due to the application of market power is offset by lower rates of return on dairies' own investment, since the exertion of market power might force farmers to exit production and shrinks the procurement base and dairies' capability to utilize their capacities fully in the long-run (Crespi et al., 2012).

While the level of oligopsony is rather low throughout the investigated period, drastic relative changes can still be observed (see Figure 16). In particular, in the summer of 2007 the market power index of dairies punctiliously collapsed by about 18%. This might be a result of increased competition for raw milk between dairies due to growing export opportunities and high prices as a result of growth in global demand for dairy products, a production shock in New Zealand that drastically reduced supply on world markets, and low public stocks of dairy products in the EU (Acosta et al., 2014; Bouamra-Mechemache et al., 2008). The growth in the market power index after this period might be a result of mergers and acquisitions boosting concentration at the dairy industry level as well as the growing raw milk supply as a consequence of the gradual abolishment of the quota, in particular since 2010 (see Figure 2). A growing supply base enhances collusive behavior among buyers (Hockmann & Vőnecki, 2009).

On the German dairy output market larger levels of market imperfections compared to the raw milk marked can be observed. The market power index of retailers ranges over the analyzed period from 0.14 to 0.22. Salhofer et al. (2012) find a similar level of market power on the Austrian butter procurement market by retailers with a market power index estimate of around 0.10. Even though, the German retail sector is highly concentrated, as the Austrian is, the level of market power is still far from monopsony or a collusive cartel. Again, the concept of capacity utilization and cost-minimization with the abundance of large storage facilities for dairy products (Loy et al., 2015) might explain the rather low market power index value in the presence of five German retailers accounting for more than 70% of revenues in German food retailing (BEV, 2016). While the market power index value is far from monopsony level, the presence of significant oligopsony conduct on the dairy output market was still proven by the results. The German retail market, in particular for dairy products where consumers are very sensitive to price changes (Loy et al., 2016), is characterized by intensive horizontal price competition. Consequently, albeit five companies controlling German food retailing, it is unlikely for these to extract oligopoly margins due to intense competition for market shares. The exertion of oligopsony power might be an attempt of retailers to increase profits or market shares on highly competitive markets by offering lower prices secured by significantly marked-down procurement prices (Anders, 2008).

As with the market power index of dairies, drastic relative adjustment in the oligopsony conduct of retailers is apparent (see Figure 17). From 2003 to around mid-2009, the parameter's value increases from around 0.18 to 0.22. The continuing concentration process in form of mergers and acquisition at the retail level, the formation of procurement alliances among larger and smaller retailers, and the growing dairy output supply as a consequence of the gradual quota abolishment as well as cooperatives' commitment to process all their members' raw milk, might have enabled retailers to exert more market power on the dairy output market. However, after this period of growth in the market power index, it suddenly drops in one year, between mid-2009 to mid-2010, from its maximum value to its minimum value of 0.14. In the next three to four years the market power index remains on this level. However, in the summer of 2014 the index starts growing again steadily to finish off with its starting value of 0.18. A possible explanation for the sudden drop in value is that even though the number of retailers might have decreased over the time period, possibly a threshold was passed that made the buyers on the dairy output market procure more competitively (Sexton, 2013). The more recent increase of market power could be a result of the growth in supply as a result of the growth in raw milk production, similar to the previously described situation on the raw milk market.

8.2 ELASTICITIES OF SUPPLY

The price elasticity of raw milk supply is rather inelastic. Its value ranges from 0.09 to 0.22 over the analyzed period (see Figure 18). Consequently, a 1%-increase in the price of raw milk would lead to an increase of the raw milk quantity of 0.09% to 0.22% depending on time t. The derived value of η^F_t is coherent with the literature. Bouamra-Mechemache et al. (2008) estimated a supply elasticity of 0.253 for Germany. For the UK Colman et al. (2005) report a value of 0.2 to 0.3. In general, raw milk supply is considered to be inelastic. For exam-

ple, Perekhozhuk et al. (2013) assume a value of 0.1 for η^F in their study of the Hungarian raw milk market.

Raw milk supply is rather inelastic, since an expansion of production requires high investment in specific assets and cannot be temporary balanced with stocks, which limit a quick adjustment to price changes in the short-run. Consequently, in the long-run raw milk supply reacts more elastic (Bouamra-Mechemache et al., 2008). With the gradual abolishment of the quota, the price elasticity of raw milk supply became more elastic, up to a value of 0.22, since the quota formed a barrier of entry, and supply can freely adjust (Graubner et al., 2011; Bouamra-Mechemache et al., 2008).

The price elasticity of dairy output is also inelastic with a value range of 0.24 to 0.40, but slightly more elastic. In the absence of comparable estimates for Germany, Gohin and Guyomard (2000) with their study on the market power of French retailers provide the only available price elasticity for dairy output with 0.33, which lies in the estimate value range. The inelasticity of dairy output supply is due to similar reasons as at the farm level. With the assumption of dairies trying to fully utilize their existing capacities, a quick response to positive price changes cannot be answered by activating free unused capacities, but rather by high investment cost in processing capacities are necessary to expand production. The possibility to store a number of dairy products to some extent, e.g. SMP and butter, might explain the more stable evolution of η_t^P compared to the one of η_t^F , as well as the slightly more elastic response of supply.

8.3 BUYER POWER INDICES

The exertion of market power on agricultural product markets is more dramatic than on other sectors' markets, since the inelastic supply magnifies the market power extend in form of a severe mark-down (Bakucs et al., 2010). The same can be said about the German raw milk and dairy output markets. Even though, the extent of oligopsony on both markets is rather low, in interaction with the corresponding inelastic price elasticity of supply, this leads to considerable mark-downs. The relative mark-down is expressed by the buyer power index (BPI).

On the raw milk market the BPI ranges from 0.21 to 0.48, meaning that the raw milk price was marked-down by oligopsony power in the range of 21% to 48% over the analyzed period (See Figure 20). Unfortunately, no study has calculated a BPI for the raw milk market so far (for more details see Perekhozhuk et al., 2016), but studies on other agricultural products also report relative large relative mark-downs, up to 1.1 for livestock in the USA (Azzam & Pagoulatos,

1990), as a result of low levels of market power but inelastic supply (e.g. Azzam & Pagoulatos, 1990; O'Donnell et al., 2007; etc.).

Interestingly, the BPI of dairies follows a similar development as the margin between raw milk price and dairy output price (see Figure 10). In periods of larger margins, the BPI is high, in particular observable in mid-2009, and low in periods of reduced margins, see the period from 2011 to 2014. Figure 22 displays the absolute mark-down in comparison to the observed raw milk price, as well as the difference between both. The absolute mark-down oscillates around $0.10 \in$ per liter raw milk, with local maxima of approximately $0.12 \in$ in 2004 and 2008, and a minimum of about $0.08 \in$ between 2010 and 2014 due to the growing price elasticity of raw milk supply. Even though, the months around the start of 2008 are characterized by lower values of the BPI, the high prices in this period lead to highest absolute mark-downs.

Figure 22: Observed raw milk price (continuous line), observed raw milk price plus absolute mark-down (taunted line), and their difference (dotted line) for the analyzed time period.



Source: own elaboration.

With a larger extent of oligopsony power on the dairy output market in combination with a similar inelastic supply, the BPI of retailers achieves higher values in the range from 0.49 to 0.82 over the analyzed period. While the literature provides at least BPIs for other agricultural markets, the only study that determined

Figure 23: Observed dairy output price (continuous line), observed dairy output price plus absolute mark-down (taunted line), and their difference (dotted line) for the analyzed time period.



Source: own elaboration.

a BPI for the processor output market is Gohin and Guyomard (2000). These authors report a BPI of about 0.20 for dairy products, 0.17 for meat products and 0.12 for other food products. The two distinct time periods in the evolution of the retailers' market power index can also be observed in the BPI as well as the margin between observed prices and prices plus mark-down (see Figure 23). The drop in oligopsony power after 2010 drastically lowers the BPI of retailers to a value of around 0.49 to 0.60. While the pre-2010 margin oscillates between 0.30 \in and 0.40 \in , the margin reduces to below 0.30 \in after 2010.

Overall, while oligopsony levels closer to perfect competition than monopsony were observed, the market imperfections in cooperation with the inelastic supply elasticity lead to drastically marked-down prices for raw milk and dairy output. As a consequence considerable amounts of rents were shifted downstream along the German dairy supply chain. However, if these rents were passed on to consumers in form of low dairy products prices, which some studies (i.e. Loy et al., 2016) state is the daily practice of German retailers, in particular of discounters, institutions like the European Commission do not assess this as anticompetitive. In contrast, the German anti-trust agency holds the opinion that even if consumers benefit in form of low retail prices, it does not justify the use of market power (Bundeskartellamt, 2009). Because in this analysis the consumer market is not

included, we have to agree with the view of the German anti-trust agency and characterize the German supply chain as anticompetitive. However, the abolishment of the milk quota boosting the elasticity of raw milk supply (Graubner et al., 2011) as well as the food retail market probably reaching a threshold level of concentration have benefited the competitiveness of the supply chain and lowered the relative as well as the absolute mark-downs at each market.

8.4 ACTIONS AGAINST PRICE MARK-DOWNS

The two markets of the German dairy supply chain are characterized by different levels of oligopsony behavior. While dairy cooperatives seem to be able to counterbalance the oligopsonistic behavior of investor-owned firms and thus the raw milk market is close to perfect competition, the dairy output market approaches at times Cournot levels. Nevertheless, due to the rather inelastic supply of both products the market power indices lead to rather high mark-downs. In the economic literature a series of suggestions exist that might be feasible for the German dairy supply chain to lower these effects.

Procurement behavior on the German raw milk market is fairly competitive. Actions to further counterbalance the market power of dairies, e.g. diversifying the homogenous product raw milk by switching to organic production, are only niche opportunities and will not greatly affect the entire market (Bundeskartellamt, 2009). Consequently, to lower the mark-downs, supply has to react more elastic to price changes (Hamilton & Sunding, 1997). A first step was already taken by abolishing entry barriers to the market in form of the milk quota, which as a result has likely increased the supply elasticity and lowered mark-downs (Graubner et al., 2011). Furthermore, other actions such as credit availability and technology transfer to dairy farmers as well as innovations at the farm level should be promoted since these allow raw milk production to respond more elastically (Atsbeha et al., 2016).

For the dairy output market, apart from the just mentioned supply elasticity enhancing policies, measures to lower the oligopsony conduct should be discussed. While cooperation among dairies might promote efficiency and the elasticity of supply, the German anti-trust agency is certain that further concentration at the dairy level will not increase their capability to achieve higher prices (Bundeskartellamt, 2009). One possibility to break the level of oligopsony is to increase the number of buyers for dairy products and thus the marketing options for dairies (Rude et al., 2011). One way to achieve this is to promote exports. However, the outcome of this approach might be rather restricted, since the quantities traded globally are growing but compared to the overall production still small.

For example, the main competitor of European dairy products New Zealand only produces a raw milk quantity similar to the magnitude of the German federal state Bavaria (Bundeskartellamt, 2009). Another way is to increase the number of domestic buyers by dissolving procurement alliances between larger retailers and smaller retailers and further lifting regulations that are entry barriers to the German retail market (Perloff et al., 2007). One more possibility to counterbalance an oligopsony is to move from homogenous, generic products to heterogeneous products (Sutton, 1998). This is in particular true for dairy cooperatives which mainly produce standard products under store label brands and thus only achieve low prices for their dairy products (Loy et al., 2016). Though, the creation of higher value-added and more heterogeneous dairy products through the establishment of brands, product innovation, and labelling in form of geographic indications or production method should be promoted (Henson & Reardon, 2005).

In general, actions that promote the price elasticities of supply along the supply chain smother the effects of market power. Nevertheless, due to the nature of raw milk and dairy production with its high asset specific the possibility to do so might be restricted. The oligopsony power of retailers could be offset by increasing the marketing possibilities by dairies, in particular cooperatives, either by increasing the number of possible buyers or promoting the diversification of dairy products through innovations and branding.

8.5 SHORTCOMINGS OF THE APPROACH

The developed theoretical model and the methodology to apply it empirically are capable to determine the level of oligopsony power on two subsequent markets only using price time series. Nevertheless, to model the German dairy supply chain strong assumption had to be drawn. Even though these assumptions seem to be valid and approximate the German dairy supply chain well, these still limit the transfer of the obtained results to the real world. In the following the shortcomings and their implications for the explanatory power of the results will be discussed.

While one of the main criticisms of the NEIO framework, the static estimation of a dynamic concept, is avoided by the estimation strategy, the model is still not completely dynamic. The negligence of any form of adjustment cost due to the complexity of the procedure leaves us with a static market power index, even though it varies with time and considers its history, its nature is not a completely dynamic one. A consequence is that the 'static' estimate of market power might be overestimating the true extent of the oligopsony level, since market participants that consider these dynamics might act more competitively (Perloff et al., 2007). Adding the dynamic component to the here described theoretic model would be an interesting task for future research.

Apart from the negligence of adjustment cost also the *a priori* choice of quantities to be the strategic variable limits the explanatory power of the approach. While the approach permits to evaluate the level of market power and through the supply elasticity its effect on the price, it is not possible to make any statement on the strategic game played by the market participants (Sckokai et al., 2013).

Another criticism is the assumption of homogenous products. While raw milk seems to be a quite homogenous product market, the aggregation across dairy products is at least questionable. Due to the lack of available data, the share weighted dairy output price was calculated using prices determined on the German dairy stock market for generic products. Nevertheless, the German dairy product market also includes a vast majority of heterogonous brand products, which achieve tremendously higher price. While the obtained results might describe the extent of market power for the generic dairy products well, the extent of market power on brand dairy product markets might be lower (Perloff et al., 2007).

The exclusion of exports is a limitation to this study. The exclusion of exports limits dairies in the theoretical model to sell their products to the German retail industry and deprives them of more marketing options. While indeed only small quantities of raw milk are imported and exported, more than half of dairies products produced by German dairies in terms of milk equivalent quantity are actually exported. Thus, excluding exports again might overestimate the market power of retailers. Nevertheless, the exclusion of trade is a necessary requirement in many market power studies due to lack on data of export prices, destinations, and quantities (Sckokai et al., 2013).

Even though, as described before, some drawn assumptions might limit the explanatory power of the obtained results, these were necessary requirements to limit the complexity or due to the availability of data. However, it is desirable to relax some assumptions in future research.

8.6 SUMMARY

The aim of this chapter was to give an interpretation to the results described in the empirical application section. In general, the estimates for the market power indices, price elasticities of supply are in line with the estimates of comparable

studies. Furthermore, the large buyer power indices are also no curiosity, but rather common on agricultural markets.

The rather low values of the market power index on the German raw milk market can be explained by the strong presence of German cooperatives, which promote the interests of dairy farmers and seem to erase the issue of a hold up by dairies. The interest of dairies in a constant flow of raw milk as an input to their production to be able to fully utilize capacities and generate the highest return on investment possibly further explains the low level of oligopsony on the German raw milk market.

While the German dairy output market is also closer to perfect competition than monopsony, the value range of 0.14 to 0.22 for the retailers' market power index proves the existence of significant market imperfections. Since five retailers control food retailing in Germany, the obtained value for the market power index is close to the Cournot outcome of 0.2 and consequently not surprising. On the consumer market fierce competition for market shares seems to exist and the exertion of oligopsony power on the procurement market, here for dairy products, is a possibility for retailers to generate larger profits. Nevertheless, the concentration of German retailing seem to have passed a concentration threshold around the year 2010 that actually has boosted competition on the market and lowered the level of oligopsony conduct (Sexton, 2013).

The price elasticity of raw milk supply as well as for dairy output is inelastic. The reason for this is the high asset specificity and thus the need for large investments to increase supply. However, the abolishment of the milk quota has abolished a barrier of entry and consequently made the supply of raw milk gradually more elastic.

Through the interaction of the market power indices and the price elasticities of supply the buyer power indices (BPI) for both markets can be determined. Due to the inelasticity of supply, the BPIs are quite large and range from 0.21 to 0.48 on the German raw milk market and 0.49 to 0.82 on the German dairy output market. In absolute terms, the German raw milk price was marked-down by $0.08 \notin to 0.12 \notin$, and the share weighted dairy output price between $0.25 \notin$ and almost $0.40 \notin$. The level of mark-down was in the latter stages of the time horizon reduced on the German raw milk market due to the growing elasticity of supply, and on the German dairy output market due to the decreasing extent of oligopsony power.

In general, the results prove that the effect of market power on the prices of raw milk and dairy output are large. Even though, the levels of oligopsony power are closer to perfect competition than monopsony, the inelastic supply of raw milk and dairy output lead to large relative and absolute mark-downs, in particular on the German dairy output market. Because the consumer market is not part of the analysis, it is not clear whether retailers have passed the lower prices for dairy products on to consumers. Without knowledge on this, the conclusion can only be that the market behavior along the German dairy supply chain is anticompetitive and has lowered the overall welfare.

The literature provides insights on how to counter the negative effects of oligopsony. In general, actions, e.g. credit availability, knowledge transfer, etc., that promote the price elasticities of supply along the supply chain lower the markdowns. Nevertheless, due to the nature of raw milk and dairy production with its high asset specific the possibility to do so might be restricted. The oligopsony power of retailers could be offset by increasing the marketing possibilities by dairies, in particular cooperatives, either by increasing the number of possible buyers or promoting the diversification of dairy products through innovations and branding.

Nevertheless, a series of drawn assumption limit the explanatory power of this analysis. The negligence of adjustment costs and the exclusion of exports are likely to lead to an overestimation of the effect of market power. However, these are common and necessary requirements in most market power studies (Sckokai, et al., 2013). The relaxation of these assumptions is a desirable achievement in future approaches.

9 CONCLUSIONS

The identification of market power along the German dairy supply chain was the primary objective of this study. The report of the German anti-trust agency as well as a descriptive analysis of the chain's structure acknowledge that on each of the two markets of the supply chain, the raw milk market and the dairy product market, a large number of sellers face a small number of buyers (Bundeskartell-amt, 2009). Consequently, the threat of oligopsony power on both markets is a realistic approximation. To address the subsequent threat of oligopsony power in the analysis of market power along the German dairy supply chain a coherent model was required.

Therefore, three approaches, structure-conduct-performance paradigm (SCPP), new empirical industrial organization (NEIO), and asymmetric price transmission (APT), were revised and their suitability for the study on hand discussed. While, the SCPP was quickly dropped from considerations, because of endogeneity issues and the use of accounting data instead of economic data, NEIO and APT have been widely used in the economic literature. Both approaches have their benefits, but also heavy drawbacks. NEIO is based on structural models that allow deriving a measure of market power, but the data requirement is high and the dynamic concept is estimated statically in most applications. If estimation is dynamic, strong assumptions on the behavior of market participants have to be drawn to be able to identify a unique market equilibrium. APT, on the other side, has a low data requirement and applies time series techniques, but is only capable of assuming that market power is the source of revealed price asymmetries rather than proofing it. In addition, the true extent of market power cannot be evaluated. Furthermore, neither NEIO nor APT provide a model that allows identifying oligopsony power on two subsequent markets of a supply chain. From the theory review it was concluded that none of the discussed approaches is suitable in this case.

In the absence of a proper model, the NEIO and APT approach were merged what permitted to derive a model of oligopsony power on two subsequent markets. The two markets were structurally modeled in the spirit of NEIO with demand and supply equations, while accounting for possible oligopsony conduct in form of market power indices. In the next step the equilibrium displacement model (EDM), a sub branch of APT, is applied by using the supply and demand equations as a system of equations. This allows deriving the simultaneous partial equilibria for all endogenous variables, the prices and quantities. From these simultaneous partial equilibria two pricing equations can be deduced, which only incorporate the prices for material input and output, other non-material costs as well as the market power indices for processors and retailers. Consequently, quantities are eliminated and the data requirement significantly lowered in comparison to the classic NEIO framework.

To estimate the two derived pricing equations simultaneously and derive nonstatic estimates for the market power indices, three time series estimation techniques had to be linked. The vector error correction model (VECM), which accounts for stationarity of long-run relationships among the variables, commonly found between price series of integrated markets, is transformed via the simple two step estimation technique of Lütkepohl and Krätzig (2004). The Kalman-Filter is applied to the transformed VECM and time-varying parameters obtained. The time-variation is assumed to be caused by changes in oligopsony conduct and is extracted in form of dynamic factors through the application of dynamic factor analysis (DFA) from the VECM's parameter. In addition, the DFA also allows separating the dynamic factors from other constant components. The results of the DFA permit to uniquely identify the time-varying market power indices.

The derived theoretical model and the estimation strategy were applied to the German dairy supply chain using monthly data over the time horizon from January 2003 to December 2015. The data consisted of price time series comprising the price for raw milk, a dairy output price derived by share-weighting the prices for skim milk powder, Emmentaler and butter, as well as prices of other cost variables involved in the product transformation process along the chain. The initial statistical analysis revealed nonstationary of the variables and two long-rung cointegration relationships among them, justifying the use of a VECM in the further steps. Furthermore, the Chow forecast test proved parameter instability of the VECM. The estimation strategy was applied to a total of nine different VECM specifications differing in lag length and assumptions on the nature of exogenous cost variables. The specification with a lag length of two and weakly exogenous variables was the best fit for the data.

The empirical application was successful and revealed theoretically consistent and statistically significant market power indices on the German raw milk market and dairy output market for the investigated time horizon. While the raw milk market is close to perfect competition with a market power index value range of 0.038 to 0.048, the dairy output market is subject to larger market imperfections. The market power index of retailers lies in the interval of 0.14 to 0.22 and reaches probably Cournot level at times. Consequently, the five firms in control of German food retailing abused their oligopsony power by splitting the procurement market for dairy products among themselves. However, since 2010 the level of market imperfection significantly dropped on the dairy output market and the firms acted more competitively.

After calculating the respective price elasticities of supply it is possible to calculate buyer power indices that measure the effect of the market power level on the observed prices. The rather inelastic supply due to asset specificity and high investment costs lead in collaboration with the derived market power indices to large mark-downs. While the German raw milk price was marked-down by 21% to 48%, the mark-down of the share-weighted dairy output price ranges between 49% and 82% over the analyzed time period. However, as a consequence of the lower levels of oligopsony power since 2010 on the dairy output market also the corresponding mark-down declined. Overall, large rents were shifted in disadvantage of dairies and in particular of dairy farmers downstream the supply chain and welfare was lost. If the mark-downs resulted in lower consumer prices this might be socially desirable. Nevertheless, consumer markets were excluded from the analysis, and the markets of the German supply chain have to be marked as anticompetitive.

To overcome the market imperfections on each market of the German dairy supply chain different steps can be taken. While cooperative action on the raw milk market seems to be capable to counterbalance a possible oligopsony behavior of investor-owned firms and only low levels of oligopsony conduct are observed, the inelastic response of raw milk supply to price changes causes the rather large mark-downs. Consequently, policies that enhance the responsive-ness of supply, e.g. credit availability and technology transfer to dairy farmers, should be implemented (Atsbeha et al., 2016). A first step was already taken by abolishing the entry barrier in form of the milk quota, which has likely increased the supply elasticity and lowered mark-downs (Graubner et al., 2011).

For the dairy output market, apart from the just mentioned supply elasticity enhancing policies, measures to lower the oligopsony conduct should be discussed. One possibility is to increase the number of buyers of dairy products and thus the marketing options for dairies (Rude et al., 2011). This could be done in the case of the German dairy supply chain by promoting exports or dissolving procurement alliances between larger and smaller retailers. Another possibility to counterbalance an oligopsony is to move from homogenous, generic products to heterogeneous products through the creation of brands (Henson & Reardon, 2005; Sutton, 1998; Weatherell et al., 2003).

Several innovations make the presented approach a novelty in the field of market power studies. To our knowledge no attempts to model oligopsony power on two

vertically integrated markets exist so far in the economic literature. Furthermore, even though, the use of quadratic-flexible functional forms increases the complexity of the model, it also allows approximating the true technologies closer than any linear functional forms that have been widely used as the foundation for determining simultaneous partial equilibria (e.g. Lloyd et al., 2009). While the markets are model in the fashion of NEIO, the derivation of partial equilibrium equations and the final pricing equations significantly lower the data requirement and the necessary number of equations for estimation. Moreover, by assuming time-variation in the oligopsony conduct and applying the developed estimation strategy, the market power indices are adaptive rather than 'static'.

Nevertheless, a series of assumption that ultimately limit the explanatory power of this analysis had to be drawn in order to ensure the approach's empirical applicability. The negligence of adjustment costs and the exclusion of exports are likely to lead to an overestimation of the market power indices. However, these are common and necessary requirements in most market power studies (Sckokai, et al., 2013). Consequently, the values of the market power indices should be interpreted with care. The relaxation of these assumptions should be the focus of future research.

REFERENCES

- Abdulai, A. (2000), "Spatial Price Transmission and asymmetry in the Ghanaian Maize Market", *Journal of Development Economics*, Vol. 63, No. 2, pp. 327-349.
- Acharya, R.N., Kinnucan, H.W., Caudill, S.B. (2011), "Asymmetric farm-retail price transmission and market power: a new test", Applied Economics, Vol. 43, pp. 4759-4768.
- Acosta, A., and Valdés, A. (2014), "Vertical Price Transmission of Milk Prices: Are Small Dairy Producers Efficiently Integrated Into Markets?", Agribusiness, Vol. 30, No. 1, pp. 56-63.
- Acosta, A., Ihle, R., and Robles, M. (2014), "Spatial Price Transmission of Soaring Milk Prices From Global to Domestic Markets", *Agribusiness*, Vol. 30, No. 1, pp. 64-73.
- Aguiar, D.R.D., and Santana, J.A. (2002), "Asymmetry in farm to retail price transmission: evidence from Brazil", *Agribusiness*, Vol. 18, No.1, pp. 37–48.
- Agrarmarkt Informations-Gesellschaft (AMI) (ed.) (different volumes), AMI Markt Bilanz Getreide, Bonn, Germany: AMI GmbH.
- Agrarmarkt Informations-Gesellschaft (AMI) (ed.) (different volumes), AMI Markt Bilanz Milch, Bonn, Germany: AMI GmbH.
- Ahn, B.-I., and Lee, H. (2010), "An equilibrium displacement approach to oligopoly market analysis: an application to trade in the Korean infant formula market", *Agricultural Economics*, Vol. 41, pp. 101-109.
- Ahn, S. K., and Reinsel, G. C. (1990), "Estimation of partially nonstationary multivariate autoregressive models", *Journal of the American Statistical Association*, Vol. 85, pp. 813-823.
- Anders, S.M. (2008), "Imperfect Competition in German Food Retailing", Atlantic Economic Journal, Vol. 36, No. 4, pp. 441-454.
- Appelbaum, E. (1982), "The Estimation of the Degree of Oligopoly Power", *Journal of Econometrics*, Vol. 19, No. 2-3, pp. 287–99.
- Appelbaum, E. (1979), "Testing Price Taking Behavior", *Journal of Econometrics*, Vol. 9, pp. 283-294.
- Atsbeha, D.M., Kristofersson, D., and Rickertsen, K. (2016), "Component supply responses in dairy production", *European Review of Agricultural Economics*, Vol. 43, No. 5, pp. 193-215.
- Awokuse, T.O., and Wang. X. (2009), "Threshold Effects and Asymmetric Price Adjustments in U.S. Dairy Markets", Canadian Journal of Agricultural Economics, Vol. 57, pp. 269-286.
- Azzam, A.M., and Pagoulatos, E. (1990), "Testing Oligopolistic and Oligopsonistic Behaviour: An Application to the U.S. Meat-Packing Industry", *Journal of Agricultural Economics*, Vol. 41, No. 3, pp. 362-369.

- Bailey, D., and Brorsen, B.W. (1989), "Price Asymmetry in Spatial Fed Cattle Markets", Western Journal of Agricultural Economics, Vol. 14, No. 2, pp. 246-252.
- Bain, J.S. (1968), Industrial Organization, New York: John Wiley & Sons.
- Bain, J.S. (1951), "Relation of Profit Rate to Industry Concentration: American Manufacturing, 1936-1940", The Quarterly Journal of Economics, Vol. 65, No. 3, pp. 293-324.
- Balke, N.S., and Fomby, T.B. (1997), "Threshold Cointegration", International Economic Review, Vol. 38, No. 3, pp. 627-645.
- Bakucs, L.Z., Fałkowski, J., and Fertő, I. (2012), "Price transmission in the milk sectors of Poland and Hungary", *Post-Communist Economics*, Vol. 24, No. 3, pp. 419-432.
- Bakucs, L.Z., Fertő, I., Hockmann, H. and Perekhozhuk, O. (2010), "Estimating Oligopsony Power in The Hungarian Market for Slaughter Hogs", *Eastern European Economics*, Vol. 48, No. 1, pp. 64-78.
- Bakucs, L.Z., and Fertő, I. (2005), "Marketing Margins and Price Transmission on the Hungarian Pork Meat Market", *Agribusiness*, Vol. 21, No. 2, pp. 273-286.
- Bayerische Landesanstalt für Landwirtschaft (Lfl) and Landesanstalt für Entwicklung der Landwirtschaft und der Ländlichen Räume (LEL) (2016), Agrarmärkte, Schwäbisch Gmünd, Germany: LEL.
- Bayerische Landesanstalt f
 ür Landwirtschaft (Lfl) and Landesanstalt f
 ür Entwicklung der Landwirtschaft und der L
 ändlichen R
 äume (LEL) (2015), Agrarm
 ärkte, Schw
 äbisch Gm
 ünd, Germany: LEL.
- Bell, R., Davies, R., and Howard, E. (1997), "The Changing Structure of Food retailing in Europe: The Implications for Strategy", Long Range Planning, Vol. 30, No. 6, pp. 853-861.
- Bhuyan, S. (2014), "Visiting an old battleground in empirical industrial organization: SCP versus NEIO", Applied Economics Letters, Vol. 21, No. 11, pp. 751-754.
- Bhuyan, S. and Lopez, R.A. (1998), "Oligopoly Power and Allocative Efficiency in US Food and Tobacco Industries", *Journal of Agricultural Economics*, Vol. 49, No. 3, pp.434-442.
- Blair, R.D., and Harrison, J.L. (1993), Monopsony: Antitrust law and economics, Princeton, N.J., USA: Princeton University Press.
- Bliss, C. (1988), "A Theory of Retail Pricing", The Journal of Industrial Economics, Vol. 36, No. 4, pp. 375-391.
- Bolotova, Y.V., and Novakovic, A.M. (2015), "An Empirical Analysis of Wholesale Cheese Pricing Practices on the Chicago Mercantile Exchange (CME) Spot Cheese Market", International Food and Agribusiness Management Review, Vol. 18, No. 3, pp. 49-66.
- Bolotova, Y.V., and Novakovic, A.M. (2012), "The Impact of the New York State Milk Price Gouging Law on the Price Transmission Process and Supermarket Pricing Strategies in the Fluid Whole Milk Market", Agribusiness, Vol. 28, No. 4, pp. 377-399.

- Borenstein, S., Cameron, A.C., and Gilbert, R. (1997), "Do Gasoline Prices respond asymmetrically to Crude Oil Price Changes?", *Quarterly Journal of Economics*, Vol. 112, No. 1, pp. 305-339.
- Bouamra-Mechemache, Z., Réquillart, V., and Jongeneel, R. (2008), "Impact of a gradual increase in milk quotas on the EU dairy sector", *European Review of Agricultural Economics*, Vol. 35, No. 4: pp. 461-491.
- Bresnahan, T.F. (1989), "Empirical Studies of Industries with Market Power", Schmalensee, R. and Willig, R.D. (eds), Handbook of Industrial Economics 2, Amsterdam, The Netherlands: North Holland, pp. 1011-1057.
- Bresnahan, T.F. (1982), "The Oligopoly Concept is identified", *Economic Letters*, Vol. 10, pp. 87-92.
- Bundeskartellamt (2014), Sektoruntersuchung Lebensmitteleinzelhandel. Darstellung und Analyse der Strukturen und des Beschaffungsverhaltens auf den Märkten des Lebensmitteleinzelhandels in Deutschland Bericht gemäß § 32 e GWB - September 2014, Bonn, Germany: Bundeskartellamt.
- Bundeskartellamt (2012), Sektoruntersuchung Milch Abschlussbericht gemäß § 32e GWB – Januar 2012, Bonn, Germany: Bundeskartellamt.
- Bundeskartellamt (2009), *Sektoruntersuchung Milch. Zwischenbericht*, Bonn, Germany: Bundeskartellamt.
- Bundesministerium für Ernährung und Landwirtschaft (BMLE) (different volumes), Statistisches Jahrbuch über Ernährung, Landwirtschaft und Forsten der Bundesrepublik Deutschland, Münster-Hiltrup, Germany: Landwirtschaftsverlag GmbH.
- Bundesministerium für Ernährung, Landwirtschaft und Verbraucherschutz (BMLEV) (different volumes), *Statistisches Jahrbuch über Ernährung, Landwirtschaft und Forsten der Bundesrepublik Deutschland*, Münster-Hiltrup, Germany: Landwirtschaftsverlag GmbH.
- Bundesministerium für Wirtschaft und Energie (BMWE) (2016), Entwicklung von Energiepreisen und Preisindizes, available at: http://www.bmwi.de/DE/Themen/ Energie/Energiedaten-und-analysen/Energiedaten/energiepreise-energiekosten.html (accessed 28.12.2016).
- Bundesvereinigung der Deutschen Ernährungsindustrie (BVE) (2016), Jahresbericht 2015-2016 - Lebensmittelherstellung in einer Kultur der Widersprüchlichkeit, Berlin, Germany: BVE.
- Bundesvereinigung der Deutschen Ernährungsindustrie (BVE) (2007), Jahresbericht 2006-2007 - Ernährungsindustrie – ein wichtiger Wirtschaftsfaktor in Deutschland, Berlin, Germany: BVE.
- Burnham, K. P., and Anderson, D. R. (2004), "Multimodel inference: understanding AIC and BIC in Model Selection", Sociological Methods & Research, Vol. 33, pp. 261–304.

- Cabral, L.M.B. (1995), "Conjectural variations as a reduced form", *Economic Letters*, Vol. 49, pp. 397-402.
- Cakir, M., and Balagtas, J.V. (2012), "Estimating Market Power of U.S. Dairy Cooperatives in the Fluid Milk Market", American Journal of Agricultural Economics, Vol. 94, No. 3, pp. 647-658.
- Capps, O., and Sherwell, P. (2007), "Alternative Approaches in Detecting Asymmetry in Farm-Retail Price Transmission of Fluid Milk", Agribusiness, Vol. 23, No. 3, pp. 313-331.
- Carman, H.F., and Sexton, R.J. (2005), "Supermarket Fluid Milk Pricing Practices in the Western United States", *Agribusiness*, Vol. 21, No. 4, pp. 509-530.
- Chambers, R.G. (1988), *Applied Production Analysis: A Dual Approach*, Cambridge, UK: Cambridge University Press.
- Chavas, J.-P., and Mehta, A. (2004), "Price Dynamics in a Vertical Sector: The Case of Butter", American Journal of Agricultural Economics, Vol. 86, No. 4, pp. 1078-1093.
- Chen, Z. (2008), *Defining Market Power*, Department of Economics, Carleton University, available at: http://http-server.carleton.ca/~zchen/Defining%20Buyer%20 Power%20 Revised.pdf (accessed 30.12.2016).
- Chen, Z. (2003), "Dominant retailers and the countervailing power hypothesis", *The RAND Journal of Economics*, Vol. 34, No. 4, pp. 612-625.
- Chidmi, B., Lopez, R.A., and Cotteril, R.W. (2004), "Retail Oligopoly Power, Dairy Compact, and Boston Milk Prices", *Agribusiness*, Vol. 21, No. 4, pp. 477-491.
- Claeskens, G., and Hjort, N. L. (2008), *Model Selection and Model Averaging*, Cambridge, UK: Cambridge University Press.
- Clarke, R., Davies, S.W., Dobson, P.W., and Waterson, M. (2003), "Buyer Power and Competition in European Food Retailing", *Review of Industrial Organization*, Vol. 22, No. 4, pp. 337-340.
- Clarke, R. and S.W. Davies (1982), "Market Structure and Price-Cost Margins", *Economica*, Vol. 49, pp. 277–87.
- Colman, D., Solomon, A., and Gill, L. (2005), "Supply response of UK milk producers", *Agricultural Economics*, Vol. 32, pp. 239–251.
- Consultative Commission on Industrial Change (2008), *The evolution of the high-volume retail sector in Europe over the past 5 years - Final Report*, available at http://www. eesc.europa.eu/?i=portal.en.industrial-sectors-retail-external-studies.6534 with_annex.doc (accessed 24 May 2016).
- Corts, K.S. (1999), "Conduct Parameters and the Measurement of Market Power", *Journal of Econometrics*, Vol. 88, No. 2, pp. 227–250.
- Crespi, J. M., Saitone, T. L., and Sexton, R. J. (2012), "Competition in the U.S. Farm Product Markets: Do Long-Run Incentives Trump Short-Run Market Power?", *Applied Economic Perspectives and Policy*, Vol. 34, No. 4, pp. 669-695.

- Christensen, L.R., Jorgenson, D.W., and Lau, L.J. (1973), "Transcendental Logarithmic Production Frontiers", *The Review of Economics and Statistics*, Vol. 55, No. 1, pp. 28-45.
- Collins, N.R., and Preston, L.E. (1969), "Price-Cost Margins and Industry Structure", *The Review of Economics and Statistics*, Vol. 51, No. 3, pp. 271-286.
- Cowling, K., and Waterson, M. (1976), "Price-Cost Margins and Market Structure", Economica, New Series, Vol. 43, No. 171, pp. 267-274.
- Davis, C., Blayney, D., Cooper, J., and Yen, S. (2009), "An Analysis of Demand Elasticities for Fluid Milk Products in the U.S.", Contributed Paper prepared for presentation at the International Association of Agricultural Economists Meeting, Beijing, China, August 16-22, 2009.
- De Mello, M., and Brandao, A., (1999), "Measuring the Market Power of the Portuguese Milk Industry", *International Journal of the Economics and Business*, Vol. 6, No. 2, pp. 209-222.
- Demsetz, H. (1973), "Industry structure, market rivalry, and public policy", *Journal of Law & Economics*, Vol. 16, No. 1, pp. 1-9.
- Deodhar, S.Y., and Sheldon, I.M. (1995), "Is Foreign Trade (Im)Perfectly Competitive?: An Analysis of the German Market for Banana Imports", *Journal of Agricultural Economics*, Vol. 46, No. 3, pp. 336-348.
- Deutsche Bundesbank (2016), Zeitreihe BBK01.SU0304: Geldmarktsätze / EONIA / Monatsdurchschnitt, available at: https://www.bundesbank.de/Navigation/DE/Statistiken/ Zeitreihen_Datenbanken/Geld_und_Kapitalmaerkte/geld_und_kapitalmaerkte_details_value_node.html?tsId=BBK01.SU0304&listId=www_s510_mb03_neu (accessed 28.12.2016).
- Dickey, D. A., and Fuller, W.A. (1981), "Likelihood ratio statistics for autoregressive time series with a unit root", *Econometrica*, Vol. 49, No. 4, pp. 1057-1072.
- Dixit, A. (1986), "Comparative Statics for Oligopoly", *International Economic Review*, Vol. 27, No. 1, pp. 107-122.
- Djuric, I., and Götz, L. (2016), "Export restrictions Do consumers really benefit? The wheat-to-bread supply chain in Serbia", *Food Policy*, Vol. 63, pp. 112-123.
- Dobson, P.W., Waterson, M., and Davies, S.W. (2003), "The Patterns and Implications of Increasing Concentration in European Food Retailing", *Journal of Agricultural Economics*, Vol. 54, No. 1, pp. 111-125.
- Dobson, P.W., and Waterson, M. (1997), "Countervailing power and consumer prices", *The Economic Journal*, Vol. 107, No. 441, pp. 418-430.
- Dockner, E.J. (1992), "A dynamic Theory of Conjectural Variations", *The Journal of Industrial Economics*, Vol. 40, No. 4, pp. 377-395.
- Dudová, B., and Bečvářová, V. (2015), "The Character of Price Transmission within Milk Commodity Chain in the Czech Republic", Acta Universitatis Agriculturae et Silviculturae Mendelianae Brunensis, Vol. 63, No. 3, pp. 887-892.
- Enders, W., and Granger, C.W.J. (1998), "Unit-root tests and asymmetric adjustment with an example using the term structure of interest rates", *Journal of Business & Economic Statistics*, Vol. 16, No. pp. 304-311.
- Engle, R.F., and Granger, C.W.J. (1987), "Co-Integration and Error Correction: Representation, Estimation, and Testing", *Econometrica*, Vol. 55, No. 2, pp. 251-276.
- European Commission (2009), Analysis of price transmission along the food supply chain in the EU, SEC(2009) 1450, Brussels, Belgium: Commission of the European Communities.
- Fahlbusch, M., Steffen, N., Spiller, A., Hellberg-Bahr, A., and Brümmer, B. (2013), "Der Markt für Milch und Milcherzeugnisse im Jahr 2012", German Journal of Agricultural Economics, Vol. 62, pp. 46-66.
- Fahlbusch, M., Steffen, N., Brümmer, B., and Spiller, A. (2011), "Der Markt für Milch und Milcherzeugnisse", German Journal of Agricultural Economics, Vol. 60, pp. 52-71.
- Fahlbusch, M., Bahr, A., Brümmer, B., and Spiller, A. (2009), "Der Markt für Milch und Milcherzeugnisse", *Agrarwirtschaft*, Vol. 58, No. 1, pp. 36-52.
- Fałkowski, J. (2010), "Price transmission and market power in a transition context: evidence from the Polish fluid milk sector", *Post-Communist Economics*, Vol. 22, No. 4, pp. 513-529.
- Farrell, M.J. (1952), "Irreversible Demand Functions", *Econometrica*, Vol. 20, No. 2, pp. 171-186.
- Fershtman, C., and Kamien, M.I. (1987), "Dynamic Duopolistic Competition with Sticky Prices", *Econometrica*, Vol. 55, pp.1151-1164.
- Frey, G., and Manera, M. (2007), "Econometric Models of asymmetric price transmission", *Journal of Economic Surveys*, Vol. 21, No. 2, pp.349-415.
- Gardner, B.L. (1975), "The Farm-Retail Price Spread in a Competitive Industry", *American Journal of Agricultural Economics*, Vol. 57, No. 3, pp. 399-409.
- Genesove, D., and Mullin, W.P. (1998), "Testing Oligopoly Models: Conduct and Cost in the Sugar Industry, 1890-1914", The RAND Journal of Economics, Vol. 29, No. 2, pp. 355-377.
- Gerlach, S., Spiller, A., and Wocken, C. (2006), "Der Markt für Milch und Milcherzeugnisse", Agrarwirtschaft, Vol. 55, No. 1, pp. 29-50.
- Gohin, A., and Guyomard, H. (2000), "Measuring Market Power for Food Retail Activities: French Evidence", Journal of Agricultural Economics, Vol. 51, No. 2, pp. 181-195.
- Gollop, F.M. and Roberts, M.J. (1979), "Firm Interdependence in Oligopolistic Markets", *Journal of Econometrics*, Vol. 10, pp. 313-331.
- Goodwin, B.K., and Brester, G.W. (1995), "Structural Change in Factor Demand Relationships in the U.S. Food and Kindred Industry", *American Journal of Agricultural Economics*, Vol. 77, pp. 69-79.

- Goodwin, B.K., and Harper, D.C. (2000), "Price Transmission, threshold Behavior, and Asymmetric Adjustment in the U.S. Pork Sector", *Journal of Agricultural and Applied Economics*, Vol. 32, No. 3, pp. 543-553.
- Goodwin, B.K., and Holt, M.T. (1999), "Asymmetric Adjustment and Price Transmission in the U.S. Beef Sector", American Journal of Agricultural Economics, Vol. 81, No. 3, pp. 630-637.
- Goodwin, B.K., and Piggott, N.E. (2001), "Spatial Market Integration in the Presence of Threshold Effects", *American Journal of Agricultural Economics*, Vol. 83, No. 2, pp. 302-317.
- Granger, C.W.J., and Lee, T.H. (1989), "Investigation of Production, Sales and Inventory Relationships using Multicointegration and non-symmetric Error Correction Models", *Journal of Applied Econometrics*, Vol. 4, pp. 135-159.
- Granger, C.W.J., and Newbold, P. (1974), "Spurious Regressions in Econometrics", *Journal of Econometrics*, Vol. 2, No. 2, pp. 111-120.
- Graubner, M., Koller, I., Salhofer, K., and Balmann, A. (2011), "Cooperative versus noncooperative spatial competition for milk", *European Review of Agricultural Economics*, Vol. 38, No. 1, pp. 99-118.
- Grau, A., Hockmann, H., and Levkovych, I. (2015), "Dairy cooperatives at the crossroads", *British Food Journal*, Vol. 117, No. 10, pp. 2515-2531.
- Green, E.J., and Porter, R.H. (1984), "Noncooperative Collusion under Imperfect Price Information", *Econometrica*, Vol. 52, pp. 87-100.
- Gregory, A.W., and Hansen, B.E. (1996), "Residual-based test for cointegration in models with regime shifts", *Journal of Econometrics*, Vol. 70, pp. 99-126.
- Guy, C., Bennison, D., Clarke, R. (2004), "Scale economies and super-store retailing: New evidence from the UK", *Journal of Retailing and Consumer Services*, Vol. 12, No. 2, pp. 73-81.
- Haavelmo, T. (1944), "The Probability Approach in Econometrics", *Econometrica*, Vol. 12, pp. 1-115.
- Hall, L.L., Tomek, W. G., Ruther, D. L., and Kyerine, S. S. (1981), "Case Studies in the Transmission of Farm Prices", Dep. Agr. Econ. A. E. Res. No. 81-12, Cornell University, Aug. 1981.
- Hamilton, J. D. (1989), "A new approach to the economic analysis of nonstationary time series and the business cycle", *Econometrica*, Vol. 57, pp. 357-384.
- Hamilton, S.F., and Sunding, D.L. (1997), "The Effect of Farm Supply Shifts on Concentration and Market Power in the Food Processing Sector", American Journal of Agricultural Economics, Vol. 79, No. 2, pp. 524-531.
- Hahn, W., Stewart, H., Blayney, D.P., and Davis, C.G. (2016), "Modeling price transmission between farm and retail prices: a soft switches approach", *Agricultural Economics*, Vol. 47, No. 2, pp. 193-203.

- Harvey, A.C. (1989), Forecasting, structural time series models and the Kalman filter, Cambridge, UK: Cambridge University Press.
- Hellberg-Bahr, A., Pfeuffer, M., Steffen, N., Spiller, A., and Brümmer, B. (2010), Preisbildungssysteme in der Milchwirtschaft – Ein Überblick über die Supply Chain Milch, Göttingen, Germany: Department für Agrarökonomie und Rurale Entwicklung, Universität Göttingen.
- Hendry, D.F. and Doornik, J.A. (2001), *Modelling Dynamic Systems using PcGive 10*, London, UK: Timberlake.
- Henson, S. and Reardon, T. (2005), "Private agri-food standards: Implications for food policy and the agri-food system", *Food Policy*, Vol. 30, pp. 241-253.
- Hockmann, H., and Vőnecki, É. (2009), "Collusion in the Hungarian Market for Raw Milk", *Outlook on Agriculture*, Vol. 38, No. 1, pp. 39-45.
- Holmes, E.E., Ward, E.J., and Wills, K. (2012), "MARSS: Multivariate Autoregressive Statespace Models for Analyzing Time-series Data", *The R Journal*, Vol. 4, No. 1, pp. 11-19.
- Hotelling, H. (1932), "Edgeworth's Taxation Paradox and the Nature of Demand and Supply Functions", Journal of Political Economy, Vol. 40, pp. 577-616.
- Houck, J.P. (1977), "An Approach to specifying and estimating nonreversible Functions", *American Journal of Agricultural Economics*, Vol. 59, No. 3, pp. 570-572.
- Hovhannisyan, V., and Bozic, M. (2013), "A Benefit-Function Approach to Studying Market Power: An Application to the U.S. Yogurt Market", *Journal of Agricultural and Resource Economics*, Vol. 38, No. 2, pp. 159-173.
- Hovhannisyan, V., and Gould, B.W. (2012), "A Structural Model of the Analysis of Retail Market Power: The Case of Fluid Milk", American Journal of Agricultural Economics, Vol. 94, No. 1, pp. 67-79.
- Hurvich, C.M., and Tsai, C.-L. (1995), "Model Selection for Extended Quasi-Likelihood Models in Small Samples", *Biometrics*, Vol. 51, pp. 1077-84.
- Hurvich, C.M., and Tsai, C.-L. (1989), "Regression and Time Series Model Selection in Small Samples", *Biometrika*, Vo. 76, pp. 297-307.
- Johansen, S. (1991), "Estimation and Hypothesis Testing of Cointegration Vectors in Gaussian Vector Autoregressive Models", *Econometrica*, Vol. 59, No.6, pp. 1551–1580.
- Johansen, S. (1988), "Statistical Analysis of Cointegration Vectors", *Journal of Economic Dynamics and Control*, Vol. 12, pp. 231-254.
- Juselius, K. (2006), The cointegrated VAR Model, Oxford, UK: Oxford University Press.
- Kadiyali, V., Sudhir, K., and Rao, V.R. (2001), "Structural analysis of competitive behaviour: New Empirical Industrial Organization methods in marketing", *International Journal* of Research in Marketing, Vol. 18, pp. 161-186.
- Kalman, R.E. (1960), "A new approach to linear filtering and prediction problems", *Journal of Basic Engineering, Transactions ASMA*, Series D82, 35-45.

- Karp, L.S., and Perloff, J.M. (1993a), "Open-loop and feedback models of dynamic oligopoly", International Journal of Industrial Organization, Vol. 11, pp. 369-389.
- Karp, L.S., and Perloff, J.M. (1993b), "A Dynamic Model of Oligopoly in the Coffee Export Market", American Journal of Agricultural Economics, Vol. 75, No. 2, pp. 448-457.
- Karp, L.S., and Perloff, J.M. (1989), "Dynamic Oligopoly un the Rice Export Market", *The Review of Economics and Statistics*, Vol. 71, No. 3, pp 462-470.
- Katchova, A.L., Sheldon, I.M., and Miranda, M.J. (2005), "A Dynamic Model of Oligopoly and Oligopsony in the U.S. Potato-Processing Industry", *Agribusiness*, Vol. 21, No. 3, pp. 409-428.
- Kreps, D.M. (1990), *Game Theory and Economic Modelling*, New York, USA: Oxford University Press.
- Kharin, S., (2015), "Vertical price transmission along the dairy supply chain in Russia", *Studies in Agricultural Economics*, Vol. 117, pp. 80-85.
- Kilpatrick, R.W. (1967): "The Choice among Alternative Measures of Industrial Concentration", The Review of Economics and Statistics, Vol. 49, pp. 258-260.
- Kinnucan, H.W., and Forker, O.D. (1987), "Asymmetry in Farm-Retail Price Transmission for Major Dairy Products", American Journal of Agricultural Economics, Vol. 69, No. 2, pp. 285-292.
- Krolzig, H.-M. (1997), Markov Switching Vector Autoregressions Modelling, Statistical Inference and Application to Business Cycle Analysis, Berlin, Germany: Springer.
- Lajdová, Z., and Bielik, P. (2015), "Vertical Price Transmission Analysis: The Case of Milk in the Slovak Dairy Sector", Applied Studies in Agribusiness and Commerce, Vol. 7, No. 4-5, pp. 89-96.
- Lass, D.A (2005), "Asymmetric Response of Retail Milk Prices in the Northeast Revisited", *Agribusiness*, Vol. 21, No. 4, pp. 493-508.
- Lass, D.A., Adanu, M., and Allen, P.G. (2001), "Impacts of the Northeast Dairy Compact on New England Retail Prices", *Agricultural and Resource Economics Review*, Vol. 30, No. 1, pp. 83-92.
- Lau, L.J. (1982), "On Identifying the Degree of Competitiveness from Industry Price and Output Data", *Economic Letters*, Vol. 10, pp. 93-99.
- Lerner, A.P. (1934), "The Concept of Monopoly and the Measurement of Monopoly Power", *The Review of Economic Studies*, Vol. 1, No. 3, pp. 157–175.
- Lloyd, T., McCorriston, S., Morgan, W., Rayner, A., and Weldegebriel, H. (2009), "Buyer Power in the U.K. Food Retailing: A 'Frist-Pass' Test", *Journal of Agricultural & Food Industrial Organization*, Vol. 7, No. 1, Article 5.
- Liu, D.J., Sun, C.-H. and Kaiser, H.M. (1995), "Market Conduct under Government Price Intervention in the U.S. Dairy Industry", *Journal of Agricultural and Resource Economics*, Vol. 20, No. 2, pp. 301-315.

- Lovell, M. (2008), "A Simple Proof of the FWL Theorem", *Journal of Economic Education*, Vol. 39, No. 1, pp. 88–91.
- Loy, J.-P., Weiss, C.R., and Glauben, T. (2016), "Asymmetric cost pass-through? Empirical evidence on the role of market power, search and menu costs", *Journal of Economic Behavior & Organization*, Vol. 123, pp. 184-192.
- Loy, J.-P., Holm, T., Steinhagen, C., and Glauben, T. (2015), "Cost pass-through in differentiated markets: a disaggregated study for milk and butter", *European Review of Agricultural Economics*, Vol. 42, No. 3, pp. 441-471.
- Lütkepohl, H., & Krätzig, M. (2004), *Applied Time Series Econometrics*. Cambridge, UK: Cambridge University Press.
- Lütkepohl, H., Saikkonen, P., and Trenkler, C. (2004), "Testing for the Cointegrating Rank of a VAR Process with Level Shift at Unknown Time", *Econometrica*, Vol. 72, No. 2, pp. 647-662.
- Madau, F.A., Furesi, R., and Pulina, P. (2016), "The existence of buyer power in the Italian fresh milk supply chain", *British Food Journal*, Vol. 118, No. 1, pp. 70-82.
- Mason, E.S. (1949), "The Current State of the Monopoly Problem in the United States", Harvard Law Review, Vol. 62, pp. 1265-1285.
- Mason, E.S. (1939), "Price and Production Policies of Large-Scale Enterprise", American Economic Review, Vol. 29, pp. 61-74.
- Marion, B.W., Mueller, W.F., Cotteril, R.W., Geithman, F.E., and Schmelzer, J.R. (1979), "The Price and Profit of Leading Food Chains", *American Journal of Agricultural Economics*, Vol. 61, No. 3, pp. 420-433.
- Marshall, A. (1936), Principles of Economics, New York, USA: Macmillian.
- Martin, S. (1979), "Advertising, Concentration, and Profitability: The Simultaneity Problem", The Bell Journal of Economics, Vol. 10, No. 2, pp. 639-647.
- McCorriston, S., Morgan, C.W., and Rayner, A.J. (2001), "Price transmission: the interaction between market power and returns to scale", *European Review of Agricultural Economics*, Vol. 28, No. 2, pp. 143-159.
- McCorriston, S., Morgan, C.W., and Rayner, A.J. (1998), "Processing Technology, Market Power and Price Transmission", *Journal of Agricultural Economics*, Vol. 49, No. 2, pp. 185-201.
- Mérel, P.R. (2009), "Measuring market power in the French Comté cheese market", European Review of Agricultural Economics Vol. 36, No. 1, pp. 31-51.
- Meyer, J., and von Cramon-Taubadel, S. (2004), "Asymmetric Price Transmission: A Survey", Journal of Agricultural Economics, Vol. 55, pp. 581-611.
- Milchindustrie-Verband e.V. (MIV) (2016), Fakten Milch Milch und mehr die deutsche Milchwirtschaft auf einen Blick, Berlin, Germany: MIV.

- Miller, R.A. (1967), "Marginal Concentration Ratios and Industrial Profit Rates", *Southern Economic Journal*, Vol. 34, pp. 259-267.
- Miller, D.J., and Hayenga, M.L. (2001) "Price Cycles and Asymmetric Price Transmission in the U.S. Pork Market", *American Journal of Agricultural Economics*, Vol. 83, No. 3, pp. 551-562.
- Morrison Paul, C.J. (2001), "Market and Cost Structure in the U.S. Beef Packing Industry: A Plant-Level Analysis", *American Journal of Agricultural Economics*, Vol. 83, No. 1, pp. 64-76.
- Morrison Paul, C.J. (1999), "Production Structure and Trends in the U.S. Meat and Poultry Products Industries", *Journal of Agricultural and Resource Economics*, Vol. 24, No. 2, pp. 281-298.
- Morrison Paul, C.J., and Siegel, D.S. (1999), "Scale economies and industry agglomeration externalities: a dynamic cost function approach", *The American Economic Review*, Vol. 89, No. 1, pp. 272-290.
- Muth, M.K,. and Wohlgenant, M.K. (1999), "Measuring the Degree of Oligopsony Power in the Beef Packing Industry in the Absence of Marketing Input Quantity Data", *Journal* of Agricultural and Resource Economics, Vol. 24 No. 2, pp. 299-312.
- Murray, B. (1995), "Measuring Oligopsony Power with shadow Prices U.S. Markets for Pulpwood and Sawlogs", *The Review of Economics and Statistics*, Vol. 77, No. 3, pp. 486-498.
- O'Donnell, C.J., Griffith, G.R., Nightingale, J.J., and Piggott, R.R. (2007), "Testing for Market Power in the Australian Grains and Oilseeds Industries", *Agribusiness*, Vol. 23, No. 3, pp. 349-376.
- Organisation for Economic Co-operation and Development (OECD) (2008), *Monopsony* and buyer power, available at: http://www.oecd.org/daf/competition/44445750.pdf (accessed 13.10.2016).
- Peltzman, S. (2000), "Prices Rise Faster than They Fall", *Journal of Political Economy*, Vol. 108, No. 3, pp. 466-502.
- Perekhozhuk, O., Glauben, T., Grings, M., and R. Teuber (2016), "Approaches and methods for the econometric analysis of market power: a survey and empirical comparison", *Journal of Economic Surveys*, forthcoming, pp. 1-23.
- Perekhozuk, O., Glauben, T., Teuber, R., and Grings, M. (2015), "Regional-Level Analysis of Oligopsony Power in the Ukrainian Dairy Industry", *Canadian Journal of Agricultural Economics*, Vol. 61, No.1, pp. 43-76.
- Perekhozhuk, O., Hockmann, H., Fertő, I., and Bakucs, L.Z. (2013), "Identification of Market Power in the Hungarian Dairy Industry: A Plant-Level Analysis", *Journal of Agricultural* & Food Industrial Organization, Vol. 11, No. 1, pp. 1-13.
- Perloff, J. M., and Shen, E.Z. (2012), "Collinearity in linear structural models of market power", *Review of Industrial Organization*, Vol. 40, No. 2, pp: 131–138.

- Perloff, J.M., Karp, L.S., and Golan, A. (2007), *Estimating Market Power and Strategies,* Cambridge, UK: Cambridge University Press.
- Popovics, P.A. (2015), "Analysis of economic issues relating to the dairy sector, with emphasis on price transmission", *Applied Studies in Agribusiness and Commerce*, available at: http://ageconsearch.umn.edu/bitstream/48338/2/Analysis%20of%20economic%20 issues%20relating%20to%20the%20dairy.pdf (accessed 28.12.16).
- Porter, R.H. (1983a), "Optimal Cartel Trigger Price Strategies", *Journal of Economic Theory*, Vol. 29, No. 2, pp. 313-338.
- Porter, R.H. (1983b), "A Study of Cartel Stability: The Joint Executive Committee 1880-1886", The Bell Journal of Economics, Vol. 14, pp. 301-314.
- Pray, C., Oehmke, J.F., and Naseem, A. (2005), "Innovation and dynamic efficiency in plant biotechnology: An introduction to the researchable issues", *AgBioForum*, Vol. 8, No. 2/3, pp. 52-63.
- Reagan, P.B., and Weitzman, M.L. (1982), "Asymmetries in Price and Quantity Adjustments by the competitive Firm", *Journal of Economic Theory*, Vol. 27, No. 2, pp. 410-420.
- Reinsel, G.C. (1993), Elements of Multivariate Time Series Analysis, New York, USA: Springer-Verlag.
- Reynolds, S.S. (1991), "Dynamic oligopoly with capacity adjustment costs", *Journal of Economic Dynamics and Control*, Vol. 15, pp. 491-514.
- Reynolds, S.S. (1987), "Capacity Investment, Preemption and Commitment in an Infinite Horizon Model", *International Economic Review*, Vol. 28, No. 1, pp. 69-88.
- Rezitis, A.N., and Reziti, I. (2011), "Threshold cointegration in the Greek milk market", *Journal of International Food & Agribusiness Marketing*, Vol. 23, No. 3, pp. 231-246.
- Richards, T.J., Patterson, P.M., and Acharya, R.N. (2001), "Price Behavior in a Dynamic Oligopsony: Washington Processing Potatoes", American Journal of Agricultural Economics, Vol. 83, No. 2, pp. 259-271.
- Roberts, M.J., and Samuelson, L. (1988), "An Empirical Analysis of Dynamic, Nonprice Competition in an Oligopolistic Industry", *The RAND Journal of Economics*, Vol. 19, No. 2, pp. 200-220.
- Rogers, R.T., and Sexton, R.J. (1994), "Assessing the Importance of Oligopsony Power in Agricultural Markets", American Journal of Agricultural Economics, Vol. 76, No. 5, pp. 1143-1150.
- Romain, R., Doyon, M., and Frigon, M. (2002), "Effects of state regulations on marketing margins and price transmission asymmetry: Evidence from the New York City and upstate New York fluid milk markets", Agribusiness, Vol. 18, No. 3, pp. 301-315.
- Rude, J., Harrison, D., and Carlberg, J. (2011), "Market Power in Canadian Beef Packing", *Canadian Journal of Agricultural Economics*, Vol. 59, No. 3, pp. 321-336.

- Rumánková, L. (2012), "Examination of Market Structure in Selected Livestock Agri-Food Chains in the Czech Republic", Acta Universitatis Agriculturae et Silviculturae Mendelianae Brunensis, Vol. 60, No. 7, pp. 243-258.
- Salhofer, K., Tribl, C., and Sinabell, F. (2012), "Market power in Austrian food retailing: the case of milk products", *Empirica – Journal of European Economics* 39, Vol. 1, pp. 109-122.
- Scalco, P.R., and Braga, M.J. (2014), "Measuring the Degree of Oligopsony Power in the Brazilian Raw Milk Market", International Food and Agribusiness Management Review, Vol. 17, No. 2, pp. 1-20.
- Schaffer, S. (1982), "Competition, Conduct and Demand Elasticity", *Economic Letters*, Vol. 10, No. 5, pp. 167-171.
- Schroeter, J. and Azzam, A. (1991), "Marketing Margins, Market Power, and Price Uncertainty", American Journal of Agricultural Economics, Vol. 73, No. 4, pp. 990-999.
- Schroeter, J. and Azzam, A. (1990), "Measuring Market Power in Multi-Product Oligopolies, the US Meat Industry", *Applied Economics*, Vol. 22, No. 10, pp. 1365-1376.
- Schroeter, J.R. (1988), "Estimating the Degree of Market Power in the Beef Packing Industry", *The Review of Economics and Statistics*, Vol. 70, No. 1, pp. 158-162.
- Sckokai, P., Soregaroli, C., and Moro, D. (2013), "Estimating Market Power by Retailers in a Dynamic Framework: the Italian PDO Cheese Market", *Journal of Agricultural Economics*, Vol. 64, No. 1, pp. 33-53.
- Serra, T., and Goodwin, B.K. (2003), "Price transmission and asymmetric adjustment in the Spanish dairy sector", *Applied Economics*, Vol. 35, No. 18, pp. 1889-1899.
- Sheldon, I., and Sperling, R. (2003), "Estimating the Extent of Imperfect Competition in the Food Industry: What Have We Learned?", *Journal of Agricultural Economics*, Vol. 54, No. 1, pp. 89-109.
- Sexton, R.J. (2013), "Market Power, Misconcepts, and Modern Agricultural Markets", American Journal of Agricultural Economics, Vol. 95, No. 2, pp. 209-219.
- Sexton, R.J. and Lavoie, N. (2001), "Food processing and Distribution: An Industrial Organization Approach", In Gardner, B., and Rausser, G.C. (eds.), Handbook of Agricultural Economics 1B, Amsterdam, The Netherlands: Elsevier Science.
- Sexton, R.J. (2000), "Industrialization and Consolidation in the U.S. Food Sector: Implications for Competition and Welfare", American Journal of Agricultural Economics, Vol. 82, No. 5, pp. 1087-1104.
- Sheldon, I. and Sperling, R. (2003), "Estimating the Extent of Imperfect Competition in the Food Industry: What Have We Learned?", *Journal of Agricultural Economics*, Vol. 54 No. 1, pp. 89-109.
- Shepard, R.W. (1970), *Theory of Cost and Production Functions*, Princeton, N.J., USA: Princeton University Press.

- Shumway, R.H., and Stoffer, D. S. (2000), *Time Series Analysis and Its Applications*, New York, USA: Springer.
- Slade, M.E. (1998), "Beer and the Tie: Did Divestiture of Brewer-Owned Public Houses Lead to Higher Beer Prices?", *The Economic Journal*, Vol. 108, pp. 565-602.
- Slade, M.E. (1995), "Empirical Games: The Oligopoly Case", Canadian Journal of Economics, Vol. 28, pp. 368-402.
- Slade, M.E. (1992), "Vancouver's Gasoline-Price Wars: An Empirical Exercise in Uncovering Supergame Strategies", *Review of Economic Studies*, Vol. 59, pp. 257-276.
- Slade, M.E. (1989), "Price Wars in price-Setting Supergames", Economica, New Series, Vol. 56, No. 223, pp. 295-310.
- Sperling, R. (2002), Estimating Oligopsony Power in the United States Market for Slaughter Hogs: An Error Correction Approach. Electronic Thesis or Dissertation available at: https://etd.ohiolink.edu/ (accessed 30.12.2016).
- Statistisches Bundesamt (2016a), Preise: Daten zur Energiepreisentwicklung Lange Reihe von Januar 2000 bis Novemeber 2016, Wiesbaden, Germany: Statistisches Bundesamt.
- Statistisches Bundesamt (2016b), Bruttoverdienste, Wochenarbeitszeit: Deutschland, Quartale, Wirtschaftszweige, Leistungsgruppen, Geschlecht, available at: https:// www-genesis.destatis.de/genesis/online/data;jsessionid= 7DA5BD39D4719%20 C518A8%20F99A584C256D7.tomcat_GO_1_3?operation=abruftabelleBearbeiten& levelindex=2&levelid=1482934850653&auswahloperation=abruftabelleAuspraegung Auswaehlen&auswahlverzeichnis=ordnungsstruktur&auswahlziel=werteabruf&select ion%20name=62321-0001&auswahltext=&werteabruf=starten (accessed 28.12.2016).
- Steen, F., and Salvanes, K.G. (1999), "Testing for market power using a dynamic oligopoly model", *International Journal of Industrial Organization*, Vol. 17, pp. 147-177.
- Stewart, H., and Blayney, D.P. (2011), "Retail Dairy Prices Fluctuate with the Farm Value of Milk", Agricultural and Resource Economics Review, Vol. 40, No. 2, pp. 201-217.
- Stiegert, K.W., Wang, S.-S., and Rogers, R.T. (2009), "Structural Change and Market Power in the U.S. Food Manufacturing Sector", Agribusiness, Vol. 25, No. 2, pp. 164-180.
- Stigler, G.J. (1964), "A Theory of Oligopoly", The Journal of Political Economy, Vol. 72, No. 1, pp. 44-61.
- Stock, J.H., and Watson, M.W. (2005), "Implications of Dynamic Factor Models for VAR Analysis", *NBER Working Papers* 11467.
- Sutton, J. (1998), *Technology and Market Structure: Theory and History*, Cambridge and London, UK: MIT Press.
- Suzuki, N., Lenz, J.E., and Forker, O.D. (1993), "A Conjectural Variations Model of Reduced Japanese Milk Price Supports", American Journal of Agricultural Economics, Vol. 75, No. 1, pp. 210-218.

- Swinnen, J.F.M., and Vandeplas, A. (2010), "Market power and rents in global supply chains", *Agricultural Economics*, Vol. 41, No. s1, pp.109-120.
- Tribl, C., and Salhofer, K. (2013), Marktmacht und räumlicher Wettbwerb entlang der Wertschöpfungskette von Milch, Abschlußbericht zu Projekt AWI/158/06 "Industrieökonomische Analysen der Sektoren Landwirtschaft und Lebensmittelverarbeitung", Vienna, Austria: AWI - Bundesanstalt für Agrarwirtschaft.
- Tsay, R. S. (1989), "Testing and Modeling Threshold Autoregressive Processes", *Journal* of the American Statistical Association, Vol. 84, No. 405, pp. 231-240.
- Tweeten, L.G., and Quance, C.L. (1969), "Positivistic Measures of aggregate Supply Elasticities: Some new Approaches", American Journal of Agricultural Economics, Vol. 59, No. 2, pp. 342-352.
- von Cramon-Taubadel, S. (1998), "Estimating asymmetric Price Transmission with the Error Correction Representation: An Application to the German Pork Market", *European Review of Agricultural Economics*, Vol. 25, No. 1, pp. 1-18
- von Cramon-Taubadel, S., and Fahlbusch, S. (1994), "Identifying asymmetric price transmission with error correction models", Poster Session EAAE European Seminar in Reading.
- von Cramon-Taubadel, S., and Loy, J.-P. (1996), "Price Asymmetry in the international Wheat Market: Comment", *Canadian Journal of Agricultural Economics*, Vol. 44, No. 3, pp. 311-317.
- Ward, R.W. (1982), "Asymmetry in Retail, Wholesale, and Shipping Point Pricing for Fresh Vegetables", American Journal of Agricultural Economics, Vol. 64, No. 2, pp. 205-212.
- Weatherell, C., Tregear, A. and Allinson, J. (2003), "In Search of the Concerned Consumer: UK Public Perceptions of Food, Farming and Buying Local", *Journal of Rural Studies*, Vol. 19, No. 2, pp. 233-244.
- Weldegebriel, H.T, Wang, X., and Rayner, A.J. (2012), "Price transmission market power and industry technology: a note", *China Agricultural Economic Review*, Vol. 4, No. 3, pp. 281-299.
- Weldegebriel, H.T. (2004), "Imperfect Price Transmission: Is Market Power Really to Blame", *Journal of Agricultural Economics*, Vol. 55, No. 1, pp. 101-144.
- Wildi, M. (2013), *An Introduction to State Space Models*, available at: https://blog.zhaw. ch/ sef/files/state_space_script.pdf (accessed 28.12.2016).
- Wohlgenant, M.K. (2001), "Marketing margins: Empirical analysis", Gardner, B.L., and Rausser, G.C. (eds), *Handbook of Agricultural Economics 1B*, Amsterdam, The Netherlands: North Holland, pp. 933-970.
- Wohlgenant, M. K. (1989), "Demand for Farm Output in a complete System of Demand Functions", *American Journal of Agricultural Economics*, Vol. 72, pp. 241-52.

- Wolffram, R. (1971), "Positivistic Measures of aggregate Supply Elasticities: Some new Approaches some critical Notes", *American Journal of Agricultural Economics*, Vol. 53, No. 2, pp. 356-359.
- Zentrale Markt-und Preisberichtsstelle (ZMP) (ed.) (different volumes), *ZMP-Marktbilanz Getreide*, Bonn, Germany: Zentrale Markt-und Preisberichtsstelle GmbH.
- Zentrale Markt-und Preisberichtsstelle (ZMP) (ed.) (different volumes), *ZMP-Marktbilanz Milch*, Bonn, Germany: Zentrale Markt-und Preisberichtsstelle GmbH.
- Zuur, A.F., Fryer, R.J., Jolliffe, I.T., Dekker, R., and Beukema, J.J. (2003), "Estimating common trends in multivariate time series using dynamic factor analysis", *Environmetrics*, Vol. 14, pp. 665-685.

APPENDIX

(A1) Derived inverse supply function of material input: $\frac{h}{2}$

$$W^{M} = \rho^{X} + \sum_{i=1}^{n} \rho_{i}^{NX} W_{i}^{F} + \rho^{TX} T^{F} + \rho^{XX} X^{M}$$

(A2) Aggregated revenue function of the processing industry:

$$\begin{split} R^{P} &= \varphi^{0} + \varphi^{P}P + \varphi^{PP} \frac{P^{2}}{2} + \varphi^{X}X^{M} + \varphi^{XX} \frac{X^{M^{2}}}{2} + \sum_{j=1}^{k} \varphi_{j}^{N}W_{j}^{P} + \sum_{j=1}^{k} \varphi_{j}^{NN} \frac{W_{j}^{P^{2}}}{2} + \varphi^{T}T^{P} + \varphi^{TT} \frac{T^{P^{2}}}{2} \\ &+ \varphi^{PX}PX^{M} + \sum_{j=1}^{k} \varphi_{j}^{PN}W_{j}^{P}P + \varphi^{PT}T^{P}P + \sum_{j=1}^{k} \varphi_{j}^{XN}W_{j}^{P}X^{M} + \varphi^{XT}X^{M}T^{P} \\ &+ \sum_{\substack{i=1\\i\neq j}}^{k} \sum_{j=1}^{k} \varphi_{ij}^{NN}W_{i}^{P}W_{j}^{P} + \sum_{j=1}^{k} \varphi_{j}^{NT}W_{j}^{P}T^{P} \end{split}$$

(A3) Derived demand function for material input:

$$W^M + \rho^{XX} X^M \Xi_t^P = \varphi^X + \varphi^{PX} P + \varphi^{XX} X^M + \sum_{j=1}^{\kappa} \varphi_j^{XN} W_j^P + \varphi^{XT} T^P$$

(A4) Derived inverse supply function of output:

$$P = -\frac{-Q + \varphi^{P} + \varphi^{PX}X^{M} + \sum_{j=1}^{k} \varphi_{j}^{PN}W_{j}^{P} + \varphi^{PT}T^{P}}{\varphi^{PP}}$$

(A5) Derived demand function for output:

$$P + \frac{Q\Xi_t^R}{\varphi^{PP}} = \vartheta^Q + \vartheta^{QQ}Q + \sum_{k=1}^m \vartheta_k^{NQ} W_k^R + \vartheta^{TQ} T^R$$



P = -

(A6) Explicit solutions for the endogenous variables X^{M} , W^{M} , Q, and P:











$$(A11) \text{ The ECTs of the ECM with two cointegration relationships:} \\ \left(\begin{array}{ccc} ECT_{FPE} \\ 0 & 1 & \beta_{2t}^{WW} \\ 0 & 1 & \beta_{2t}^{WW} \end{array} \right) = \left(\begin{array}{cccc} B_{1t}^{W} \\ 0 & 1 & \beta_{2t}^{W} \\ 0 & 1 & \beta_{2t}^{W} \end{array} \right) = \left(\begin{array}{cccc} B_{1t}^{W} \\ 0 & 1 & \beta_{2t}^{W} \\ 0 & 1 & \beta_{2t}^{W} \end{array} \right) \left(\begin{array}{cccc} B_{1t}^{W} \\ B_{2t}^{W} \\ \beta_{2t}^{W} \end{array} \right) \left(\begin{array}{cccc} B_{1t}^{W} \\ B_{2t}^{W} \\ \beta_{2t}^{W} \end{array} \right) \left(\begin{array}{cccc} B_{1t}^{W} \\ B_{2t}^{W} \\ \beta_{2t}^{W} \end{array} \right) \left(\begin{array}{cccc} B_{1t}^{W} \\ B_{2t}^{W} \\ \beta_{2t}^{W} \end{array} \right) \left(\begin{array}{cccc} B_{1t}^{W} \\ B_{2t}^{W} \\ \beta_{2t}^{W} \end{array} \right) \left(\begin{array}{cccc} B_{1t}^{W} \\ B_{2t}^{W} \\ \beta_{2t}^{W} \end{array} \right) \left(\begin{array}{cccc} B_{1t}^{W} \\ B_{2t}^{W} \\ \beta_{2t}^{W} \end{array} \right) \left(\begin{array}{cccc} B_{1t}^{W} \\ B_{2t}^{W} \\ \beta_{2t}^{W} \end{array} \right) \left(\begin{array}{cccc} B_{1t}^{W} \\ B_{2t}^{W} \\ \beta_{2t}^{W} \end{array} \right) \left(\begin{array}{cccc} B_{1t}^{W} \\ B_{2t}^{W} \\ \beta_{2t}^{W} \end{array} \right) \left(\begin{array}{cccc} B_{1t}^{W} \\ B_{2t}^{W} \\ \beta_{2t}^{W} \end{array} \right) \left(\begin{array}{cccc} B_{1t}^{W} \\ B_{2t}^{W} \\ \beta_{2t}^{W} \end{array} \right) \left(\begin{array}{cccc} B_{1t}^{W} \\ B_{2t}^{W} \\ \beta_{2t}^{W} \end{array} \right) \left(\begin{array}{cccc} B_{1t}^{W} \\ B_{2t}^{W} \\ \beta_{2t}^{W} \end{array} \right) \left(\begin{array}{cccc} B_{1t}^{W} \\ B_{2t}^{W} \\ \beta_{2t}^{W} \end{array} \right) \left(\begin{array}{cccc} B_{1t}^{W} \\ B_{2t}^{W} \\ \beta_{2t}^{W} \end{array} \right) \left(\begin{array}{cccc} B_{1t}^{W} \\ B_{2t}^{W} \\ \beta_{2t}^{W} \end{array} \right) \left(\begin{array}{cccc} B_{1t}^{W} \\ B_{2t}^{W} \\ B_{2t}^{W} \\ \beta_{2t}^{W} \end{array} \right) \left(\begin{array}{cccc} B_{1t}^{W} \\ B_{2t}^{W} \\ B_{2t}^{W} \\ B_{2t}^{W} \\ B_{2t}^{W} \end{array} \right) \left(\begin{array}{cccc} B_{1t}^{W} \\ B_{2t}^{W} \\ B_$$



Figure (A1): P-values of time-varying parameters of FPE coint. vector of model with 1 lag and endogenous non-material input variables (dotted: p-value of 0.05 and 0.1).

Source: own elaboration.



Figure (A2): P-values of time-varying parameters of FPE coint. vector of model with 2 lags and endogenous non-material input variables (dotted: p-value of 0.05 and 0.1).

Source: own elaboration.



Figure (A3): P-values of time-varying parameters of FPE coint. vector of model with 12 lags and endogenous non-material input variables (dotted: p-value of 0.05 and 0.1).

Source: own elaboration.



Figure (A4): P-values of time-varying parameters of FPE coint. vector of model with 1 lag and weakly exogenous non-material input variables (dotted: p-value of 0.05 and 0.1).

Source: own elaboration.



Figure (A5): P-values of time-varying parameters of FPE coint. vector of model with 2 lags and weakly exogenous non-material input variables (dotted: p-value of 0.05 and 0.1).

Source: own elaboration.



Figure (A6): P-values of time-varying parameters of FPE coint. vector of model with 12 lags and weakly exogenous non-material input variables (dotted: p-value of 0.05 and 0.1).

Source: own elaboration.



Figure (A7): P-values of time-varying parameters of FPE coint. vector of model with 1 lag and exogenous non-material input variables (dotted: p-value of 0.05 and 0.1).

Source: own elaboration.



Figure (A8): P-values of time-varying parameters of FPE coint. vector of model with 2 lags and exogenous non-material input variables (dotted: p-value of 0.05 and 0.1).

Source: own elaboration.



Figure (A9): P-values of time-varying parameters of FPE coint. vector of model with 12 lags and exogenous non-material input variables (dotted: p-value of 0.05 and 0.1).

Source: own elaboration.



Figure (A10): P-values of time-varying parameters of PRE coint. vector of model with 12 lags and exogenous non-material input variables (dotted: p-value of 0.05 and 0.1).

Source: own elaboration.



Figure (A11): P-values of time-varying parameter of PRE coint. vector of model with 2 lags and weakly exogenous non-material input variables (dotted: p-value of 0.05 and 0.1).

Source: own elaboration.



Figure (A12): P-value of dynamic factor of FPE with 5%- and 10%-significance bands.

Source: own elaboration.



Figure (A13): P-value of dynamic factor of PRE with 5%- and 10%-significance bands.

Source: own elaboration.

Studies on the Agricultural and Food Sector in Transition Economies

edited by Leibniz Institute of Agricultural Development in Transition Economies (IAMO)

ISSN 1436-221X

- Vol. 1 The importance of institutions for the transition in Central and Eastern Europe with emphasis on agricultural and food industry ed. by Klaus Frohberg and Witold-Roger Poganietz 1998, 137 pages, ISBN 3-8175-0258-3
- Vol. 2 The significance of politics and institutions for the design and formation of agricultural Policies
 ed. by Klaus Frohberg and Peter Weingarten
 1999, 254 pages, ISBN 3-8175-0289-3
- Vol. 3 Food processing and distribution in transition countries. Problems and perspectives
 ed. by Monika Hartmann and Jürgen Wandel
 1999, 349 pages, ISBN 3-8175-0293-1
- Vol. 4 Die private Nachfrage nach Nahrungsmitteln im Transformationsprozeß Tschechiens und Polens
 Stephan Brosig (PhD)
 2000, 171 Seiten, ISBN 3-8175-0319-9
- Vol. 5 Integrating Estonia into the EU: Quantitative analysis of the agricultural and food sector Achim Fock (PhD) 2000, 286 pages, ISBN 3-8175-0320-2
- Vol. 6 Competitiveness of agricultural enterprises and farm activities in transition countries ed. by Peter Tillack and Frauke Pirscher 2000, 216 pages, ISBN 3-8175-0322-9
- Vol. 7 Конкурентоспособность сельскохозяйственных предприятий и фермерской деятельности в странах переходного периода под редакцией Петера Тиллака и Фрауке Пиршер 2000, 253 страницы, ISBN 3-8175-0324-5

- Vol. 8 Perspectives on agriculture in transition: Analytical issues, modelling approaches, and case study results ed. by Witold-Roger Poganietz, Alberto Zezza, Klaus Frohberg and Kostas G. Stamoulis 2000, 433 pages, ISBN 3-8175-0323-7
- Vol. 9 Land ownership, land markets and their influence on the efficiency of agricultural production in Central and Eastern Europe
 ed. by Peter Tillack and Eberhard Schulze
 2000, 485 pages, ISBN 3-8175-0325-3
- Vol. 10 Landwirtschaft und Industrie in Russland Der Transformationsprozeß in der Ernährungsindustrie Jürgen Wandel (PhD) 2000, 361 Seiten, ISBN 3-8175-0334-2
- Vol. 11 Food consumption in Russia. An econometric analysis based on household data Karin Elsner (PhD) 2001, 256 pages, ISBN 3-8175-0335-0
- Vol. 12 Alexander Wasiljewitsch Tschajanow Die Tragödie eines großen Agrarökonomen
 hrsg. u. übers. von Eberhard Schulze
 2001, 192 Seiten, ISBN 3-8175-0342-3
- Vol. 13 Analysis of food consumption in Central and Eastern Europe: Relevance and empirical methods
 ed. by Stephan Brosig and Monika Hartmann
 2001, 253 pages, ISBN 3-8175-0349-0
- Vol. 14 Wettbewerbsprozesse und Firmenwachstum in der Transformation am Beispiel der polnischen Fleischindustrie Agata Pieniadz (PhD) 2002, 291 Seiten, ISBN 3-8175-0360-1
- Vol. 15 Agricultural enterprises in transition: Parallels and divergences in Eastern Germany, Poland and Hungary
 ed. by Ludger Hinners-Tobrägel and Jürgen Heinrich 2002, 455 pages, ISBN 3-8175-0366-0

- Vol. 16 Agricultural technology and economic development of Central and Eastern Europe. Results of the workshop in Halle, 2nd-3rd July 2001
 ed. by Peter Tillack and Ulrich Fiege
 2002, 160 pages, ISBN 3-86037-199-1
- Vol. 17 Региональные аспекты аграрных преобразований: Политика, реструктуризация, рыночная адаптация под редакцией Петера Тиллака и Виталия Зиновчука 2003, 236 страницы, ISBN 3-928466-55-0
- Vol. 18 Alexander Vasilievich Chayanov The tragedy of an outstanding agricultural economist ed. by Eberhard Schulze 2003, 188 pages, ISBN 3-86037-201-7
- Vol. 19 Development of agricultural market and trade policies in the CEE Candidate Countries by the Network of Independent Agricultural Experts in the CEE Candidate Countries 2003, 72 pages, ISBN 3-86037-212-2
- Vol. 20 Large farm management ed. by Alfons Balmann and Alexej Lissitsa 2003, 396 pages, ISBN 3-86037-213-0
- Vol. 21 Success and failures of transition The Russian agriculture between fall and resurrection
 ed. by Eberhard Schulze, Elke Knappe, Eugenia Serova, Peter Wehrheim 2003, 521 pages, ISBN 3-9809270-1-6
- Vol. 22 Subsistence agriculture in Central and Eastern Europe: How to break the vicious circle? ed. by Steffen Abele and Klaus Frohberg 2003, 233 pages, ISBN 3-9809270-2-4
- Vol. 23 Pfadabhängigkeiten und Effizienz der Betriebsstrukturen in der ukrainischen Landwirtschaft – Eine theoretische und empirische Analyse
 Andriy Nedoborovskyy (PhD)
 2004, 197 Seiten, ISBN 3-86037-216-5

- Vol. 24 Nichtmonetäre Transaktionen in der ukrainischen Landwirtschaft: Determinanten, Spezifika und Folgen
 Olena Dolud (PhD)
 2004, 190 Seiten, ISBN 3-9809270-3-2
- Vol. 25 **The role of agriculture in Central and Eastern European rural development: Engine of change or social buffer?** ed. by Martin Petrick and Peter Weingarten 2004, 426 pages, ISBN 3-9809270-4-0
- Vol. 26 Credit rationing of Polish farm households A theoretical and empirical analysis
 Martin Petrick (PhD)
 2004, 254 pages, ISBN 3-9809270-6-7
- Vol. 27 Drei Jahrhunderte Agrarwissenschaft in Russland: Von 1700 bis zur Gegenwart Alexander Alexandrowitsch Nikonow und Eberhard Schulze 2004, 232 Seiten, ISBN 3-9809270-8-3
- Vol. 28 Russlands Weg vom Plan zum Markt: Sektorale Trends und regionale Spezifika Peter Voigt (PhD)

2004, 270 Seiten, ISBN 3-9809270-9-1

- Vol. 29 Auswirkungen des Transformationsprozesses auf die sozioökonomischen Funktionen ukrainischer Landwirtschafts-unternehmen Helga Biesold (PhD)
 2004 182 Seiten, ISBN 3-938584-00-9
- Vol. 30 Agricultural policies and farm structures Agent-based modelling and application to EU-policy reform
 Kathrin Happe (PhD)
 2004, 291 pages, ISBN 3-938584-01-7
- Vol. 31 How effective is the invisible hand? Agricultural and food markets in Central and Eastern Europe
 ed. by Stephan Brosig and Heinrich Hockmann
 2005, 361 pages, ISBN 3-938584-03-3

- Vol. 32 Erfolgsfaktoren von landwirtschaftlichen Unternehmen mit Marktfruchtanbau in Sachsen-Anhalt Kirsti Dautzenberg (PhD) 2005, 161 Seiten, ISBN 3-938584-06-8
- Vol. 33 Agriculture in the face of changing markets, institutions and policies: Challenges and strategies
 ed. by Jarmila Curtiss, Alfons Balmann, Kirsti Dautzenberg, Kathrin Happe
 2006, 544 pages, ISBN 3-938584-10-6
- Vol. 34 Making rural households' livelihoods more resilient The importance of social capital and the underlying social networks
 ed. by Gertrud Buchenrieder and Thomas Dufhues
 2006, 106 pages, ISBN 3-938584-13-0
- Vol. 35 Außerlandwirtschaftliche Diversifikation im Transformations-prozess. Diversifikationsentscheidungen und -strategien ländlicher Haushalte in Slowenien und Mazedonien Judith Möllers (PhD) 2006, 323 Seiten, ISBN 3-938584-14-9
- Vol. 36 Accessing rural finance The rural financial market in Northern Vietnam

Thomas Dufhues (PhD) 2007, 166 Seiten, ISBN 3-938584-16-5

- Vol. 37 Страхование посевов в Казахстане: Анализ возможностей эффективного управления рисками Раушан Бокушева, Олаф Хайдельбах, Талгат Кусайынов 2007, 82 Seiten, ISBN 3-938584-17-3
- Vol. 38 Rethinking agricultural reform in Ukraine
 Zvi Lerman, David Sedik, Nikolai Pugachov, Aleksandr Goncharuk
 2007, 167 Seiten, ISBN 3-938584-18-1
- Vol. 39 Sustainable rural development: What is the role of the agri-food sector?
 ed. by Martin Petrick, Gertrud Buchenrieder
 2007, 293 pages, ISBN 3-938584-22-X

- Vol. 40 Efficiency of selected risk management instruments An empirical analysis of risk reduction in Kazakhstani crop production
 Olaf Heidelbach (PhD)
 2007, 223 Seiten, ISBN 3-938584-19-X
- Vol. 41 Marktstruktur und Preisbildung auf dem ukrainischen Markt für Rohmilch Oleksandr Perekhozhuk (PhD) 2007, 274 Seiten, ISBN 978-3-938584-24-8
- Vol. 42 Labor market behavior of Chinese rural households during transition Xiaobing Wang (PhD)
 2007, 140 Seiten, ISBN 978-3-938584-25-5
- Vol. 43 Continuity and change: Land and water use reforms in rural Uzbekistan. Socio-economic and legal analyses for the region Khorezm ed. by Peter Wehrheim, Anja Schoeller-Schletter, Christopher Martius 2008, 211 Seiten, ISBN 978-3-938584-27-9
- Vol. 44 Agricultural economics and transition: What was expected, what we observed, the lessons learned (Vol I and II)
 ed. by Csaba Csáki, Csaba Forgács
 2008, 634 Seiten, ISBN 978-3-938584-31-6
- Vol. 45 Theoretical and methodological topics in the institutional economics of European agriculture. With applications to farm organisation and rural credit arrangement Martin Petrick
 2008, 223 Seiten, ISBN 978-3-938584-31-6
- Vol. 46 Agri-food business: Global challenges Innovative solutions ed. by Thomas Glauben, Jon H. Hanf, Michael Kopsidis, Agata Pieniadz, Klaus Reinsberg 2008, 152 pages, ISBN 978-3-938584-33-0
- Vol. 47 Eine Analyse der Transformationsberatung für die "kollektive Landwirtschaft" während der ersten Transformationsphase (1989-1991) am Beispiel Ostdeutschlands: Lehren für Korea Jeong Nam Choi (PhD) 2009, 225 Seiten, ISBN 978-3-938584-36-1
- Vol. 48 Croatia's EU accession. Socio-economic assessment of farm households and policy recommendations
 Judith Möllers, Patrick Zier, Klaus Frohberg, Gertrud Buchenrieder and Štefan Bojnec
 2009, 196 Seiten, ISBN 978-3-938584-35-4
- Vol. 49 Structural change in Europe's rural regions. Farm livelihoods between subsistence orientation, modernisation and non-farm diversification ed. by Gertrud Buchenrieder Judith Möllers 2009, 166 Seiten, ISBN 978-3-938584-39-2
- Vol. 50 Motive beim Weinkonsum Unterschiede zwischen deutschen und ukrainischen Konsumenten Astrid Lucie Rewerts (PhD) 2009, 267 Seiten, ISBN 978-3-938584-40-8
- Vol. 51 Rural development as provision of local public goods: Theory and evidence from Poland Andreas Gramzow (PhD) 2009, 203 Seiten, ISBN 978-3-938584-41-5
- Vol. 52 Multi-level Processes of Integration and Disintegration. Proceedings of the Third Green Week Scientific Conference
 ed. by Franziska Schaft, Alfons Balmann
 2009, 216 Seiten, ISBN 978-3-938584-42-2
- Vol. 53 Zur Bestimmung der Wettbewerbsfähigkeit des weißrussischen Milchsektors: Aussagefähigkeit von Wettbewerbsindikatoren und Entwicklung eines kohärenten Messungskonzepts Mikhail Ramanovich (PhD) 2010, 202 Seiten, ISBN 978-3-938584-44-6
- Vol. 54 Die Internationalisierung landwirtschaftlicher Unternehmen. Das Beispiel deutscher, dänischer und niederländischer Direktinvestitionen in den ukrainischen Agrarsektor Henriette Stange (PhD)
 2010, 296 Seiten, ISBN 978-3-938584-45-3

- Vol. 55 Verhandlungsverhalten und Anspruchsanpassung im inter-nationalen Verhandlungsprozess: Die WTO-Agrarverhandlungen zum Abbau exportwettbewerbsfördernder Maßnahmen Ildiko Lajtos (PhD) 2010, 195 Seiten, ISBN 978-3-938584-48-4
- Vol. 56 Challenges of education and innovation. Proceedings of the Fourth Green Week Scientific Conference
 ed. by Kelly Labar, Martin Petrick, Gertrud Buchenrieder
 2010, 155 Seiten, ISBN 978-3-938584-49-1
- Vol. 57 Agriculture in the Western Balkan Countries ed. by Tina Volk 2010, 249 Seiten, ISBN 978-3-938584-51-4
- Vol. 58 **Perspectives on Institutional Change Water Management in Europe** ed. by Insa Theesfeld, Frauke Pirscher 2011, 127 Seiten, ISBN 978-3-938584-52-1
- Vol. 59 Der ukrainische Außenhandel mit Produkten der Agrar- und Ernährungswirtschaft: Eine quantitative Analyse aus Sicht traditioneller und neuer Außenhandelstheorien Inna Levkovych (PhD) 2011, 232 Seiten, ISBN 978-3-938584-53-8
- Vol. 60 Regional structural change in European agriculture: Effects of decoupling and EU accession
 Christoph Sahrbacher (PhD)
 2011, 244 Seiten, ISBN 978-3-938584-58-3
- Vol. 61 Structural Change in Agriculture and Rural Livelihoods:
 Policy Implications for the New Member States of the European Union ed. by Judith Möllers, Gertrud Buchenrieder, Csaba Csáki 2011, 247 Seiten, ISBN 978-3-938584-59-0
- Vol. 62 Improving the functioning of the rural financial markets of Armenia Milada Kasarjyan (PhD)
 2011, 121 Seiten, ISBN 978-3-938584-60-6

- Vol. 63 Integrierte Strukturen im Agrar- und Ernährungssektor Russlands: Entstehungsgründe, Funktionsweise, Entwicklungsperspektiven und volkswirtschaftliche Auswirkungen Jürgen Wandel 2011, 758 Seiten, ISBN 978-3-938584-61-3
- Vol. 64 Goal Achievement in Supply Chain Networks A Study of the Ukrainian Agri-Food Business
 Taras Gagalyuk (PhD)
 2012, 204 Seiten, ISBN 978-3-938584-63-7
- Vol. 65 Impacts of CAP reforms on farm structures and performance disparities – An agent-based approach Amanda Sahrbacher (PhD) 2012, 284 Seiten, ISBN 978-3-938584-64-4
- Vol. 66 Land fragmentation and off-farm labor supply in China Lili Jia (PhD) 2012, 143 Seiten, ISBN 978-3-938584-65-1
- Vol. 67 Ausprägung interregionaler Disparitäten und Ansätze zur Entwicklung ländlicher Räume in Mittel- und Osteuropa
 Sabine Baum (PhD)
 2012, 214 Seiten, ISBN 978-3-938584-68-2
- Vol. 68 Patterns Behind Rural Success Stories in the European Union: Major Lessons of Former Enlargements
 ed. by Axel Wolz, Carmen Hubbard, Judith Möllers, Matthew Gorton, Gertrud Buchenrieder
 2012, 190 Seiten, ISBN 978-3-938584-69-9
- Vol. 69 Motives for remitting from Germany to Kosovo Wiebke Meyer (PhD) 2012, 142 Seiten, ISBN 978-3-938584-70-5
- Vol. 70 Effizienz russischer Geflügelfleischproduzenten: Entwicklung und Determinanten Elena Epelstejn (PhD) 2013, 272 Seiten, ISBN 978-3-938584-72-9

- Vol. 71 Econometric impact assessment of the Common Agricultural Policy in East German agriculture Patrick Zier (PhD) 2013, 172 Seiten, ISBN 978-3-938584-73-6
- Vol. 72 Determinants of non-farm entrepreneurial intentions in a transitional context: Evidence from rural Bulgaria
 Diana Traikova (PhD)
 2013, 136 Seiten, ISBN 978-3-938584-75-0
- Vol. 73 Human capital differences or labor market discrimination? The occupational outcomes of ethnic minorities in rural Guizhou (China) Bente Castro Campos (PhD) 2013, 266 Seiten, ISBN 978-3-938584-76-7
- Vol. 74 Identifying and understanding the patterns and processes of forest cover change in Albania and Kosovo
 Kuenda Laze (PhD)
 2014, 152 Seiten, ISBN 978-3-938584-78-1
- Vol. 75 Flexibilität von Unternehmen. Eine theoretische und empirische Analyse
 Swetlana Renner (PhD)
 2014, 194 Seiten, ISBN 978-3-938584-79-8
- Vol. 76 Impact of policy measures on wheat-to-bread supply chain during the global commodity price peaks: The case of Serbia
 Ivan Djuric (PhD)
 2014, 160 Seiten, ISBN 978-3-938584-80-4
- Vol. 77 Marktwirtschaftliche Koordination: Möglichkeiten und Grenzen.
 Symposium anlässlich des 75. Geburtstages von Prof. Dr. Dr. h.c. mult.
 Ulrich Koester
 ed. by Jens-Peter Loy
 2014, 94 Seiten, ISBN 978-3-938584-82-8
- Vol. 78 Participatory governance in rural development: Evidence from Ukraine Vasyl Kvartiuk (PhD)
 2015, 200 Seiten, ISBN 978-3-938584-84-2

- Vol. 79 Agricultural transition in Post-Soviet Europe and Central Asia after 25 years. International workshop in honor of Professor Zvi Lerman
 ed. by Ayal Kimhi, Zvi Lerman
 2015, 314 Seiten, ISBN 978-3-938584-95-8
- Vol. 80 Three essays on the Russian wheat export Zsombor Pall (PhD) 2015, 150 Seiten, ISBN 978-3-938584-86-6
- Vol. 81 Milchproduktion zwischen Pfadabhängigkeit und Pfadbrechung: Partizipative Analysen mit Hilfe des agentenbasierten Modells AgriPoliS Arlette Ostermeyer (PhD) 2015, 335 Seiten, ISBN 978-3-938584-88-0
- Vol. 82 Competitiveness and performance of EU agri-food chains ed. by Heinz Hockmann, Inna Levkovych, Aaron Grau 2016, ISBN 978-3-95992-006-3
- Vol. 83 Market uncertainty, project specificity and policy effects on bioenergy investments. A real options approach
 Lioudmila Chatalova (PhD)
 2016, 202 Seiten, ISBN 978-3-95992-017-9
- Vol. 84 Too much but not enough: Issues of water management in Albania in light of climate change
 Klodjan Rama (PhD)
 2016, 222 Seiten, ISBN 978-3-95992-034-6
- Vol. 85 Business groups in agriculture. Impact of ownership structures on performance: The case of Russia's agroholdings
 Andriy Matyukha (PhD)
 2017, 128 Seiten, ISBN 978-3-95992-039-1
- Vol. 86 Russian demand for dietary quality: Nutrition transition, diet quality measurement, and health investment theory Christine Burggraf (PhD)
 2017, 225 Seiten, ISBN 978-3-95992-040-7

- Vol. 87 **Trade policy impacts on net wheat importers in the CIS: Three essays** Nurzat Baisakova (PhD) 2017, 154 Seiten, ISBN 978-3-95992-041-4
- Vol. 88 Labour rationing of different farm types in Kazakhstan: A shadow price analysis Katharina Vantomme (PhD) 2017, 152 Seiten, ISBN 978-3-95992-043-8
- Vol. 89 Factor productivity in EU agriculture: A microeconometric perspective Mathias Kloss (PhD) 2017, 196 Seiten, ISBN 978-3-95992-054-4
- Vol. 90 Market power in the German dairy supply chain Aaron Grau (PhD) 2018, 184 Seiten, ISBN 978-3-95992-057-5, pdf: ISBN 978-3-95992-062-9

In der Schriftenreihe *Studies on the Agricultural and Food Sector in Transition Economies* werden durch das IAMO Monographien und Tagungsberichte herausgegeben, die sich mit agrarökonomischen Fragestellungen zu Mittel- und Osteuropa beschäftigen. Wissenschaftlern, die in diesem Bereich forschen, steht die Schriftenreihe als Diskussionsforum offen.

In its series *Studies on the Agricultural and Food Sector in Transition Economies* IAMO publishes monographs and proceedings focusing on agricultural economic issues specific to Central and Eastern Europe. This series offers a forum to researchers studying this area.